

Archaeological Research Services Ltd

North East Rapid Coastal Zone Assessment: Phase 2



Human footprints at Low Hauxley, Northumberland, filled with sand and pebbles deeply impressed within the peat that survives as a thin layer within the inter-tidal zone.

ARS Ltd Report 2010/42 May 2010

Compiled by:

Andrew Burn BA PIfA Archaeological Research Services Ltd Baltic Business Centre Saltmeadows Road Gateshead NE8 3DA

> **Checked by:** Dr Clive Waddington

Tel: 0191 477 5111 Fax: 0191 477 7687 admin@archaeologicalresearchservices.com www.archaeologicalresearchservices.com

North East Rapid Coastal Zone Assessment: Phase 2

Archaeological Research Services Ltd Report 2010/42

May 2010

© Archaeological Research Services Ltd mmx

Contents

	List of Figures	5
	List of Tables	14
	Executive Summary	15
1.	Introduction	18
2.	Aims and Objectives	22
3.	Project scope	26
4.	Methodology	33
5.	Archaeological survey results	42
5.1	Introduction	42
5.2	Overdale Wyke	44
5.3	Saltburn and Hummersea	60
5.4	Greatham Creek	76
5.5	Crimdon Dene	94
5.6	Trow Point	106
5.7	Robert's Battery	120
5.8	Druridge Bay	137
5.9	Low Hauxley	151
5.10	Alnmouth	169
5.11	Amble hulks	184
5.12	Scremmerston	193
5.13	Fenham	203
5.14	Nessend, Holy Island	215
5.15	Bamburgh Bowl hole	229
5.16	Northumberland Coast AONB survey	240
5.17	Summary and conclusions	273

6	Palaeoenvironmental survey and sampling	274
6.1	Introduction	274
6.2	Radio Carbon dating results	276
6.3	Hartlepool Bay	279
6.4	Crimdon Dene	283
6.5	Low Hauxley	286
7	Managing the resource	294
7.1	Introduction	294
7.2	Priority sites at 'imminent risk'	302
7.3	Site assessments	302
7.4	Conclusions	338
8	References	335

Appendix : Chapter 5 fold out illustrations 345

© Archaeological Research Services Ltd 2009

List of Figures

1.1	Remains of Tudor period fort overlooking Holy Island harbour, with Lindisfarne castle in the background	19
1.2	Investigating a layer of inter-tidal peat at Low Hauxley, Northumberland	20
1.3	Remains of a chapel on St Cuthbert's Isle, seen eroding as a result of wave action.	21
3.1 3.2	North East Rapid Coastal Zone Assessment Phase 2 Survey Area Graded extant earthwork remains of the buildings of a former medieval grange farm of Lindisfarne priory at Fenham looking south from the hamlet of Fenham	27 28
3.3	Crop marks visible on the ground revealing the trace of medieval broad ridge and furrow at Overdale Wyke, north Yorkshire, view looking West	29
3.4	Unusual double pillbox on the shore at Goswick, survey by the Northumberland coast AONB survey, linked by underground access.	30
3.5	The Knights Hospitaller preceptory at Low Chibburn that was re-used as a WWII pillbox	31
3.6	Recording of a rock cut feature with the Magellan GPS equipment, possibly a fish trap or "hulley", on the foreshore at Hummersea, North Yorkshire	32
4.1	Utilising the Magellan CX mapping grade GPS at Skinnigrove.	35
4.2	Cleaning the buried organic land surface at Crimdon Dene, County Durham.	37
4.3	Investigating peat layers at Low Hauxley Northumberland	38
4.4	Undertaking sediment coring at Hartlepool Bay, Teeside.	39
5.1.1	Surveying earthworks of Second World War defences preserved above Skiningrove Harbour	42
	Archaeology recorded at Overdale Wyke	2.45
5.2.1		- 345
5.2.1 5.2.2	Overdale Wyke Site	345 44
5.2.1 5.2.2 5.2.3	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures	345 44 45
5.2.1 5.2.2 5.2.3 5.2.4	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures.	345444546
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south	 345 44 45 46 47
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface.	 345 44 45 46 47 48
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south	 345 44 45 46 47 48 49
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground.	 345 44 45 46 47 48 49 50
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground. part of possible winch system at Kettleness	 345 44 45 46 47 48 49 50 51
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9 5.2.9	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground. part of possible winch system at Kettleness Anti tank blocks in Sandsend beck beneath the remains of a former	 345 44 45 46 47 48 49 50 51 52
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9 5.2.10	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground. part of possible winch system at Kettleness Anti tank blocks in Sandsend beck beneath the remains of a former briek built railway viaduct	 345 44 45 46 47 48 49 50 51 52
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9 5.2.10 5.2.11	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground. part of possible winch system at Kettleness Anti tank blocks in Sandsend beck beneath the remains of a former brick-built railway viaduct Two views of Pillbox 515 showing how it has slumped further onto the beach at Sandsend in 4 months	 345 44 45 46 47 48 49 50 51 52 53
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9 5.2.10 5.2.11	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground. part of possible winch system at Kettleness Anti tank blocks in Sandsend beck beneath the remains of a former brick-built railway viaduct Two views of Pillbox 515 showing how it has slumped further onto the beach at Sandsend in 4 months.	 345 44 45 46 47 48 49 50 51 52 53 54
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9 5.2.10 5.2.11 5.2.11	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground. part of possible winch system at Kettleness Anti tank blocks in Sandsend beck beneath the remains of a former brick-built railway viaduct Two views of Pillbox 515 showing how it has slumped further onto the beach at Sandsend in 4 months. Previously unrecorded Vickers machine gun post at Sandsend.	 345 44 45 46 47 48 49 50 51 52 53 54 57
5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9 5.2.10 5.2.11 5.2.12 5.2.12 5.2.13	Overdale Wyke Site Overdale Wyke from the south showing the location of the cliff top enclosures Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures. Goldsbrough Roman signal station looking south Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface. Tunnel mouth at Kettleness end looking south Pier base at Sandsend Beck looking south with an <i>in situ</i> WWII anti tank cube in the foreground. part of possible winch system at Kettleness Anti tank blocks in Sandsend beck beneath the remains of a former brick-built railway viaduct Two views of Pillbox 515 showing how it has slumped further onto the beach at Sandsend in 4 months. Previously unrecorded Vickers machine gun post at Sandsend. The cliffs of Sandsend alum works showing the extent of inundation by the tide together with the direct and on going erosive effects of the sea.	 345 44 45 46 47 48 49 50 51 52 53 54 55

5.2.15	Features Identified by the survey of Overdale Wyke	58
5.3.1	Archaeology Recorded at Hummersea and Saltburn	346
5.3.2	Hummersea Site Location	60
5.3.3	Remains of alum house eroding out of the cliff at Hummersea.	60
5.3.4	View of Skinningrove harbour taken from the north.	61
5.3.5	Possible cliff top location of Roman signal station recorded by	62
	antiquarian excavation.	
5.3.6	Fan house at Skinningrove ironstone mine	63
5.3.7	Remains of Hummersea alum works slumping and eroding	64
5.3.8	Archaeological remains of Saltburn alum works eroding.	65
5.3.9	Surveying structures eroding from the Alum works at Loftus	65
5.3.10	Extant remains of settling tanks	66
5.3.11	Extant remains of large rock-cut features on the foreshore at	67
0.0111	Hummersea.	01
5.3.12	One of the smaller rock cut post holes at Loftus	67
5313	Surviving extent of stone alum pier at Loftus	68
5314	Recording one of the numerous rutway between Saltburn and	68
5.5.11	Skinningrove	00
5315	Slumped pillbox on the beach north of Skinningrove harbour	69
5 3 16	The same slumped pillbox as above more exposed to wave action. This	69
5.5.10	photograph was taken two months after fig 5.3.15 and demonstrates the	07
	changeable nature of the beach deposits and environmental conditions	
5317	A preserved weapons pit on the western end of Skinningrove pier	70
5 3 18	Archaeological features recorded by NERCZA Dase 2at Skippingrove	73
5.5.10	Hummersee	15
5310	Archaeological features recorded by NERCZA Dhase 2at Salthurn	74
5.5.17	Skinningrove	/+
5/11	Archaeology recorded at Greatham Greek	347
542	Creatham Creak: Site Location	76
5.4.2	Stanton type air rold shelter at Creatham Creak, part of former	75
5.4.5	bombing descy	15
511	Congrator house for hombing decountry the Stanton shelter in	77
J. 1 . 1	background	11
515	Bacogling one of the Saltern mounds, modified in the cold war to	79
5.4.5	house a Revel Observer Corporatorion	70
516	Bomment of the realement of landscape press much to the west of North	70
5.4.0	Core	19
547	Gare.	00
3.4./	Former railway embankment reused as a platform for a pliloox and	80
E 4 0	Section post.	00
3.4.8	Partially visible remains of Snipwreck (158) on the foreshore at North	80
F 4 0	Gare. Define the first set of the first	01
5.4.9	Partly buried section post (145) showing their distinctive chevron shape	81
F 4 4 0		00
5.4.10	The remains of pillboxes used to reinforce the North Gare pier	82
5.4.11	remains of loop holes in concrete wall fragments from demolished	82
=	pillbox	0.0
5.4.12	Fortlet 152 overlooking Greatham Creek made up of two large section	83
F 4 4 2	posts.	0.1
5.4.13	Spigot mortar base 141 partially protruding from track edge.	84
5.4.14	Possible remains of demolished pillbox 142.	85
5.4.15	Recording remains of a surviving anti-glider trench 182.	86

 5.4.17 Remains of barbed wire and picket wire at North Gare. 5.4.18 Remains of a WWI seaplane base (203) at North Gare. 5.4.19 Remains of sealed hatch into ROC Cold War monitoring post. 5.4.20 Archaeological sites recorded by NERCZA phase II at Greatham Creek / North Gare 5.4.21 Recorded archaeological features at the WW1 seaplane base found at North Gare 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	 87 88 89 91 92 348 94 93 95 96 97 98
 5.4.18 Remains of a WWI seaplane base (203) at North Gare. 5.4.19 Remains of sealed hatch into ROC Cold War monitoring post. 5.4.20 Archaeological sites recorded by NERCZA phase II at Greatham Creek / North Gare 5.4.21 Recorded archaeological features at the WW1 seaplane base found at North Gare 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	 88 89 91 92 348 94 93 95 96 97 98
 5.4.19 Remains of sealed hatch into ROC Cold War monitoring post. 5.4.20 Archaeological sites recorded by NERCZA phase II at Greatham Creek / North Gare 5.4.21 Recorded archaeological features at the WW1 seaplane base found at North Gare 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	 89 91 92 348 94 93 95 96 97 98
 5.4.20 Archaeological sites recorded by NERCZA phase II at Greatham Creek / North Gare 5.4.21 Recorded archaeological features at the WW1 seaplane base found at North Gare 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	 91 92 348 94 93 95 96 97 98
 / North Gare 5.4.21 Recorded archaeological features at the WW1 seaplane base found at North Gare 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	92 348 94 93 95 96 97 98
 5.4.21 Recorded archaeological features at the WW1 seaplane base found at North Gare 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	 92 348 94 93 95 96 97 98
 North Gare 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	348 94 93 95 96 97 98
 5.5.1 Archaeology recorded at Crimdon Dene 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	 348 94 93 95 96 97 98
 5.5.2 Crimdon Dene Site Location 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	94 93 95 96 97 98
 5.5.3 Crimdon Dene in full flow in winter. 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	93 95 96 97 98
 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	95 96 97 98
 the effects of ongoing erosion. 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	96 97 98
 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	96 97 98
 Raistrick and Coupland. 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	97 98
 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	97 98
 Dene. 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	98
 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	98
 to be the land on the right of the picture. 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	
 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park. 5.5.9 The Blackhall colliery site. 	
furrow at Crimdon Dene Caravan park.5.5.9 The Blackhall colliery site.	99
5.5.9 The Blackhall colliery site.	
	100
5.5.10 Anti-tank battery at Crimdon Dene, showing access on the roof.	101
5.5.11 Pillbox (82) built to defend the anti-tank battery (83).	101
5.5.12 One of the exposed pillboxes (106) on the beach at Crimdon Dene.	102
5.5.13 Anti tanks block utilised as a foundation for a modern bridge and	103
Cache of anti tank cubes removed from their original location.	
5.5.14 eroding unidentified military remains at Crimdon Dene	104
5.5.15 Archaeological features recorded at Crimdon Dene by NERCZA phase	106
Π	
5.6.1 Archaeology recorded at Trow Point	349
5.6.2 Trow Point Site Location	108
5.6.3 The site of the experimental late Victorian "disappearing gun" on Trow	107
point.	
5.6.4 Truncated sub-circular earthwork, the only surviving candidate for a	110
possible barrow on Trow point.	
5.6.5 The Location of the disappearing gun at Trow Point.	111
5.6.6 Rock Cut steps at Trow Point.	112
5.6.7 The position and depth of one of the weapons pits with one of the	113
WW2 gun positions visible in the foreground.	
5.6.8 Recording one of the trenches on Trow Point	113
5.6.9 One of the two gun mountings 122 and 123 on Trow Point.	114
5.6.10 One of the eroding reinforced gun positions (129) overlooking the	115
beach at Trow Point.	
5.6.11 The gun position 139 on the beach having fallen from the top of the	115
cliff on to the beach.	
5.6.12 The surviving overgrown earthwork remains of Frenchman's Battery	116
south of the view looking north towards Trow Quarry	
5.6.13 The cliff directly underneath the gun emplacement at Trow Point.	117
5.6.14 Archaeological features recorded by NERCZA Phase II at Trow Point	
	118

5.7.2	Roberts Battery Site Location	120
5.7.3	The former Officers Quarters and Observation Post at Robert's	119
	Battery.	
5.7.4	View south to St Mary's Island from the Observation Post surviving on	121
	top of Fort House	
5.7.5	Earthwork remains of medieval ridge-and-furrow ploughing.	122
5.7.6	The WW1 water tower and store rooms underneath.	123
5.7.7	The defended latrine block (440) with blocked up loopholes now	123
01111	visible.	
5.7.8	Exterior of Fort House looking north.	124
5.7.9	Internal features indicating the higher status of the Officer's Mess and a	124
	view of the main corridor.	
5.7.10	The serving hatch, showing the thickness of the interior walls.	125
5.7.11	Wartime map visible inside a modern cupboard. The back wall is the	125
	original wall.	
5.7.12	View north from the Observation Tower.	126
5.7.13	Hatch leading into basement level with concrete stairs.	126
5.7.14	The basement level.	127
5.7.15	Possible ventilation pipe 437 against the inside of the south wall.	127
5716	The original plans and elevations of Fort House as held by the owner	128
5717	Eroding trench (444) on the cliff edge with Fort House in the	129
5.1.11	background	12/
5718	One of the three earthwork mounds surviving on the firing range	129
5719	The concrete bases of one of the removed radio masts	130
5720	Circular pillbox (505) on Sandy Island, partially buried in sand	130
5 7 21	Circular pillbox (506) showing brick-blocked aperture	131
5722	Location of Robert's Battery collapsed structures and earthwork	131
0.1.22	remains.	1.51
5.7.23	The eroding cliffs at Hartley with St Mary's Island and lighthouse in the	133
	background.	
5.7.24	Archaeological features recorded at Robert's Battery	134
5.7.25	Archaeological features recorded at Seaton Sluice	135
5.8.1	Archaeology recorded at Druridge Bay (North)	351
5.8.1a	Archaeology recorded at Druridge Bay (South)	352
5.8.2	Druridge Bay Site Location	137
5.8.3	Druridge Bay viewed from Cresswell looking north towards Low	136
5.0.5	Hauxley	150
5.8.4	The dune and till cliff at Low Hauxley.	138
5.8.5	Part of the original Preceptory at Low Chibburn, with Second World	139
	War alterations to windows.	
5.8.6	One of post-medieval enclosures for a small house at Cresswell.	140
5.8.7	The later farmstead at Low Chibburn	140
589	A line of 13 anti-tank blocks exposed in the mouth of a small burn	141
5.8.10	One of the observation posts. Note the large blocked entrance and	142
5.0.10	aperture at the front, distinguishing it from a pillbox.	1 12
5.8.11	Disguised pillbox located north of Blakemoor Farm.	142
5.8.12	Earthwork remains of military camp at Cresswell	143
5813	Possible base for a range finder or searchlight	144
5814	Part of the bombing range marker foundation (201)	144
5815	Pillbox exposed in dune in central Druridge Bay	145
5.8.16	Froding military structure	146
5.0.10	Lioung millary surveiule.	1 10

5.8.17	Druridge Bay looking south from Low Hauxley	147
5.8.18	Archaeological features recorded at the North end of Druridge Bay	148
5.8.19	Archaeological features recorded at southern end of Druridge Bay	149
5.9.1	Low Hauxley Site Location	152
5.9.2	The small cist containing a human cremation that also lay under	154
	Bonsall's Cairn 1.	
5.9.3	Human footprints filled with sand and pebbles deeply impressed within	156
	the peat that survives as a thin layer within the inter-tidal zone.	
5.9.4	Examples of some of the animal footprints that can be seen amongst	157
0.001	the human footprints.	101
5.9.5	Location of the Type and Wear interventions at Low Hauxley	157
5.9.6	Mapping the visible peat at Low Hauxley	158
5.9.7	Recording the extent of sandstone wall (360)	159
598	One of the square rock cut features at Low Hauxley	160
599	Anti-tank blocks used as part of the outflow of the Bondicarr Burn	161
5910	A heavily eroded anti-tank block at Low Hauxley	162
5911	Section of picket wire found eroded onto the beach	163
5912	Archaeological features recorded at Low Hauxley	166
5 10 1	Archaeology recorded at Alpmouth	353
5.10.2	Alpmouth Site Location	168
5.10.2	Recording the location of one of the ovster beds	167
5.10.5	Modern cross on top of Church Hill overlooking Alnmouth In the	169
5.10.4	foreground is the foundation of a former heyagonal pillbox	107
5 10 5	The enclosure bank at Beacon Hill	170
5.10.5	The termination of the boundary (218)	170
5.10.0	Rempart medieval ridge and furrow still clearly visible in a field behind	172
5.10.7	the post-medieval wall	1/2
5 10 8	The post medieval chapel at Church Hill	173
5.10.0	The extant Ovster Beds	174
5 10 10	The chapel and hollow way viewed from the east	174
5 10 11	The dedication stope on Alnmouth gup battery	175
5 10 12	Alamouth battery showing original features and later loop holes	176
5 10 13	Former guopo storage shed south of the River Alp	176
5 10 14	A Second World War loophole placed in the wall of the post medieval	170
5.10.14	mapo shed	1//
5 10 15	Erromentary remains of a military building possibly disquised south of	178
5.10.15	the River Aln	170
5 10 16	Archaeological features recorded at Alpmouth	180
5.10.17	Archaeological features of the Alpmouth Oyster ponds	181
5 11 1	Archaeology recorded at Amble	354
5 11 2	Amble Site Location	183
5.11.2	One of the larger hulks visible at Amble	182
5.11.5	The rudder of vessel (352)	185
5 11 5	The hold of the largest vessel (352)	105
5.11.5	Examining the details of carried built bulk (352)	105
5.11.0	Vessel 355 laving against the harbour well (453)	100
5.11.7	One of the smaller, prosumphy later, fishing yessels	107
J.11.0 5 11 0	The bulks as they appear at low tide	10/ 100
J.11.9 5 11 10	Archaeological features recorded surrounding the Amble huller	100
5 1 2 1	Archaeology recorded at Scremerston	355
J.12.1 5 1 2 2	Scrementon Site Location	100
J.1Z.Z	SCIENCESION SHE LOCATION	194

5.12.3	The field containing the crop marks of the Scremerston enclosure, demonstration proximity to railway cutting and coast .	191
5.12.4	The field containing the crop marks. No surface evidence of the enclosure was visible.	194
5.12.5	The former railway cutting, showing the modern route of the line behind the stone wall.	195
5126	Kiln 458 on the cliff edge is being eroded	195
5 1 2 7	The Transmitter/Receiver block at the Scremerston radar station	196
5.12.8	Gun hattery 465 south of Scremerston	197
5.12.9	Eroding concrete structure built utilising concrete sandbags, possibly of Second World War date.	197
5.12.10	The foreshore and cliffs to the south of Scremerston	198
5.12.11	Archaeological features recorded at Scremerston North	199
5 12 12	Archaeological features recorded at Scremerston South	200
5 1 3 1	Fenham Site Location	202
5132	One of the cliff face slumps below the Fenham enclosure	201
5133	The field containing the crop marks of a enclosure at Fenham	203
5.13.4	The view from the centre of the enclosure looking south	203
5.13.5	View north from the enclosure at Eenham Ranging pole marks the	204
5.15.5	contro of the surviving romains	205
5.13.6	The cliff faces below the enclosure. The ranging pole is roughly parallel	205
F 4 0 7	to the centre of the enclosure.	201
5.13.7	The earthwork remains of the manor and grange at Fenham.	206
5.13.8	One of the two shooting butts on the foreshore.	207
5.13.9	The remains of a small quarry and associated kiln preserved as earthworks north of Fenham	207
5.13.10	Picket wire and barbed wire (not the fence in the foreground) exposed on the cliff edge at Fenham.	208
5.13.11	Anti-tank blocks at Fenham with a small brick built shooting butt or defensive wall clearly visible.	209
5.13.12	Row of anti-tank blocks north of Fenham	209
51313	The anti-glider poles on Goswick Sands and Fenham Flats	$\frac{-32}{210}$
51314	Archaeological features recorded by NERCZA Phase II at Fenham	212
5 14 1	Archaeology recorded at Nessend	356
5 1 4 2	Nessend Site Location	214
5.14.3	The earthwork and structural remains of the chapel on St Cuthbert's Isle	213
5 1 1 1	Part of the exposed area of the lithic scatter	216
5.14.5	One of the early medieval long houses at the Green Shiels settlement	210
5.14.6	A stone hearth surviving in the centre of one of the long houses at Crean Shiels	217
5117	The aroding section of the change (388) on St Cuthhort's Isla	210
5.14.7	The main standing remains of the changel with Lindiaforms Drivery	210
5.14.0	visible in the background.	219
5.14.9	The earthwork remains of a drain for the chapel and associated building.	219
5.14.10	Possible earthwork remains of a medieval or later longhouse.	220
5.14.11	The impressive Lime Kilns (403) below Lindisfarne Castle	221
5.14.12	The wooden and stone pier (398) south of Lindisfarne castle	221
5.14.13	The holloway (399) leading to pier (398)	222
5.14.14	Two lime kilns preserved within the quarry (410)	222

5.14.15	The stone built fire step or loading bay and raised bank for the lime kilns.	223
5.14.16	The keep of the Tudor fort (396), overlooking the harbour	224
5 14 17	Partially upstanding gable end of Longhouse (409)	224
5 14 18	Archaeological features recorded by NERCZA Phase II on Holy Island	226
5 1 5 1	Archaeological features recorded by NERCZA Phase II at Bamburgh	357
5 1 5 2	Bamburgh Castle viewed from the village looking east	228
5 1 5 2	Bamburgh Site Logation	220
5.15.5		227
5.15.4	The beach and dunes with Bamburgh Castle in the background, looking south.	229
5.15.5	One of the possible burial mounds at the Bowl Hole.	230
5.15.6	A hexagonal pillbox in the dunes at Bamburgh.	231
5 1 5 7	The front aperture of one of the pillboxes	232
5.15.8	The rear entrance of the same pillbox	232
5 1 5 0	The front of the 6 inch gun employement (370)	232
5.15.10	The same employeement looking head on at the sup aparture	233
5.15.10	The same emplacement looking field on at the gun aperture.	233
5.15.11	The concrete mounting for a 6 inch gun which would have been	234
	overlain by a metal mounting for the gun to swivel on.	
5.15.12	One of the brick lined trenches $(3/6)$ found around Bamburgh.	235
5.15.13	Anti-tank blocks painted as dice to the north of Bamburgh Castle.	235
5.15.14	Anti-tank blocks exposed on the beach at Bamburgh.	237
5.16.1	Pillbox constructed of concrete sandbags south of Dunstan Steads.	239
5.16.2	Extent of survey within the Northumberland coast area of outstanding	240
	natural beauty showing key locations marked in the text	
5.16.3	Pillbox located in dunes at Dunstanburgh	242
5.16.4	Pillbox constructed of concrete sandbags south of Dunstan Steads	243
5.16.5	Distribution of surviving pillboxes within the Northumberland Coast	244
	Area of Outstanding natural beauty showing key locations marked in	
	the text	
5.16.6	Schematic plans of pillbox types found in Northumberland	245
5.16.7	Weapons pit preserved as an earthwork, directly south of	246
	Dunstanburgh castle.	
5.16.8	Recording a Second World War Fire Trench, preserved as an earthwork	247
	north of Dunstanburgh Castle	
5.16.9	Distribution of surviving weapons pits and trenches within the	248
	Northumberland coast area of outstanding natural beauty showing key	
	locations marked in the text	
5.16.10	Distribution of surviving anti-tank blocks within the Northumberland	250
	coast area of outstanding natural beauty showing key locations marked	
	in the text	
5.16.11	Anti-tank blocks on the beach directly north of the mouth of the River	251
	Cognet	
5 16 12	Coastal battery gun emplacement covering the potential landing beach	252
5.10.12	at Coswick (463)	252
5 16 12	Budle bettern (500), clearly demonstrating cools of construction	252
5.10.15	Marking and analytic strain and in dama and a f Dada 1	200
5.10.14	wachine gun emplacement exposed in dunes south of Beadnell.	200
5.16.15	Surviving buildings at Craster Kadar Station, later used to house	254
.	Prisoners of War.	a = -
5.16.16	Plan and elevations of Craster coast defence/ chain home low	255
	transmitter/ receiver block	
5.16.17	The surviving remains of the former runway at RAF Boulmer looking	256

	north-west from the road.	
5.16.18	Plan of the Developments of RAF Boulmer and its Environs	257
5.16.19	Air Raid Shelter (629) in the churchvard at Boulmer. The sunken	258
	entrances are still visible at both ends.	
5.16.20	Stanton air raid shelter in garden of the Old Rectory, Howick.	259
5.16.21	Destroyed pillbox (672) in the dunes between Seahouses and	260
	Bamburgh.	
5.16.22	Concrete of Second World War date eroding out of the cliff base on the	260
	beach near Boulmer.	
5.16.23	Probable pillbox eroded onto the beach near Seahouses; there are many	261
	such examples along this stretch of coast.	
5.16.24	Evidence of the former minefield between Dunstanburgh and	262
	Beadnell,. This is one of several craters created by the removal of the	
	mines	
5.16.25	Concrete base for winding gear for the quarry 2km to the south.	263
5.16.26	The track for the quarry leading to the winding gear base at Budle	264
5.16.27	Distribution of all surviving sites within the Northumberland coast area	265
	of outstanding natural beauty showing key locations marked in the text	
5.16.28	Distribution of previously recorded site within the Northumberland	266
	coast area of outstanding natural beauty showing key locations marked	
	in the text	
5.16.29	SMP2 projected loss for AONB (North)	267
5.16.30	SMP2 projected loss for AONB (South)	268
5.16.31	Distribution of all surviving remains within the Northumberland coast	269
	area of outstanding natural beauty showing key locations marked in the	
	text	
5.16.32	Extent of survey within the Northumberland coast AONB	270
5.17.1	The Whitburn fish trap as seen on the ground from the west.	271
5.17.2	Second World War pillbox at the First World War seaplane base at	272
	North Gare, Teesside.	
5.17.3	The quarry at Loftus Alum works, North Yorkshire, viewed from	273
	above.	a= (
6.1	Band of exposed peat exposed in the cliff face at Low Hauxley	274
6.2	Excavation and recording of a test pit at Crimdon Dene, County	275
<i>(</i>)	Durham.	070
6.3	Low Hauxley and Hartlepool Bay radio carbon results	2/8
6.4 < 5	Crimdon Dene radio carbon results	279
0.5	Small bay east of Victoria Harbour, where samples were collected	279
0.0 (7	Location of samples recovered from Hartlepool	282
0./	Crimdon Dene viewed looking south	283
0.8	Cliff from the next sector of Directed from Crimdon Dene	285
0.9	collected	200
6 10	Area of shall middon at Low Houvelov	207
6.11	Location of samples collected at Low Hauwley	207
7.1	Location of previous excavation at Low Hauxley eroding from the cliff	303
7.1	Location of samples at Low Hauxley	305
73	The briefly exposed footprints at Low Hauxley	305
7.4	Actively eroding remains at St Cuthbert's Isle Northumberland	308
75	Location of St Cuthbert's Isle Northumberland	309
7.6	Erosion of the late prehistoric enclosure at Fenham Northumberland	310
1.0	Erosion of the face premisione enclosure at remain, Northumberland	510

7.7	Location of the late prehistoric enclosure at Fenham, Northumberland	312
7.8	Gun emplacement at Budle Bay, Northumberland	313
7.9	Features surrounding the gun emplacement at Budle Bay	315
	Northumberland	
7.10	Trow Point viewed from the West	316
7.11	Location of archaeological features at Trow Point	317
7.12	Three of the Amble hulks, located in the River Coquet	319
7.13	Location of the Amble hulks	320
7.14	Location of the Scremerston late prehistoric enclosure, viewed from the	322
	North	
7.15	The location of the late prehistoric enclosure at Scremerston	324
7.16	Second world war gun emplacement at Scremeerston	325
7.17	Location of archaeological features surrounding the gun emplacement	327
	at Scremerston	
7.18	Budle bay fish traps viewed from the south	328
7.19	Slipway of First World War seaplane base at North Gare	329
7.20	Location of First World War seaplane base at North Gare	331
7.21	Location of Robert's Battery looking south	332
7.22	Location of archaeological features surrounding Robert's Battery	333
7.23	Area of exposed clay containing lithic material at Nessend, Holy Island	335
7.24	Location of Nessend lithic scatter	336
7.25	Rapid recording of an eroding pillbox at Warkworth Northumberland	338

List of tables

4.1	Example of completed data table attached to each recorded feature	48
5.2	North East Shoreline Management Plan 2 policies for Management	69
	Areas 21 and 22 (Sandsend to Kettleness)	
5.3	Shoreline Management Policy 2 policies for Saltburn and Skinningrove	85
5.4	Shoreline Management Plan 2 policies for the Greatham Creek area	103
5.5	Shoreline Management Plan 2 policies for the area of Crimdon Dene	118
5.6	The SMP 2 policy for Trow point and its environs	130
5.7	The Shoreline Management Plan 2 policy information for Seaton Sluice	146
	and Robert's Battery	
5.8	Shoreline Management Plan 2 policy plans for the Druridge Bay area	160
5.9.1	Summary of radiocarbon dates from the various archaeological	168
	interventions at Low Hauxley	
5.9.2	Rates of coastal erosion taken from SMP2 report (Guthrie et al. 2009,	176
	190)	
5.10	Shoreline Management Plan 2 polices for Alnmouth area	191
5.11	The Shoreline Management Plan 2 policies for the area the hulks lie in	200
5.12	The Shoreline Management Plan 2 policy for Scremerston	210
5.13	Shoreline Management Plan 2 Policy for the area of Fenham	222
5.14	Shoreline Management Plan policy for the area of Holy Island	238
5.15	Shoreline Management Plan 2 policy for Bamburgh	249
6.1	Radio carbon results from Hartlepool Bay	280
6.2	Radio carbon results from Crimdon Dene	283
6.3	Radio carbon results from Low Hauxley	288
7.1	Prioritised list of threatened heritage assets on the North East coast of	295
	England based on the results of the NERCZA Project.	
7.2	Key to colours used in table 7.1	302

EXECUTIVE SUMMARY

This document reports on the Phase 2 Rapid Coastal Zone Assessment for North-East England based on the methodology outlined in version 4 of 'A Brief for Rapid Coastal Zone Assessment Surveys' (English Heritage 2005). The project was undertaken by Archaeological Research Services Ltd for English Heritage with the project data fed back into the HERs of the various local authority partners. These included the North York Moors National Park Authority, North Yorkshire County Council, Tees Archaeology (for Redcar and Cleveland, Middlesbrough, Stockton-on-Tees and Hartlepool), Durham County Council, Tyne and Wear Specialist Conservation Team (for South Tyneside and North Tyneside) and Northumberland County Council.

The project aimed to rapidly survey all extant archaeological features visible at each of ten survey locations identified by Phase 1 of the project and survey and sample inter-tidal peat deposits at an additional four locations. The results of this survey and sampling can then be utilised to enhance the existing HER and NMR record and provide a useful tool to feed back to the Shoreline Management Plans and better inform the management of the archaeology of coastal margin in future. The results and outputs of the NERCZA project can also be used to help local communities better engage with their coastal heritage in future.

During the course of this project some unexpected discoveries were made including most notably the identification and recording of burials falling out of the cliff face at Low Hauxley and an area of human and animal footprints impressed within a previously unknown inter-tidal peat also at Low Hauxley. Both of these sites were drawn to the attention of ARS Ltd staff by a local amateur archaeologist. A separate report has been prepared for the excavation of the two burials. The footprints and associated peat are discussed in detail in Chapters 5.10 and 6.4 of this report. A further separate condition report that summarises the results of previous archaeological interventions at Low Hauxley and future management options for the site has also been produced as part of this project. This report forms the main project report and contains the results of the field survey that formed the bulk of this phase of the NERCZA project.

The project has delivered the following outputs:

- Updated Phase 1 report and Executive Summary document in the light of the new SMP2
- Enhancement of six local authority HERs
- A project GIS that contains surveys, photographs and records of threatened sites and inter-tidal peat
- Phase 2 project report (this document)
- A standalone Phase 2 Executive Summary
- Talks to local societies (e.g. Druridge Bay Liaison Group, Architectural and Archaeological society of Durham and Northumberland)
- Articles for local magazines and short academic article (e.g. Low Hauxley newsletter, archaeology in Northumberland, Archaeology County Durham, and Journal of the Fortress Study Group)
- Forthcoming academic article for IfA journal The Historic Environment: Policy and Practice
- Excavation and report on a rescue excavation at Low Hauxley
- A review of archaeological interventions and site condition at Low Hauxley
- An A4 fold-out leaflet
- A report on the rapid survey of WWII remains within the Northumberland Coast AONB

The Phase 2 survey identified and surveyed 609 features and added 135 new records to the HERs of Northumberland, Teesside, North Yorkshire Moors National Park and County Durham. Any repetition of existing or known sites within the HER provided significant enhancement to the current

record due to the detailed current condition statement provided as part of the survey. A total of 5 extant peat beds were mapped at Low Hauxley and all of these were radiocarbon dated as well as samples taken from organic layers observed in the inter-tidal zone at Hartlepool Bay and Crimdon Dene in County Durham, the results of which have been fed into English Heritage's national inter-tidal peat database.

The project area comprised a strip of land from the lowest astronomical tide (LAT) level to 1 km in-land form Mean High Water Springs (MHWS). The study area ran from Whitby in the south to the Anglo-Scottish border in the North, an area of approximately 200km that falls within SMP2 for Northumberland and SMP2 for North East England which replaces SMP Cell 1.

Phase 2 of the NERCZA carried out field survey of fifteen threatened areas between September 2009 and February 2010. Each site was investigated and surveyed utilizing mapping grade Global Positioning System (GPS) equipment, with extensive digital and paper records maintained along with extensive site photography. A digital photo archive of 2733 JPEG images was amassed during the project, as well as additional detailed imagery of eroding peat layers at Low Hauxley, Northumberland. Upon completion of the initial fieldwork more than 500 sites had been recorded, of which 115 were new records and the remainder provided significant enhancement to existing records. Along with this a further 100 Second World War sites have been identified within the Northumberland Coast Area of Outstanding Natural Beauty as part of a separate survey project which focussed only on the remains surviving from the Second World War. This project was undertaken on behalf of the Northumberland Coast AONB and undertaken after the completion of the NERCZA Phase 2 fieldwork. The same methodology was employed by Archaeological Research Services Ltd in the course of this survey during February 2010. This additional work provided an additional 20 new records. The data from this survey has been added to the project GIS along with the photographic and paper archive, increasing the scope of the NERCZA Phase 2 field survey. This gave a total of 609 records for the project with 135 of these being new records not previously incorporated into the regional HER data.

At least 75% of sites recorded at these locations related to the defence of the coastline during the Second World War. Pillboxes, gun emplacements, anti-tank defences, and observation posts were all observed and recorded. A great number of these sites had been known previously and recognised on wartime aerial photography; however some sites were new discoveries. The remainder of previously known sites recorded by the survey provided significantly enhanced records. The level of survival of earthwork military remains, such as trenches and weapons pits, as well as remains of mine fields was particularly notable. The temporary nature of these structures make the large number identified during the fieldwork surprising and potentially important, as many are preserved in a thin strip of land between the retreating coastline and inland development. Earthwork remains from the Second World War represent some of the most commonly encountered archaeology along the North East Coast and also some of the most exposed to marine transgression and development. Concentrations of Second World War military earthworks were identified at Bamburgh, Dunstanburgh, Boulmer, and Beadnell in Northumberland and at Greatham Creek and North Gare on Teesside.

Results from the survey of these sites show that many sites that have been previously identified which were thought to have disappeared are still extant in some form. For example a pillbox recorded by Phase 1 that has been thought to have been removed or eroded away may still survive as earthworks or as buried foundations partially visible. Other remains identified as at risk range from the Mesolithic period to the Cold War and all now have an assessment of the level of threat that they face, along with accurate positional information to within 0.5m, a significant improvement for many records. This information base compiled into the project GIS will form a useful tool for land management, local planners and help to adapt the Shoreline Management Plans to take into account management of high risk archaeological assets and better identify those of significant special interest. Specific site by site recommendations on management can now be made using SMP2 as a reference point and the archaeological and environmental evidence from NERCZA Phase 2 can be used to support or challenge the current policy based on the threat to any heritage assets. This assessment has been undertaken by the project team following subjective scoring of the 70 sites of special interest on 5 key criteria; threat, condition, significance, potential and rarity. This produced a ranked list of sites. The top threatened sites of special interest were then evaluated and possible management strategies proposed for each and these are discussed further in Chapter 7 of this report. This assessment does not provide definitive answers for management but clear proposals have been made in order to provide a basis for future discussion of the management of coastal heritage assets.

1. INTRODUCTION

1.1 **Project Outline**

This report documents the Phase 2 of the Rapid Coastal Zone Assessment for the North East Coast of England from Whitby in the south to the Anglo-Scottish border just north of Berwick upon Tweed in the north. Phase 1 comprised an aerial photographic transcription and desk based assessment of the same area, which commenced in March 2007 and was completed in December 2008 (Tolan-Smith 2008) and updated in the light of the publication of Shoreline Management Plan 2 in early 2009, with Phase 2 of the project commencing in May 2009.

During the course of the desk based assessment and aerial photography transcription exercise 968 new records were added to North East HERs and 270 existing records were enhanced. Phase 1 of the project also identified numerous archaeological sites in the study area which are facing imminent threat from natural processes such as coastal erosion and rising sea levels (Chapter 10 of Phase 1 project report). Further damage could be caused to such sites by the construction of sea defences as a result of the recommendations from the Shoreline Management Plans (SMP2 for Northumberland and the North East) which aim to manage such threats. The archaeological sites under threat were identified as being in urgent need of rapid ground surveys and recommendations and prioritisation for their future management and conservation. The Phase 1 survey initially highlighted ten areas for further rapid field survey which were undertaken as part of Phase 2, together with five additional locations that were added as further areas of interest or threatened sites were identified and surveyed opportunistically. In addition, to this rapid survey work four locations were selected for survey and sampling of inter-tidal 'peat' layers.

The rapid field survey examined the fifteen highlighted locations in greater detail, recording the visible archaeological remains and taking field notes (Fig 1.1). All of these locations are at risk from some form of ongoing erosion, whether immediately or in the long term, and the specific site reports include assessments of the level of threat to historic assets at each survey location visited.

In Phase 2 of the project rapid field survey recorded 609 archaeological features at the fifteen locations, approximately 30% of which were new records or rediscovered features thought to no longer exist. All records collected by NERCZA Phase 2 contained detailed condition statements in the form of attached data tables. These contained data on threat, condition, site type, period and coastal setting and this information can now be used to significantly enhance any existing records in the HER and NMR. Newly identified archaeological features included prehistoric human and animal footprints at Low Hauxley, surviving graded earthwork remains of a prehistoric enclosure at Fenham and numerous Second World War military sites all along the coast. Approximately 75% of the features recorded by the Phase 2 fieldwork dated to the Second World War. This can be accounted for due to the excellent survival of many earthwork remains that were recorded for the first time and the addition of data from the Northumberland Coast AONB military survey project data, also undertaken by Archaeological Research Services Ltd using the same methodology as NERCZA.



Fig 1.1 Remains of Tudor period fort overlooking Holy Island harbour, with Lindisfarne castle in the background

The detailed location reports summarise the archaeological features recorded at each site, discuss and evaluate the threats that they face with consideration of current SMP2 policy. This information is complied in section 5 of this report. This section covers each of the fifteen sites surveyed and a description of the Second World War military sites recorded as part of the Northumberland Coast AONB military survey undertaken by Archaeological Research Services Ltd are also included.

The palaeoenvironmental sampling element of the project was undertaken at five key sites identified by Phase1. Suitable samples with enough material for radio carbon dating were obtained from three of these. Low Hauxley (Fig 1.2), Crimdon Dene, and Hartlepool Bay all successfully yielded enough material for dating and pollen analysis. The results of this are discussed further in Section 6 of this report.



Fig 1.2 Investigating a layer of inter-tidal peat at Low Hauxley, Northumberland

Chapter 7 of this report deals with the key management issues relating to each of the locations surveyed. This was done using an onsite assessment of threat by the project team, considering coastal erosion, potential for future flooding and land use. The assessment was undertaken in a subjective manner and is intended only to propose possible strategies for further discussion in the future. The outcomes of this are not intended to provide definitive proposals for what should be done to manage these sites for the future. The assessment of threat to each site was used together with an assessment of special interest using criteria outlined in English Heritage's guidance for Scheduled Ancient Monuments (formerly Annexe 4 of PPG 16). This allowed each site to be given a score out of fifty and ranked accordingly. The results of these risk assessments were tabulated and the most threatened sites of special interest are discussed in greater detail in Chapter 7 of this report. This process enabled proposals for the management of the archaeological resource at each site to be put forward for discussion by the project team and these are also discussed in Chapter 7.

The project has produced this full report, along with a non-technical executive summary document outlining all of the sites recorded and possible management options. This will be produced alongside the project GIS which has allowed dissemination of the information recorded at each of the 609 sites to the Historic Environment Records. Also the project archive of 2,773 digital images along with an additional 694 non archived supplementary photos has been made available for the use of the regional Historic Environment Records.

In addition to the Phase 2 project report and the executive summary document NERCZA has produced supplementary reports for the excavation at Low Hauxley (Waddington 2009), the Northumberland coast AONB military survey (Burn 2010) and a summary of previous work at Low Hauxley (Waddington 2010). Specialist reports concerning worked wood, palaeoenvironmental samples and radiocarbon dates have also been produced. All of these reports will form a part of the project archive along with the paper records and field notebooks. All data collected in the field as a part of these subsidiary projects has been included in the NERCZA project GIS.



Fig 1.3 Remains of a chapel on St Cuthbert's Isle, seen eroding as a result of wave action.

A full list of the 609 sites recorded by the rapid field survey can be found as a separate gazetteer document accompanying this report and are also summarised separately in the updated project executive summary. These records from the field survey form the core data recorded by the project and can be directly referenced with the archived photo records also included in the gazetteer.

2. AIMS AND OBJECTIVES

2.1 Project Aims

This Phase 2 of NERCZA is primarily focused around rapid field survey which aims to substantiate and complement the results of the desk-based study undertaken as Phase 1 of this project. This survey focused on areas identified in Phase 1, especially in the inter-tidal zone where historic assets were identified as being under threat. The overarching aims of this project are as follows:

- Provide further heritage information which can be fed directly into Defra's latest Shoreline and Estuary Management programme, thereby helping to ensure appropriate protection, or mitigation of damage, to historic assets.
- Provide further enhancement and additional information to the HERs and NMR record of coastal heritage assets. This will enable an improved curatorial response to strategic coastal planning or management initiatives at both a national and regional level
- Enhance the factual evidence-base for the curatorial response to individual applications in advance of developments or coastal protection schemes.
- Provide further information on the likely archaeological potential and vulnerability of the coast.

2.2 Specific Objectives

2.3.1 Upgrade the Phase 1 report in the light of the completion of SMP2, the draft of which has recently been completed by Royal Haskoning for the area north of the Tyne.

2.3.2 Provide more detail on archaeological sites under threat within the study area which will be feed into Defra's Shoreline and Estuary Management programme, the NMR, the various HER of the various project partners and recommendations to EH of sites to consider for designation.

2.3.3 Verify site identifications made during Phase 1 of the Project, particularly those identified as part of the aerial photograph transcription work.

2.3.4 Locate and characterise sites and features undetected during Phase 1 of the Project.

2.3.5 Determine the geomorphological/sedimentary context for features.

2.3.6 Assess whether features are eroding.

2.3.7 Selectively sample features with particular attention to the inter-tidal peat layers to ascertain their extent and date range.

2.3.8 Identify sites in urgent need of additional recording, dating or characterisation work at specific sites to take place after the completion of the Phase 2 survey.

2.3.9 Test fieldwork methodologies and assess the practicalities and logistics of future fieldwork including any required mitigation measures and/or required additional recording, dating or characterisation work at specific sites to take place after the Phase 2 survey.

2.3.10 Produce data that can be used to inform the North East Regional Research Framework (NERRF) and the Yorkshire Research Framework (YRF).2.3.11 Provide data that will be of assistance to other coastal managers, other coastal surveys and researchers.

2.3.12 Increase the understanding of the archaeology of the North East coast amongst the public and the research community through varied dissemination.

2.4 **Project Integration**

As an adjunct to this project a ground survey of the military archaeological remains surviving within the Northumberland Coast Area of Outstanding Natural Beauty was also undertaken. This has provided additional data for this project and has provided further assistance in enhancing the Northumberland HER, as well as the NMR. A separate standalone report has been produced for this sub-project (Burn 2010).

A further additional sub-project was also undertaken during the course of this project. This comprised a detailed review of the archaeology, condition and history of previous interventions and location of archives for the eroding Mesolithic and Early Bronze Age site at Low Hauxley. This site has a complex history and this site-specific study has assisted in focusing in on the key management issues and options for the future. A separate standalone report has been produced for this sub-project (Waddington 2010).

2.4.1 SMP's and Conservation

The project has produced survey data that has enhanced the various HER's within the study area, which will enable an evidence-based response within the planning process. The project has provided heritage information that has been made available to Defra's Shoreline and Estuary Management Programme (SMP2) which will help to ensure protection and management and/or mitigation of damage by natural processes. The project has also informed other bodies, such as the Northumberland Coast AONB and Durham Heritage Coast, and has contributed to the aims set out in their respective Management Plans.

2.4.2 Research Frameworks

The project has contributed detailed condition assessments for all of the features recorded. Also updated or proposed alternative interpretations for previously recorded features were included within the field survey. These new archaeological records, include military features such as weapons pits and trenches, rock cut features on the foreshore, industrial remains and human and animal footprints preserved in exposed inter-tidal peat. The expansion of the number of records and improvement of existing records provided by the project has contributed significantly to the regional research framework. In particular the area in the research framework of Defence and Fortification has been contributed to by the project.

Maritime and coastal archaeology (Petts and Gerrard 2006) has also been contributed to by the recording and assessment of features such as the Amble hulks and rock cut fish traps by Phase 2 of the NERCZA project. The increased understanding of the Bronze Age and earlier archaeology at the site of Low Hauxley that has come as a result of the NERCZA Phase 2 investigation has also contributed to the relevant section of the research framework.

2.4.3 Future Research Possibilities

During Phase 2 of the project further avenues have been explored for a HLF project given the rejection by Northumberland County Council to running such a project through the Northumberland Coast AONB on 'purchasing' grounds resulting from EU legislation. A project proposal has been made to the

LEADER and also to the EH regional office for a local capacity building grant to assist local communities on the North-East Coast to engage with their built and natural heritage on the coast and to monitor impacts over the medium term. This project takes the form of a proposed evaluation followed by full excavation at Low Hauxley with community involvement, allowing local communities to further engage with their coastal heritage. This project will involve the provision of archaeological training opportunities, installation of interpretative material and educational and access opportunities for young people and communities. It will have the added potential to fit directly into the Coal and Coast project proposed by the Northumberland Wild Life Trust as well as meeting the needs of the regional research framework (see section 2.4.2).

Phase 2 of the NERCZA project identified several locations of which the site either requires further work to fully understand the nature of the archaeology or to preserve eroding remains through record. For example the site at Crimdon Dene has the potential to yield further information on the Mesolithic activity if further work is undertaken to relocate the lithic scatter on the County Durham coast, however further investigation work is required to fully understand the location and realise the full potential of the site. Also the site at St Cuthbert's Isle in Northumberland is fairly well understood archaeologically, however the imminent nature of the threat from erosion and visibly eroding archaeological deposits. Here further work is required to record rapidly eroding remains of potentially significant archaeological deposits.

All of the proposals for further work that have been identified by Phase 2 of this project are discussed further in Chapter 7. Proposals for further work have been based on the special interest of the site along with the immediate level of threat faced by the surviving archaeology. Using this evaluation process for each of the top sites of special interest identified in Chapter 7, three options for further work and management of these sites have been proposed. These are only proposals intended to promote discussion on the future management of these key sites and not a definitive guide for what must be done.

2.4.4 Project Outputs

Phase 2 of the NERCZA project has produced the following outputs as a direct result of the field survey work:

- Project Report (this document)
- Executive Summary document summarising cumulative results of Phase 1 and Phase 2 of the project (Burn and Johnson 2010)
- Low Hauxley Excavation report (Waddington 2009)
- Low Hauxley summary of archaeological interventions report (Waddington 2010)
- Northumberland Coast AONB survey of military archaeology report (Burn 2010)
- Integrated project GIS database containing 609 records plus additional records for sample locations and all data from phase I of the project
- Digital photographic archive of 2,773 JPEG images
- Approximately 135 new HER records based on the data used in Phase 1, although this does not account for HER records updated during the course of the NERCZA project (2008 -2010)

- 609 enhanced HER records as current survey has assessed threat level and condition for every site recorded
- Guided walk with Architectural and archaeological society of Durham and Northumberland, project update talks to NYM archaeology group and North East Maritime Forum.
- 2 articles to be published in 2010 in Archaeology in Northumberland, Archaeology County Durham, and 1 in 2011 in Casemate (Fortress study Group Journal). A third academic article is currently in preparation for the IfA journal; The Historic Environment: Policy and Practice.
- Project leaflet summarising results for distribution to the public and local authorities
- Proposed inset days for teachers from local schools in Northumberland, allowing them to engage further with coastal heritage. This has yet to be confirmed and finalised.

3. SCOPE OF THE SURVEY

3.1 Geographical Scope

The aim of the aerial survey mapping element of the project was to produce accurate mapping from aerial photographs and a record of all archaeological features from all periods that could be identified within the study area as part of an in-depth desk-based assessment. The Phase 2 field survey of the locations identified during Phase 1 aimed to ground truth and expand upon the archaeological features mapped by the aerial photography and enhance these surveys with metric survey data. The aerial photography transcription data was employed in the field during the course of the survey. It was used to help interpret newly identified archaeological features and to re-evaluate interpretations of those that had been previously recorded.

The project area extends across the following local authorities (from north to south): Northumberland County Council, Tyne and Wear Metropolitan Borough Council, Durham County Council, the unitary authorities of Hartlepool, Middlesbrough, Stockton-On-Tees, Redcar and Cleveland and North Yorkshire County Council. With the exception of Whitby, the majority of the project area that runs through North Yorkshire lies almost wholly within the North York Moors National Park Authority. The fifteen sites identified for further work subsequent to Phase 1 represent each of these local authority areas. The field survey focused on fifteen sites identified within the strip of land mapped in Phase 1, from the lowest astronomical tide (LAT) to a width of 1km in-land from the high tide level (MHWS) and running from the Anglo-Scottish border in the North to Whitby in the South (Figure 3.1). The project area falls over an area of 560km² but only c. 402km² extends over exposed land; the remainder falls within the inter-tidal zone. The field survey was carried out to Level 2 standard (Ainsworth et al 2007) and covered a length of 142km of this coastline, although this figure does not account for the coverage of foreshore, cliff top dunes and estuarine hinterland that was covered during the investigation.

In February 2010 an additional area was also surveyed and added to the project GIS as part of the sub-project, the Northumberland Coast AONB military archaeology survey. It was felt that Phase 2 of the NERCZA project would benefit from extra coverage in this area even though this sub-project was targeted at specifically Second World War military remains. This additional work was funded separately by the Northumberland Coast AONB but the results have now been integrated into the wider NERCZA project, thus enhancing the project GIS and providing important value-added benefits for this area of coastline.



3.2 Geology

The solid geology of the study area is reviewed in detail in Chapter 3 of the Phase 1 report (Tolan-smith 2008) and so does not need to be repeated here, but can be broadly summarised as follows. In County Durham, the principal rock exposed by processes of erosion is Magnesian limestone while to the north the Northumberland coast is more varied with localised exposures of limestone, sandstone and occasionally coal. Around Teesside and North Cleveland the

coastline is relatively featureless where the wide estuary of the Tees reaches the coast. However, to the south the coastline becomes the most dramatic of the study area with high, rugged cliffs around Saltburn and Boulby Head, in North Yorkshire where Liassic shales and sandstones are exposed (Kent 1980). Along the coast drift deposits, principally tills but also sand and gravels, cover the underlying bedrock. It often comprises a combination of clay resting on limestone or sandstone which meets the coast in the area under study (Beaumont 1970). In some areas the clay can reach considerable depths as, for example, at Whitley Bay, where the clay has a depth of 6m (Spratt 1979).

3.3 Archaeological Scope

3.3.1 Earthwork archaeology

All extant earthworks identified as archaeological in origin on aerial photographs were mapped as part of the Phase 1. Available RCHME /EH ground survey plans were used to assist and enhance this mapping. The data from Phase 1 was used on the GPS display in the field to identify previously known earthwork remains some of which were then re-interpreted. The field survey recorded using mapping grade GPS all earthwork remains encountered (Figure 3.2) in basic plan form utilising lines and polygons were appropriate as required by a level 2 survey (Ainsworth *et al* 2007). If earthwork sites had already been recorded as part of a recent detailed level 3 survey, for example Kettleness alum works, then they were recorded as a basic polygon with an attached condition statement rather than in detailed so that there was no repetition of survey work.



Fig 3.2 Graded extant earthwork remains of the buildings of a former medieval grange farm of Lindisfarne priory at Fenham looking south from the hamlet of Fenham

3.3.2 Levelled archaeology

All crop marks, soil marks and parch marks identified as archaeological in origin were mapped by Phase 1 and three crop mark sites in particular at Fenham, Scremmerston and Overdale Wyke were identified for field survey on the ground as part of Phase 2. The data collected from Phase 2 was used to precisely re locate the position of these sites to see if there were any extant features visible on the surface (Chapter 5.2). In addition to crop mark sites the flint scatters at Crimdon Dene and Ness end were also targeted for further investigation. The same methodology was undertaken using known positional information on the GPS with field reconnaissance to attempt to relocate the extent of these features.

3.3.3 Post medieval and modern field boundaries

Field boundaries that have been removed but are still extant, and depicted on first edition Ordnance Survey or later edition maps, were generally not surveyed by phase 2. This was unless they formed a key component of another significant archaeological feature that was identified as under threat from erosion. An example of this can be seen in the possible medieval or post medieval plough headlands identified on the cliff edge at Saltburn (Chapter 5.3)

3.3.4 Medieval and post medieval ridge and furrow

Ridge and furrow (Figure 3.3)was mapped using the GPS were it was either newly recognised, identified as being under threat or presented an exceptional example of preservation. It was also recorded if there was clear or imminent threat of erosion or the site could be used as a case study for management such as the graded ridge and furrow seen at Alnmouth (Chapter 5.). The ridge and furrow fields were recorded as polygons with the GPS using a simple graphical depiction, delineating the extent of area and direction of the furrows.



Fig 3.3 Crop marks visible on the ground revealing the trace of medieval broad ridge and furrow at Overdale Wyke, north Yorkshire, view looking West.

3.3.5 Industrial features and extraction

Large and small-scale quarries were mapped with the GPS and recorded as polygons, irrespective if they were depicted on any Ordnance Survey map or within the Phase 1 data. Detail was then picked out within these polygons using

"dumb data" i.e. lines with no attached data tables. Coal mining and associated features, such as tramways, were mapped and recorded as with other features. Large collieries or open cast mining complexes were also mapped generally as an extent of area as with those seen at Blackhall as there was no scope within this project for undertaking further detailed survey.

3.3.6 Post Medieval and 20th Century military features

Former Post Medieval, First and Second World War military sites and installations were mapped. Extensive military complexes and sites were outlined as an extent of area with descriptive data attached. Anti-landing obstacles and tank traps were recorded as lines to show their alignment. Surviving installations such as pill boxes and coastal gun/searchlight batteries were also mapped (Figure 3.4). As many sites of this period and function were by nature short lived and transitory emphasis was placed on the identification and general extent of activity when appropriate, rather than the accurate depiction of single features such as local track ways although in some cases this was also necessary to interpret the nature of the surviving remains. Significant features within these outlined areas were mapped either "as seen" or schematically as dumb data, according to the visible extent on the ground and the size of each feature. Where such remains were fragmentary or insubstantial, a single point was used to record their position.



Fig 3.4 Unusual double pillbox on the shore at Goswick, survey by the Northumberland coast AONB survey, linked by underground access, see section 5.16

3.3.7 Buildings

The foundations of buildings visible as earthworks, or ruined stonework were surveyed using the GPS, regardless of if they were depicted on first edition Ordnance Survey or later edition maps. Standing roofed or unroofed buildings or structures such as the Knights Hospitaller Preceptory at Low Chibburn (Figure 3.5) were also recorded if they had a particular association in the context of industrial or military remains identified by the field survey. Medieval castles and monastic sites previously recorded and extensively surveyed and mapped by the Ordnance Survey were generally already mapped by Phase 2 as an extent of area if they were relevant to the context of the site being investigated, e.g. Bamburgh castle and Lindisfarne Priory, and so were not recorded by Phase 2.



Fig 3.5 The Knights Hospitaller preceptory at Low Chibburn that became a dower house of the Widdrington Castle estate and was later re-used as a WWII pillbox (see Chapter 5.).

3.3.8 Geomorphological features or natural deposits

Geomorphological features when encountered in association with known archaeological deposits were recorded. For example the collapsing dune cliff seen at Crimdon Dene, was recorded in basic plan form as it directly threatened surviving military archaeology. Also any visible peat layers were recorded as part of the palaeoenvironmental sampling element of Phase 2. For example the extent of visible peat at Low Hauxley was recorded in plan form using the GPS and also a full photographic survey of the cliff face undertaken to serve as a future monitoring tool. The peat that contained human and animal footprints, possibly of prehistoric date also was recorded in plan form.

Where significant organic deposits such as these were identified in association to significant archaeological remains, a program of sampling was undertaken. This was the case most notably at Crimdon Dene and Low Hauxley but also at the harbour entrance at Hartlepool.

3.3.9 Parkland, landscaped parks, gardens and country houses

None of these features were encountered within the project area as part of the field survey.

3.3.10 Maritime Features

Ship wrecks and fish traps visible in the inter-tidal zones were recorded if visible on the foreshore (figure 3.6). They were fixed more accurately than Phase 1 data would allow utilising the GPS equipment.



Fig 3.6 Recording of a rock cut feature with the Magellan GPS equipment, possibly a fish trap or "hulley", on the foreshore at Hummersea, North Yorkshire (see Chapter 5).

4. METHODOLOGY AND RECORDING PRACTICE

4.1 Survey Methodology for field survey of threatened sites (Objectives 4.2.2-4.2.6, 4.2.8-4.2.12)

4.1.1 Introduction

Phase 1 of this project identified fourteen locations where coastal erosion is currently degrading heritage assets and to this was added a fifteenth at Whitburn that was subsequently identified as being in need of survey on account of reported past erosion. These are detailed in Chapter 10 of the Phase 1 study, whilst the detailed survey reports can be found below in Chapter 5 of this report. Each survey location had specific issues and reasons for instigating rapid survey and these area covered in more detail in the site descriptions contained in Chapter 5 of this report.

4.1.2 Summary of targeted sites

A summary of the sites targeted and surveyed as part of Phase 2 is listed below, from South to North:

- Overdale Wyke prehistoric enclosures, North Yorkshire.
- Hummersea and Saltburn alum works, North Yorkshire.
- Greatham Creek Second World War defence area, Teeside.
- Crimdon Dene Mesolithic flint scatter, County Durham.
- Roberts Battery, Seaton sluice, Northumberland.
- Druridge Bay Second World War defence area Northumberland.
- Low Hauxley Bronze Age cemetery and Mesolithic site, Northumberland.
- Amble 19th century hulks, Northumberland.
- Alnmouth oyster ponds, Northumberland.
- Bamburgh bowl hole, Northumberland.
- Fenham multivallate enclosure, Northumberland.
- Scremmerston multivallate enclosure, Northumberland.
- Nessend Mesolithic flint scatter, Holy Island, Northumberland.
- Northumberland Coast Area of Outstanding Natural Beauty. (surveyed as a separate Archaeological Research Services Ltd project recording all Second World War archaeology within the AONB, using the same methodology).

All the above locations were identified as being in urgent need of a field visit and rapid survey in order to assess the condition of the surviving remains and the extent to which these remains are exposed to on-going erosion. Any surviving remains that were then identified and recorded by the field survey were then assessed in terms of special interest and threat and this assessment has been used to put forward possible options for discussion of their future management (see Chapter 7). This was a subjective basement undertaken by the project team and intended to put forward initial ideas and allow discussion of the possible future management of these heritage assets.

4.1.3 Survey Methodology

The survey of archaeological remains involved surface identification of surviving features followed by rapid detailed recording. This entailed the digital

photography of the remains, along with extensive notes on nature and extent of survival, dimensions, interpretation, setting and additional environmental information. This information was also recorded digitally directly onto the GPS equipment as attached data for each record. This allowed direct download of field data into the project GIS without an extensive data entry exercise. Additional data was also recorded on pro forma recording sheets, although this was not always practical due to adverse weather conditions such as high wind, heavy rain and even snow making detailed paper recording impossible. In these situations the pro forma were abandoned in favour of a weather proof notebook, the data then transcribed onto pro forma at a later date. A weatherproof notebook was used at all times during the fieldwork to keep track of the numbering of features and photographs.

Each targeted survey location was expanded upon to add context to the archaeological records for the targeted sites. This provided a much wider assessment of the target locations and eventually covered 142km of coastline. Each site was broken down into manageable sections with the foreshore, cliff tops, dunes and estuarine locations all investigated. For example the survey would initially progress along the foreshore and then back along the cliff top or through the dunes to cover as much of the threatened area as possible.

The survey recording procedure adopted involved the use of a handheld DGPS unit and digital photographic equipment. The handheld DGPS unit, a Magellan MobileMapper CX with post-processing hardware kit, offers real-time sub-metre accuracy and sub-foot post-processed accuracy using MobileMapper Office, running on Microsoft Windows CE. NET 5.0. The equipment provides both vector and raster map support through Digiterra 5 software, including datasets in ESRI .shp file format as well as MapInfo and Autodesk file format support. Relevant information (e.g. AP transcriptions, OS base mapping) from the project GIS will be loaded onto the GPS unit to inform the fieldwork.

The direct entry of field data into the GPS unit in a format that could be directly downloaded into the project GIS had the added benefit of using drop down lists to select from for each column heading. These drop down lists were MIDAS compliant and used the INSCRIPTION wordlists. This meant that the data entry across the project was much more consistent no matter which surveyor was using the GPS as there was set parameters to choose from.

This allowed the use of Ordnance survey, NERCZA Phase 1, NMR, HER and SMP2 projected coastline data to be used in the field. This aided the interpretation and assessment of threat of each feature in the field as part of the survey process. The GPS was found to consistently offer accuracy within 0.4m without post processing, and the post processing software rarely improved on this level of accuracy. For this reason after the initial survey days the GPS was used in handheld mode only, and this subsequently increased productivity and allowed further length of coastline to be covered opportunistically by the project.



Fig 4.1Utilising the Magellan CX mapping grade GPS with external antenna for post processing, at Skinnigrove.

Data collected in the field was logged directly to the GPS unit in a data entry form format to collect data which is MIDAS Heritage standard compliant and uses the INSCRIPTION wordlists. This dataset is based on the compliance tables presented in section 4 of *MIDAS Heritage - a data standard for the historic environment* (English Heritage 2007) and was entered directly on the GPS into the following form:

Column heading	Data to be entered (Example)
UID	601
Site Type	Defence
Period	WW2
Description	Pillbox
Notes	Hexagonal pillbox, surviving mostly
	intact
Erosion Notes	eroding at base of cliff face
Coastal Setting	Embayment
Inter-tidal Setting	Sand Beach
Estuarine Setting	
Onshore Setting	Rock Cliffs
Geology	Till
Weather	Calm and Clear

Surveyor	AB
Photo	796
Level of risk	Imminent
Type of Natural erosion	Bedrock cliff retreat
Type of Artificial erosion	Other
Tidal level	Low
Easting	414858
Northing	636945

Table 4.1 Mocked up example of a completed data table attached to each recorded feature

The data has been downloaded from the GPS unit and integrated into the project GIS as a database. This now includes data from NERCZA Phase 1, HERs, NMR, Ordnance survey 1st edition coastline, SMP2 projected coastlines with preferred policy and No Active intervention for 20, 50 and 100 year periods and the Phase 2 survey data. Fieldwork also involved the completion of a written proforma (also MIDAS Heritage compliant and use the INSCRIPTION wordlists) which provide comprehensive information on each of features surveyed. The pro-forma no longer need to be digitised as this will duplicate the information on the digital form and supplementary sketches can be replaced by the survey data and hyper-linked digital photography.

4.1.4 Collection/Excavation Strategy

The aim of Phase 2 of the project was not to excavate features or collect artefacts. However, during the survey several artefacts were identified which would otherwise have been lost to erosion. These were collected and their precise location recorded. For example, flint, pottery and metal artefacts were all identified in positions were they would imminently be subject to erosion. A full 12 figure grid reference was recorded for each with the GPS as detailed above. Following completion of the project an archive including these finds will be deposited with the appropriate museum.19th/20th Century structural remains such as bricks/concrete were not collected but small and significant objects such as metal military artefacts, e.g. preserved picket wire, were collected. Brief reports and specialist's assessment reports have been included as an Appendix of this report.


Fig 4.2 Cleaning the buried organic land surface at Crimdon Dene, County Durham.

4.2 Survey methodology for field survey of peat shelves and 'submerged forests', or other organic deposits (Objectives 4.2.2, 4.2.7-4.2.11)

Inter-tidal peat has been identified at a number of locations along the NE coast, notably at Hartlepool Bay, Whitburn Bay, Cresswell and Low Hauxley. Crimdon Dene has subsequently been added to this list as organic layers were recognised exposed in the vicinity of a known lithic scatter. At Hartlepool these are associated with archaeological deposits dating from the Mesolithic, Neolithic and Romano-British periods (Waughman 2005) while artefacts and palaeofuanal remains have been recovered at the other localities. Access to these deposits is generally difficult and only possible at low tide. However, at Low Hauxley the peat deposits and land surface preserved beneath extend to the coastal cliffs below a later dune system from where Bronze Age burials continue to emerge. At Low Hauxley it was also possible to photograph the visible extent all the way along the cliff face from south of the Bondicarr burn to Low Hauxley itself and this photography forms part of the final photographic archive.

It was identified in Phase 1 that such deposits are particularly vulnerable to alterations in the wave regime that can be brought about by the construction of sea defences. In order to assess the threat to such deposits posed by various mitigation strategies their full extent was established, together with dating evidence and assessment of their archaeological and palaeoenvironmental potential.

The survey has recorded the aerial extent where possible, utilising the GPS, of the surviving peat. Using the same methodology as for the archaeological features the extent of the deposits, their condition of preservation, their potential to contain palaeoenvironmental resources and resistance to erosion were all recorded. Survey of the peat deposits listed above was undertaken in order to accurately identify their precise location and extent. The depth of the peat deposits will be measured by an auger survey although this only met with success at Hartlepool and Low Hauxley. Following consultation with the Regional Science Adviser, the technique employed was to establish the stratigraphy of the inter-tidal peat using hand-operated gouge augers which facilitated the rapid identification of the depth and character of sedimentary sequences.

Sampling and analysis of pollen/14C was then undertaken on representative sediment cores extracted from Hartlepool and Low Hauxley using a Russian peat corer. Works have been undertaken at Hartlepool (Waughman 2005) and at Low Hauxley (Drury et al 1995) but the precise extent and depth of the deposits were not recorded. The auger survey was undertaken in the form of regular spaced transects across areas of peat with the position of the auger points accurately recorded by GPS (see above). The survey has provided details of the depth of the exposed peat deposits that will provide a reference against which future measurements can be taken to confirm the rate at which the deposits are eroding and these are discussed further in Chapter 6 of this report. The basal and uppermost deposits have been examined by Charlotte O'Brian at Durham University who has identified the appropriate samples for dating and provide confirmation of the potential for the peat to provide information by determining the survivability of pollen *etc.* These samples were then sent to the English Heritage Scientific dating team for C14 dating of the peat layers.



Fig 4.3 Investigating peat layers at Low Hauxley Northumberland

The English Heritage Science Adviser has recommended that as part of the survey it would be advantageous to obtain radiocarbon dates for the uppermost and basal deposits of the peat beds as it is not known how synchronous they are. This has now been undertaken by the English Heritage Scientific dating team and



all information obtained has been supplied to the National database for inter-tidal and off-shore peat and is included in Chapter 6 of this report.

Fig 4.4 Undertaking sediment coring at Hartlepool Bay, Teeside.

4.3 Sample Walkover Survey of sites identified from aerial photographic survey in Phase 1 (Objectives 4.2.3, 4.2.4, 4.2.6, 4.2.8-4.2.11)

Phase 2 of the project was designed to be flexible to take into account difficulties of access to features such as the inter-tidal peat deposits. Alternative survey work was pinpointed in advance in order to avoid wasted survey days due to inaccessibility of proposed targets for survey. For example the identification of an additional site at Whitburn and the expansion of some survey areas where additional potential had been recognised was undertaken when the inaccessibility of peat layers at Cresswell and Whitburn originally proposed in the project design became an issue.

When time was available for alternative work a sample walkover survey was undertaken to identify the presence, nature and extent on the ground of features identified from aerial photographs. This led to natural expansion of each of the proposed survey areas in order to cover more of the overall coastline and surrounding environs of each site. This involved the investigation of a large number of World War 2 military features which represented 74% of all features recorded in the aerial photographic survey undertaken as part of Phase 1 of this project. Priority was given to the two WW2 defence areas within the study area at Greatham Creek and Druridge Bay and a representative sample of other features identified during Phase 1 were surveyed within the environs of each targeted survey location.

The survey was undertaken to the same level as outlined in Phase 1 and detailed proposals made for the future management of each site are made in the Chapter 7 of this report

4.4 Production of data from the surveys in a form compatible with HER and NMR database systems (Objectives 4.2.2, 4.2.11)

The data produced has now been incorporated into the existing project's GIS. Output is in ESRI shape files which have now been incorporated into all the project partners HERs following on from the outputs already delimited during Phase 1. HERs have been consulted on the incorporation of data prior to dissemination and some of the key sites surveyed discussed. The data will also be provided to the SMP consultants (Royal Haskoning) and Defra and allocated to the relevant Policy Unit for SMP2 for the North East and Northumberland. The NMR has been consulted (Martin Newman) and data will be provided to them in ESRI shape files in a form that will be both MIDAS and INSCRIPTION compliant.

4.5 Reporting (Objectives 4.2.2, 4.2.10-12)

This will take the form of this internally produced integrated report (including the various specialist reports) (A4 spiral bound or similar) which is provided on CD. The report will be distributed to EH, the project partners, Natural England, the National Trust, Royal Haskoning and Seazone and to consultants and developers upon request. The report (or a synthesis depending on available file space) has also been uploaded on to the OASIS system where it can be consulted on-line by the public and the project data has been archived with the ADS and incorporated within the various HERs.

This report contains a detailed account of the methods used and constraints experienced (see above). It summarises the principal results of survey, subdivided in terms of coastal Policy (Management) Units. It includes a preliminary assessment of the regional (and, where appropriate, national) significance of sites recorded (Chapter 5), and their vulnerability to erosion. It indicates areas meriting further survey, assessment, recording and monitoring (Chapter 7) and identifies sites, structures or buildings potentially meriting protective legislation. It includes an assessment of the archaeological potential of samples taken and artefacts collected, and their potential for further analysis (Chapter 6). The report also attempts to broadly classify the archaeological potential of the coast, consider the implications of the survey in terms of the relevant Shoreline Management Plans or strategy documents and include an executive summary suitable for circulation to non-archaeological coastal managers and planners (Chapter 7).

In addition to this report a detailed executive summary has been produced and will be provided to the SMP consultants, Natural England and the Coastal group. This has tabulated all records from Phase 2 (this output has already been

completed for Phase 1), sub-divided in terms of SMP2 policy units and includes appraisals of significance and vulnerability, defined by objective criteria as much as possible. This also includes much of the assessment included within Chapter 7 of this report.

4.6 Outreach (Objectives 4.2.10-4.2.12)

The reports now produced by this part of the NERCZA project will be available to the public and research institutions on-line via OASIS and the ADS. The conclusion of the project has also been marked by talks to organisations such as the North York Moors Archaeology Group, North East Maritime forum and a guided walk of Druridge Bay for the Architectural and Archaeological Society for Durham and Northumberland. Additional talks are proposed for key groups such as Natural England, Northumberland wildlife trust and other local groups as well as relevant county Archaeology Days, learned and local societies and other useful/appropriate conferences as available. Local societies have been contacted as appropriate for further talks and the publication of 10000 A4 fold-out leaflets summarising the results and contribution of the project is at the draft stage at the time of writing. It is proposed that the latter will be distributed at county Archaeology Days, with county archaeological magazines and learned society newsletters, and at Tourist Information Centres along the North East coast, as well as direct mail shots to consultancies and internally within the various County Halls and English Heritage. This will raise awareness of the project and the availability of enhanced HER /NMR records and improved understanding of coastal heritage assets and their risk from erosion.

4.7 Copyright

All outputs will be the copyright of Archaeological Research Services Ltd and licence to use the data will be extended to English Heritage and the project partners.

The project partners will have unrestricted use of all aspects of the data produced by the project for the purposes of research, education and non-commercial publication.

4.8 **Project Archive**

On completion of the project all files created during the project will be copied to DVD and passed to the ADS who will apply for a separate archiving grant. A project summary will be uploaded to the OASIS system. The GIS will be placed on the computer system of the various SMR and HER together with backup copies on disk.

The results of this project will be archived with English Heritage's National Monuments Record (NMR) and the respective SMR/HER as appropriate. The air photographic datasets, for which English Heritage will have sole responsibility for the curation and archiving of, will consist of the NMP map data (layered AUTOCAD MAP drawing with attached data tables) and entries to the English Heritage's AMIE database.

5. ARCHAEOLOGICAL INVESTIGATION OF THE NORTH EAST COAST

5.1 Introduction

The following Chapter discusses each of the fifteen locations targeted for rapid field survey. In sections 5.2 - 5.15 each survey location will be discussed in terms of its landscape setting, topography, previous research, known history and land use. The visible remains are discussed broadly by period and specific case studies examined. The specific impact and nature of any erosion at each site is then considered in relation to specific archaeological remains, to allow quantification and assessment of specific threats.

During all of the fieldwork, data from the first phase of NERCZA was utilised to provide informed interpretations and aid the selection of targeted fieldwork locations. Phase 1 data was also used as a reference in the description of each of these survey locations to demonstrate both confirmed and updated interpretations of what had been recorded as part of Phase 1 compared to what was seen on the ground during Phase 2. The description and discussion of the fifteen survey locations considers each location separately from south to north. Supplementary locations that were visited as part of the project but not initially targeted after Phase 1 are discussed separately in the summary of this Chapter.

Section 5.17 deals with fieldwork data relating to the Second World War archaeology of the Northumberland Coast AONB recorded as part of a subsidiary project undertaken by Archaeological Research Services Ltd. This data was also incorporated into this project's GIS. A table summarising all recorded sites by period from all of the fieldwork undertaken is included in a separate gazetteer of sites. Implications and recommendations for future management of all the sites surveyed are discussed in the summary within each site report and in more depth in Chapter 7 of this report.



Figure 5.1.1 Surveying earthworks of Second World War defences preserved above Skiningrove Harbour

5.2 Overdale Wyke

5.2.1 Background

Phase I of the NERCZA project highlighted two possible prehistoric enclosures and a possible ploughed out barrow thought to be Bronze Age in date (Tolan-Smith 2008 84,). These are located at Overdale Wyke north of Sandsend in North Yorkshire within the North York Moors National Park. The aim of the survey at this location was to identify any surviving earthworks, the extent of erosion and the risk faced by the monument from increased rates of erosion, as well as the collection and recording of any artefacts in the vicinity (Waddington & Chatterton 2009, 13). The survey also aimed to identify any other archaeological features within the environs of the Overdale Wyke enclosures to provide additional contextual information and determine whether other as yet unknown sites are at risk.

The survey of Overdale Wyke did not reveal any surface evidence of the enclosures but did record many other features in the area surrounding Overdale, Kettleness and Sandsend. Many of these features where newly discovered while all of the records updated the current knowledge base and provided up to date condition statements for the features they related to.

5.2.2 Location and geology

Overdale Wyke (NU 85489 14278) (SMP PU 21.3 - 22.3) is located on a stretch of the North Yorkshire coast that includes some of the highest cliffs along the North East coast (Figure 5.2.2). The geology consists of Whitby Mudstone, with clay overburden and shale. The steep cliffs show the scars of the alum industry, which historically was prevalent in this area. The foreshore comprises a rock cut platform visible at low tide from Sandsend in the south to Runswick Bay in the north. The cliffs along this 5.1km length of surveyed coast rise to a height of 97m from the sandy beach at Sandsend. The cliffs are susceptible to erosion from ongoing slumping after under cutting by wave action and several significant erosion events have been recorded recently (SMP2). Current land use includes small fishing and tourist villages of Sandsend and Runswick Bay with mostly arable agricultural land and small farm settlements and hamlets. The land that the enclosures sit within is currently owned and managed by the Mulgrave Estate and seems to have been left fallow for sometime to encourage grouse.





Figure 5.2.3 Overdale Wyke from the south showing the location of the cliff top enclosures

5.2.3 Previous research

The NERCZA Phase 1 study looked at this part of the coastline as part of block 1 of the study area (Tolan-Smith 2008, 84). Although a holistic approach was taken looking at all archaeological elements, the key sites identified were the industrial remains of the former alum industry around Sandsend and Kettleness and the enclosures, surviving as crop marks, at Overdale Wyke. It is these sites that were targeted for rapid survey.

The alum industry of the North Yorkshire coast has been investigated in detail by English Heritage (Jecock and Hunt 2005). Alum works at Kettleness and at Loftus have both been subjected to detailed analytical field survey and historical research by the English Heritage Archaeological Survey Team. These detailed level 3 (Ainsworth *et al* 2007) surveys produced accurate plans of the complex earthworks that survive at these industrial sites. The history of the alum industry was also investigated, specific to each site as part of these investigations.

5.2.4 NERCZA Phase 2 Archaeological Investigation

The archaeological survey carried out as part of this study quickly established that the dense vegetation cover in the vicinity of the Overdale Wyke enclosures would obscure any surviving earthwork remains. Furthermore, the landscape having been heavily ploughed and improved in the past would suggest that there was little likelihood of upstanding earthwork remains surviving. If there were surviving remains they would now be heavily truncated and only visible as very subtle features. In addition, any surviving remains would be obscured due to the current level of vegetation cover. Utilisation of available Lidar data does not provide coverage of this area, high quality 0.5m resolution Lidar could indicate the presence of earthworks, except heavily graded ones.



Figure 5.2.4 Overgrown Vegetation obscuring the location of the Overdale Wyke prehistoric enclosures.

Having established that the survey of the enclosures themselves was impractical due to vegetation cover, the boundaries of the survey area were extended. The Overdale Wyke survey therefore included a much wider area and this has provided much more detail on what coastal archaeology survives resulting in a good information base for its future management. The survey covered the entire 5.8km cliff top from the beach at Sandsend to the alum works at Kettleness (Fig 5.2.14). Survey work was also undertaken further inland and on small areas of foreshore where access was safe at low tide (Fig 5.2.14).

The survey revealed archaeology dating from a range of periods including prehistoric and the Second World War and highlighted the industrial development of this part of the coast in the post-medieval period. The following sections will discuss the features identified and broadly characterise this part of the coast by period.

5.2.5 Prehistoric

The reason for targeting this location for rapid survey was to learn more about the survival of the Overdale Wyke enclosures identified on aerial photography, however as outlined above full detailed inspection on the ground was impossible due to dense vegetation cover. This said, the extensive ploughing in the vicinity makes the survival of visible earthwork remains extremely unlikely. Despite this set back upon investigating the surrounding area a struck flint (record number 430) was recovered from a ploughed field just to the west of the enclosures (Fig 5.2.14). The project does not include full field walking within its scope, but this chance find was accurately recorded and suggests the possibility of further positive results from any future programme of field walking in the vicinity of the enclosures. This one flint alone does not provide evidence of widespread prehistoric activity and could be merely residual. This evidence combined with previous investigation undertaken as part of the North East Yorkshire Mesolithic project (Grahame *et al* 2008, Waughman *et al* 2006)) in this vicinity indicates that further field walking on the coastal margin may yield more worked lithic material. This could provide additional context to the prehistoric activity in the area and possibly the enclosures located at Overdale Wyke.

5.2.6 Romano British

The Goldsbrough Roman signal station (429) (Fig 5.2.4) is a well known monument along this part of the North Yorkshire coast. It is currently a scheduled ancient monument (SAM no 32476). It has been subject to much previous work including antiquarian and later excavations which revealed it to be similar in nature to the completely excavated example at Scarborough Castle (Pearson 2009). The rapid survey recorded its condition to provide a complete picture of the archaeology of this coastline. The site is surrounded by post medieval ridge and furrow ploughing, although the monument itself does not appear to have been over ploughed.



Figure 5.2.5 Earthwork remains of Goldsbrough Roman signal station on a high point east of Goldsbrough village, viewed looking south

5.2.7 Early Medieval

No features were observed or recorded dating to this period.

5.2.8 Medieval

The most obvious medieval feature still visible within the landscape is evidence of ridge and furrow agriculture that can be seen as crop marks on aerial photographs (Fig 5.2.5). These crop marks can be seen around Kettleness and Goldsbrough broken up with later post-medieval ploughing. Evidence of ridge and furrow was extensively recorded by the Phase 1 survey. Although recognised by the Phase 2 fieldwork there was no time to record large areas of ridge and furrow ploughing due to the rapid nature of the survey. Also, ridge and furrow survives only as crop-marks in this area and not as positive earthworks, detracting further from their importance as an archaeological monument on account of their poor survival.



Figure 5.2.6 Medieval ridge and furrow visible as subtle crop marks on an otherwise even field surface.

5.2.9 Post-Medieval

During the post-medieval period this stretch of coastline was exploited by the alum industry. This was an important and profitable industry that produced fixative for dyes for the textile industry and which can be described as the first chemical industry in Britain (Jecock and Hunt 2005). There were several large alum works along the North Yorkshire coast around Kettleness (425). One of these complexes lies 2.6km to the north of the Overdale Wyke enclosures while another large alum quarry (415) lies immediately to the south. The alum quarries have left large scars in the cliff faces of this part of coast and the lunar landscape they have created as a result is easily recognisable. Again, because recent detailed work has been undertaken by English Heritage the quarries were not surveyed in detail as part of this project although a rapid walkover survey was undertaken.

What the survey did record was the surviving infrastructure surrounding the alum quarries, most significantly the rail link from Whitby to Redcar which

survives in part as an earthwork embankment (427) and cutting (413). The railway also survives as a tunnel between Overdale Wyke and Kettleness and part of the original route that was subsequently abandoned still survives south of Kettleness (422). This original cliff edge route was not completed and was ultimately abandoned due to the instability of the cliffs and risk of subsidence. This led to the construction of the tunnel (421) (Fig 5.2.6) and diversion of the route inland.



Figure 5.2.7 Tunnel mouth (422) at Kettleness end looking south.

Later in the nineteenth century the industrial railway developed into a passenger line and two station buildings survive, one at Sandsend (432) and one at Kettleness, along with platforms and outbuildings. Along the former track bed some fragments of sleeper survive along with voids where others would have been. At Sandsend the railway continued south to Whitby across a large viaduct, which has now gone, however the pier bases are still visible in Sandsend beck and on the beach (Fig 5.2.7).



Figure 5.2.8 Pier base for a now removed 19th century railway viaduct at Sandsend Beck looking south with an *in situ* WWII anti tank cube in the foreground.

Several other ephemeral industrial features were also recorded surrounding the alum quarry at Kettleness (Fig 5.2.8). Some of these may relate to a later exploratory stone mine opened in the 20th century, although others may relate to the alum works themselves. A large metal pole, possibly part of a winch system but currently of unknown date or function, is an example of such a feature.



Figure 5.2.9 part of possible winch system at Kettleness (scale = 2m).

5.2.10 20th Century

The 20th century archaeology is perhaps the most obvious and easy to identify at this location, the defence of the coast during the Second World War had an especially profound impact that can still be seen today at many locations. The most visible WWII remains can be seen at Sandsend itself with anti tank defences (434, 433) visible at the mouth of Sandsend Beck (Fig 5.2.8). These anti-tank blocks were placed deliberately to defend against possible incursion of tanks and armour up the creeks in the event of an invasion. The ant-tank blocks are visible at low tide and are intermittently covered with sand by the Sandsend Beck, which obscures them from view (Fig 5.2.9).



Figure 5.2.10 Anti-tank blocks in Sandsend beck beneath the remains of a former brick-built railway viaduct.

Also visible on the beach at Sandsend are the remains of a pillbox (515), evidently slumped from its original cliff top position, on to the beach (Fig 5.2.10). It has clearly been destabilised by ongoing erosion and now lies slumped on the beach exposed to wave action. During the course of the survey further slumping was observed between September 2009 and January 2010, the pillbox had moved a further 1.2m down slope according to the GPS position. This variation is accurate to within 0.3m so there has been at least 0.5m of slumping since the pillbox was first observed during the fieldwork.

The defences around Sandsend were placed to prevent a successful landing on the wide sweeping beach which would have provided access inland for heavy armour and enemy troops during WWII. This is paralleled on many beaches up and down the North East Coast which were suitable for a landing, most notably Druridge Bay in Northumberland which is discussed in detail in section 5.9.



January 2010 Figure 5.2.11 Two views of Pillbox 515 showing how it has slumped further onto the beach at Sandsend in 4 months.

Along the cliffs heading north from Sandsend towards Kettleness there is further evidence of WWII defences. An example of this is a camp recorded from aerial photography as part of Phase 1. This location was visited as part of the Phase 2 fieldwork. It had previously been thought that there would be no trace of the camp surviving as it comprised temporary structures and was relatively small in size. This conclusion had been reached as no evidence was visible on the latest aerial photography. However, several elements of the camp survive including heavily truncated earthworks together with some faint crop marks which delineate the extent of the camp (417), and 1940s brickwork (416) can be seen in the make up of the track, within the camp extents.

The most visible element of the camp is the remains of a Vickers twin anti aircraft machine gun post (420) (Fig 5.2.11), which survives intact with the exception of the gun itself. The brick weapons pit, metal frame and springs all survive. Evidence of this exceptional survival demonstrates that there is still potential for more subtle WWII remains, such as trenches and weapons pits, to survive intact further along this stretch of coast.



Figure 5.2.12 Previously unrecorded Vickers machine gun post at Sandsend.

5.2.11 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.26m per year (SMP2 for north East England). The base of the cliffs along this part of the coast are exposed to direct wave action during high tides. This has been observed at Sandsend in particular where there is clear evidence of undercutting and destabilisation of the mudstone cliffs. This has caused several episodes of slumping, for example the one described in section 5.1.10 and shown in figure 5.1.10 where the whole foundation of a pillbox (515) has collapsed and fallen on to the beach below. This is the common threat faced by this whole section of coast due to the direct action of the sea upon the base of the cliff.

The development of the alum industry along this particular piece of coastline has also contributed to the destabilisation of the tops of the cliffs. Large quarries and test pits along with the associated infrastructure, and notably the creation of the original and ultimately abandoned rail link between Whitby and Redcar in the mid 19th century, have all contributed to the destabilisation of the cliff edge.

Utilising the SMP2 preferred policy predictions as a guide it can be seen that several heritage assets are under long term threat in this area (Fig 5.1.14). The predicted retreat due to sea level rise and increased erosion from storm events can be seen to encroach on to 90% of the newly recorded features in this location within the next 100 years. It must be emphasised that the projected retreat is only a prediction which will be subject to change due to the variable factors that affect sea level change and coastal erosion.



Figure 5.2.13 The cliffs of Sandsend alum works showing the extent of inundation by the tide together with the direct and on going erosive effects of the sea.

Based on these predictions most of the archaeological features recorded during the Phase 2 project will not be directly affected for at least 50 years. This is not to say there is no current significant threat from erosion, as there is clearly an ongoing issue with land slumps and direct wave action at the base of the cliff. However, it is important to note that not all sites within this survey area are under the same level of threat. By combining Phase 1 interpretations and the field data collected by Phase 2 and with the SMP2 datasets, the level of threat that each individual recorded asset currently faces can be estimated. This can be done to project threat in the immediate future and also in the long term, to see what future threats heritage assets may face.

5.2.12 Summary and conclusions

Overdale Wyke presents an example of soft mudstone cliffs being undermined by direct erosion. This has led to increased instability of the cliffs along the coast which has caused some archaeological assets to slump from their original positions on to the beach. On the beach these assets are under further threat form direct wave action and more vigorous erosion. The erosion of these cliffs is an ongoing problem that is only going to become worse when considering possible sea level rise and the resultant and more frequent storm events. This more severe erosion will eventually cause increased destabilisation of the cliff and

an increase in the rate of erosion. The precise causes of this erosion have been studied in detail by the Boulby Geoscience Research Group (Steadman *et al* forthcoming), from Durham University, using advanced technical monitoring to research long term causes of cliff erosion (see http://www.boulbygeoscience.org/).



Figure 5.2.14 A rapidly eroding Alum pier at Sandsend that is exposed to direct wave action.

Some of the causes of erosion in this area would seem to stem from the historic land use of the coastal cliffs. Alum quarrying, along with the construction of other industrial structures, and notably the original route of the Whitby – Redcar branch railway, has contributed significantly to cliff top erosion. This has destabilised the cliffs and encouraged the slumping initiated by the natural processes. However the threat from the sea alone can be seen by the extent of change in a very short period of the location of the pillbox (515) at Sandsend. This may be a type of monument which is common along the coast and relatively recent in date, but it represents some of the most threatened archaeology along this part of the coastline.

As stated in section 5.2.14 the same level of threat cannot be applied to the whole stretch of coastline, therefore the survey has attached a level of threat (measured from 1 - 10), agreed by the project team in the field, for each recorded asset. The complete list of these can be found in the gazetteer of sites produced as a separate document. This data can then be used to identify an average level of threat for all of the assets for this particular stretch of coastline. This will then allow an assessment of risk to heritage assets for each stretch of coast to be undertaken by aggregating the monuments at each survey location together. This process is undertaken and described in detail in Chapter 7.

Management		Policy	Policy			
Area		unit	Plan			Comment
			2025	2055	2105	
	Runswick Bay to					
MA21	Sandsend Ness	21.3	NAI	NAI	NAI	
						Consideration of works
						associated with the unit to
MA22	Sandsend Wyke	22.1	NAI	NAI	NAI	the east
MA22	Sandsend Wyke	22.2	HTL	HTL	HTL	

Table 5.2 North East Shoreline Management Plan 2 policies for Management Areas 21 and 22 (Sandsend to Kettleness).

The Shoreline Management Plan (SMP2) (Table 5.2) for this stretch of coastline recognises the importance of the alum industry at both Sandsend and Kettleness. However, the prehistoric enclosures at Overdale Wyke and surviving military archaeology are not directly referred to. Despite this information the coastline between Kettleness alum works and Sandsesnd village is designated as No Active Intervention for the next 95 years. This means that these heritage assets should be prioritised for any further recording work as they will continue to be lost in coming years.Sandsend village itself is unsurprisingly designated Hold the Line due to the settlement and infrastructure there. This also effectively protects the archaeological remains of industrial and military origin that are present within the settlement. However, a predicted baseline erosion rate of 0.1m a year at Kettleness and at Sandsend cliffs and 0.25m (SMP 2) at Sandsend village indicates that erosion is ongoing along this entire stretch of coastline. By comparing the projected loss of coastline for 2025, 2055 and 2105 the immediate threat to sites such as the Overdale Wyke enclosures can be estimated (fig 5.2.14).

By looking at this projected data we can see that the easternmost enclosure at Overdale Wyke is under threat, eroding within the nest 20 - 50 years. This also applies to the alum quarries (415 and 426) and military remains (419, 420, arc431, 414) along this stretch of coastline. This projected risk from erosion over the next 20 - 50 years, combined with the lack of above ground evidence of the enclosures at Overdale Wyke, means that there is scope for further work. Further investigation of the enclosures such as geophysical survey together with full level 3 earthwork surveys and field walking if the field containing the enclosures was stripped of vegetation could be usefully undertaken to assess the significance and survival of these heritage assets. Limited excavation would provide further information to inform both condition of survival and characterisation of the site as well as helping in establishing the significance of the site. Once such work has been undertaken then a long term plan for dealing with this threatened site can be implemented.



5.3 Saltburn, Hummersea and Skinningrove

5.3.1 Background

The North East Rapid Coastal Zone Assessment Phase 1 identified the remains of alum works in the cliff face at Hummersea as being at particular risk from erosion. These remains are currently 8m above sea level. the main surviving structures have been engulfed by a landslip from the cliff above an what is visible exposed in the cliff face is of a very fragmentary nature. All the North Yorkshire alum works were situated close to cliffs and several have been reduced by cliff collapse and landslip. English Heritage has undertaken detailed photogrammetric recording of the remains exposed in the cliff face at Loftus alum works (near Skinnigrove). However, beneath the cliff at Loftus, on the foreshore, there are a series of docks and piles of both burnt and unburnt shale which were all once part of the alum industry. These features are located in a very hostile coastal environment and under imminent threat of erosion. The English Heritage survey did not deal with the foreshore and there is an urgent need for a survey to record these threatened features before they are removed for good. Tees Archaeology manages a local group of active volunteers who are keen to be involved in such projects and it is intended to work with the group and foster the continued monitoring of these historical assets in the future.

There are other industrial remains in the area surrounding Hummersea including the remains of rut ways and other rock cut features probably relating to the 19th century alum works between Saltburn and Loftus. Also visible in the area surrounding Hummersea are several features surviving from the Second World War including pillboxes, trenches and weapons pits. Given the vulnerability of this stretch of coast the rapid survey investigated an area 4km north and 2km south of the Hummersea alum works in order to gain a wider perspective on how this is affecting surviving archaeology.



Figure 5.3.3. Remains of alum house exposed in the cliff face at Hummersea.



5.3.2 Location and geology

The alum works at Hummersea are located on cliffs between Saltburn and Sandsend (NGR NZ 72658 19821) (SMP PU15.4 -17.3). This part of the North Yorkshire coast is located on geology of Whitby Mudstone, with clay overburden and shale. The steep cliffs show the scars of the alum industry which historically was prevalent in this area. The foreshore comprises a rock cut platform with sand beaches at Hummersea Bay, Skinningrove and Saltburn.

Current land use in the area is agricultural with some small industrial development at Skinningrove. There is also small-scale fishing industry activity while the settlement at Saltburn is an early 20th century seaside resort. Historically, the main activities in the area would have been industrial related to the alum and ironstone works and fishing at the numerous small bays and harbours.



Figure 5.3.4 A view of Skinningrove harbour taken from the north, the pier of the ironstone works is visible extending into the North Sea.

5.3.3 Previous research

Tees archaeology manage a group of local volunteers who have been systematically recording the evidence of rutways visible on the rock cut platforms of the foreshore between Saltburn and Skinningrove. This project, known as the Saltburn rutways survey, has also recorded other features such as postholes and a possible fish trap (Green 2008, 2009). In preparation of reports in 2007, 2008 and 2009 background research into the alum industries has been undertaken. English Heritage has undertaken detailed survey of the alum works at Loftus as part of their ongoing study of the Yorkshire alum works. This highlighted the key buildings and earthwork elements that survive within the alum quarry. Also, key elements of what remains on the foreshore where identified including areas of burnt shale and the stone built alum pier. Several articles and publications on the alum industry have also covered the area surrounding Hummersea and been published in local journals including in the 'Cleveland Industrial Archaeologist', however none of these investigations has presented a complete survey of the archaeology or included detailed threat assessments.

5.3.4 NERCZA Phase 2 Archaeological Investigation (5.3.1)

5.3.5 Prehistoric

Phase 2 of the NERCZA survey revealed no surviving prehistoric remains on the coast between Saltburn and Loftus.

5.3.6 Romano-British

A Roman signal station is said to have stood on the cliffs above Huntcliff Foot, however, despite field investigation the rapid survey identified no upstanding or eroding remains related to the signal station. There maybe buried remains surviving and if this is the case they are certainly at imminent risk of collapse on to the foreshore. The size and extent of these signal stations can be seen at both Scarborough (Pearson 2009) and Goldsbrough (Burn 2010), so this site may have substantial survival below the soil, and further investigation by close-spaced geophysical survey followed by evaluation trenching may be of use in gaining further understanding of what actually survives below the surface, its potential significance and its vulnerability to erosion.



Figure 5.3.5. Possible cliff top location of Roman signal station recorded by antiquarian excavation, notice the evidence of significant slumping of the cliff.

5.3.7 Early Medieval

Phase 2 of the NERCZA survey revealed no surviving Early Medieval remains on the coast between Saltburn and Loftus.

5.3.8 Medieval

One possible medieval feature was identified by the survey in the form of a linear bank close to the edge of the cliff. The bank was approximately 0.4m high and 1m wide, and it is possible that this represents a medieval plough headland. The current ploughed fields are set further inland and the Cleveland Way footpath runs between these fields and the possible headland bank. Any other evidence of medieval ploughing has therefore been removed by later ploughing and the creation of the Cleveland way.

5.3.9 Post-Medieval

The most prevalent surviving archaeology in this area dated to the post-medieval period. This is unsurprising considering the scale of both the alum and ironstone industries in the area. The survey identified several large structural remains associated with both these industries together with earthworks and rock cut features that help to mark their former extent across the landscape. These features include rock cut postholes, rutways, earthwork remains of industrial buildings and boundaries.



Figure 5.3.6 Remains of the former fan house at Skinningrove ironstone mine, looking west.

The most prominent industrial features are the upstanding structural remains, such as the fan house (19) for the ironstone mine at Skinningrove (Fig 5.3.6). Structural elements of this survive either side of the railway branch line that still serves the current ironstone works. These structures have signed interpretation and are set well back from the edge of the cliffs on National Trust land. This means that they can be characterised as low risk structures and are not under any immediate threat from erosion.

Perhaps the most threatened remains seen were those visible in the cliff face. Such features were observed at Hummersea and Saltburn, where structural remains are eroding out of the cliff. The former alum house at Hummersea is the most prominent example of this, and is threatened not only from direct wave action at high tide but also from ongoing slumping of the cliff from above (Fig 5.3.7).



Fig 5.3.7 Remains of Hummersea alum works slumping and eroding

Similar remains can be seen at Saltburn (Fig 5.3.8) although the exact form and function of these buildings is not precisely known. Further investigation of the cliff top remains, not visible on the surface, could be carried out to gain a better understanding and preserve them through detailed record. Further historical research has been undertaken as part of the Saltburn rutways project (Green 2008, 2009) but no attempt has been made to investigate the Saltburn alum works other than the NERCZA Phase 1 rapid survey.



Fig 5.3.8 Archaeological remains of Saltburn alum works eroding.



Fig 5.3.9 Surveying structures eroding from the alum works at Loftus

Other upstanding remains can be seen further south at Loftus where the alum works have already been extensively surveyed by English Heritage (Hunt *et al* 2005). The NERCZA survey rapidly re-established the position of the structures within the alum works and assessed the level of threat to them. Here, the proximity of many of the surviving remains to the edge of the quarry cliff meant

that many have partly collapsed and further degraded since the English Heritage survey (Hunt *et al* 2005).

Other structural remains identified included alum liquor settling tanks, still visible in overgrown cliffs (Fig 5.3.10). These remnants of the alum industry are well preserved and not as threatened as those within the alum quarry, directly on the cliff edge. However, they are still exposed to threat from long term cliff retreat. Also, all elements of the alum works must be considered as a whole as the understanding of the operation of the site suffers if part of it is lost or damaged by erosion.

More ephemeral remains can also be seen on the foreshore below the alum works at Hummersea and Loftus. Here rock cut features such as large rectangular holes (Figs 5.3.11 and 5.3.12) could be interpreted several different ways. They could be what are locally referred to as hulleys, rock cut holes that were used for the storage of bait and fishing equipment in the inter-tidal zone. There is no easy way to date these features but they most likely date to the 18th or 19th centuries due to the clean cutting of the rock. They could also be post holes and this is more likely of (78, 79) below the Loftus alum works, as a linear pattern can be seen. This may be part of a foreshore structure or an earlier alum pier, predating the stone pier that still survives (76).



Fig 5.3.10 Extant remains of settling tanks on the cliff top looking South.



Fig 5.3.11 Extant remains of large rock-cut features on the foreshore at Hummersea.



Fig 5.3.12 One of the smaller rock cut post holes at Loftus

The other rock cut feature that was frequently observed and recorded during the course of the Phase 2 survey was rut ways. It was deemed unnecessary to attempt to record all of these as firstly it would take more time and resources than were

available, and secondly this would overlap with the ongoing work of the Saltburn Rutways Survey. However, several were recorded (70, 80) and an extent of visible rutways (701) also recorded to show the area in which they have been seen to be still extant by the NERCZA survey team.



Fig 5.3.13 Surviving extent of stone alum pier at Loftus



Fig 5.3.14 Recording one of the numerous rutways between Saltburn and Skinningrove.

5.3.10 20th Century

The stretch of coast between Saltburn and Loftus also contains many surviving Second World War remains. Between Saltburn and Skinningrove the structural remains of two pillboxes survive, one sits in pasture fields west of the Cleveland Way (48) and although the area surrounding the upstanding remains has been heavily ploughed there is little direct threat to this monument. Another pillbox can be seen on the beach at Skinningrove (17). This structure has slumped down from the cliff onto the beach and is exposed to direct wave action (Figs 5.3.15 and 5.3.16). This pillbox has been observed both covered with sand and almost fully exposed demonstrating the changeable nature of its setting. It is set within the inter-tidal zone and repeatedly covered with sand and then re-exposed to the elements.



Figure 5.3.15 Slumped pillbox on the beach looking south towards Skinningrove harbour as seen in September 2009.



Figure 5.3.16 The same slumped pillbox as above more exposed to wave action, this time looking north. This photograph was taken in early December 2009 and demonstrates the changeable nature of the beach deposits and environmental conditions.

Other Second World War features identified include the earthwork remains of fire trenches and weapons pits seen particularly around Skinningrove. One well-preserved example (36) previously identified during the Phase 1 project is still extant on Skinningrove pier (Fig 5.3.17). The foxhole element is dug against two of the pier walls with a low defensive bank still extant internally. This would have been enhanced with sandbags and perhaps barbed wire defences.

These earthwork elements can be seen as some of the most at risk archaeology along this section of coast. These features were originally intended as temporary defences, and therefore were not expected to last more than a few years. They are also preserved in a thin strip of preserved land between agricultural land, ploughed fields, and the coastline. This makes the survival of these earthwork remains potentially significant, as they are increasingly under threat with ongoing erosion and retreat of the coast. This can be seen elsewhere along the coast and as approximately 75% of the newly identified archaeological sites date from the Second World War, it represents the largest proportion of directly threatened archaeology.



Figure 5.3.17 A preserved weapons pit on the western end of Skinningrove pier.

5.3.11 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.1m per year (SMP2 for north East England). The till cliffs along this stretch of coast are some of the areas most exposed to erosion in this region. There is little dune or beach protection and the archaeological features, especially those identified on the foreshore, such as rutways, postholes and other structures, are most at risk from erosion. The changeable nature of this stretch of coast is highlighted by the changing situation of pillbox (17) within just 3 months. Recent work by the Boulby Geoscience Project from Durham University in partnership with Cleveland Potash have suggested the rates of erosion here might not be as

severe or as rapid as previously thought (Steadman *et al* forthcoming). This research claims to have identified ancient landslips still visible along the coast. There is, however, also evidence that the cliffs here do periodically destabilise and collapse causing occasional rock fall erosion events that build up and increase the likelihood of one-off major rock fall events. The Boulby Project has identified this erosion pattern through long-term monitoring of the cliff face and repeated high resolution laser scanning of the cliff to track the development of rock falls. Although there is little overlap between the area covered by NERCZA Phase 2 and the Boulby Geoscience Project, there is a clear case for direct comparison due to the proximity of Boulby to Skinningrove, Loftus and Overdale Wyke.

It s clear that features on the foreshore along this stretch of coast are at imminent risk from erosion. Ongoing storm events will eventually remove the remains of rock cut features such as rut ways and post holes as well as other features such as shipwrecks and wooden posts. Also at risk are the cliff top features that relate to the military and industrial development of this coast as the cliff retreat undermines these remains resulting in their collapse and slumping. Along this stretch of coast it is not so much the rate of erosion that is alarming but when rare, but significant, erosion events occur large amounts of archaeological material is removed very quickly.

5.3.12 Summary and conclusions

This stretch of the coast within policy units 15.4 - 17.3 is mostly covered by a policy plan of No Active Intervention. The only areas covered by management plans are the area of Saltburn itself which is designated as Hold The Line and a comment of investigation of the potential threat to the railway line at Skinningrove (16.1). This leaves a substantial section of the coast containing possible Roman, Medieval, Post-Medieval and $19^{th}/20^{th}$ century remains vulnerable to coastal retreat. While the rate of erosion of these hard rock cliffs may not be as rapid as some other areas along the coast, the potential for significant rock falls leading to cliff destabilisation and collapse has been demonstrated (SMP2 for the North East 2008).

Management		Policy	Policy			
Area		unit	Plan			Comment
			2025	2055	2105	
	Marske and					
	Saltburn					
MA15	Sands	15.4	HTL	HTL	HTL	
						Investigate potential threat to
MA16	Huntcliffe	16.1	NAI	NAI	NAI	railway
	Skinningrov					
MA17	e	17.3	NAI	NAI	NAI	

Table 5.3 Shoreline Management Policy 2 policies for Saltburn and Skinningrove

The survival of above-ground archaeology from the post-medieval period can be seen to be generally good with industrial, structural and military features surviving very well. However, little is known about the extent of survival below ground, especially of medieval and earlier remains. Further investigation of sites such as the Roman signal station (Sherlock, 2005) south of Saltburn would also be beneficial, and possible features associated with the alum industry, which may survive well below ground but at present this is unconfirmed without recourse to further investigative fieldwork.




5.4 Greatham Creek and North Gare, Teesside.

5.4.1 Background

The SMP2 recommendation for the areas adjoining Greatham Creek is 'No Active Intervention' to be followed in the middle term by 'Managed Retreat' which may involve the construction of sea defences. Three categories of asset in this area can be considered vulnerable to the effects of flooding due to rising sea levels and/or the construction works involved in the erection of sea defences. These consist of a prehistoric midden identified by the Phase 1 study, a group of medieval and later salterns and the military features of the Greatham Creek Defence Area, the latter being one of the most complete surviving Second World War defensive complexes in the North East.



Fig 5.4.3 Stanton type air raid shelter at Greatham Creek, part of former bombing decoy.

5.4.2 Location and geology

The Greatham Creek defence area is located between Middlesbrough and Hartlepool, north of the mouth of the River Tees (NGR NZ 50721 25534) (SMP PU 13.1 - 13.5) (Fig 5.3.1). The area is made up of superficial deposits of alluvium with sands and silts underlain by glacial till. The solid geology beneath this is Sherwood sandstone (BGS). The area is mainly salt marsh with a large proportion of reclaimed land for industry surrounding this. Land use is currently a mixture of industrial complexes, brine fields and open pasture. The area is low lying and open and as a result is at risk of flooding if projected sea level rise is correct (see projected sea levels on Fig 5.4.1).



5.4.3 Previous research

Areas of Greatham Creek have been investigated separately before as part of Environmental Impact Assessments which have included cultural heritage chapters. One of these was undertaken on behalf of the environment agency by Northern Archaeological Associates but as with assessments undertaken as part of commercial development this did not cover a wide area or include a high level of detail on what still survives. Studies of the military landscape such as the Defence of Britain project have also covered this area. However, no holistic study of the surrounding area has been undertaken. As a result of this the area surrounding Greatham Creek around the North Gare (NGR NZ 53677 28231) was also investigated in order to give as complete a picture of the surviving defences as was possible.

The Defence of Britain project examined detailed Ministry of Defence records in order to piece together a picture of what was planned for the Greatham Creek Defence area. However the limitations of the Defence of Britain project, for example inconsistent fieldwork methodology, have previously been recognised and are discussed in greater detail in section 5.17. In addition, what was recorded by the Ministry of Defence during the Second World War as planned was not necessarily built in the same way or at all during the Second World War. So although an excellent record of planned defences survives this does not necessarily reflect what was actually built and what survives on the ground. This principal applies equally to other sites with surviving military remains recorded by NERCZA along the coast.



Fig 3.4.4 Generator house for bombing decoy with Stanton shelter in background.

5.4.4 NERCZA Phase 2 Archaeological Investigation

5.4.5 Prehistoric – Early Medieval

There was no prehistoric, Romano British or early medieval archaeology identified during the course of the survey

5.4.6 Medieval

The main medieval remains identified were the extensive survival of medieval saltern mounds. These mounds were identified from the air during Phase 1 of and within Phase 2 the mounds were assessed in terms of their level of threat and their extent and position accurately mapped. A concentration of these (184) was located directly west of the North Gare, in a field surrounded by reclaimed land, dunes and developed areas.



Fig 5.4.5 recording one of the Saltern mounds, modified in the cold war to house a Royal Observer Corps station.

The medieval salterns are a predecessor to the modern brine fields, both used for salt production. They can also be seen to the south in the area directly surrounding Greatham Creek.

5.4.7 Post-Medieval-Modern

Post-medieval archaeology in the area takes the form of reclaimed patches of land and industrial components. Many of these areas are still in use and as a result have evolved considerably. One of the most obvious elements of the post-medieval archaeology is the reclamation of land from the 19th century onwards. Historic Ordnance Survey mapping shows the development of the landscape from the 1860s onward and clearly demonstrates that the area west of North Gare, to the south of the medieval salterns, was not reclaimed until late in the 19th century. The survey actually picked up evidence of this in the form of part of a former dune system visible on the 1st and 2nd edition OS maps. This was

recorded initially with the thought that it may be a saltern mound. However, inspection of the historic mapping indicated that this area was in fact not reclaimed until the late 19th century. This ruled out a medieval date and when the position corresponded to the position of dune 'islands' it was realised that this was actually a remnant of the pre-reclamation landscape preserved within the reclaimed fields.



Fig 5.4.6 Remnant of pre-reclamation landscape preserved to the west of North Gare.

A relict railway embankment (151) also survives running through the brine fields at NGR (NZ 50942 24989). This was formerly attached to the brine works but fell out of use in the early 20th century. The embankment is clearly extant, although some sections have now been removed. It has been re-used later as the foundation bank for pillboxes and section posts. This can be seen elsewhere in Northumberland where gun emplacements have been seen placed on top of such embankments to provide better firing positions.



Fig 5.4.7 Former railway embankment reused as a platform for a pillbox (left) and section post (right).

The survey also revealed evidence for a possible shipwreck in proximity to the North Gare. The whole outline of the vessel can be seen on aerial photograph transcription from Phase 1. However, now only a small section is visible (156). This was photographed and when revisited 3 months later the location was re-established but the wreck was no longer visible at all.



Fig 5.4.8 Partially visible remains of Shipwreck (158) on the foreshore at North Gare.

5.4.8 20th Century Military archaeology

The majority of archaeological features recorded a Greatham Creek and North Gare related to the Second World War defence area. The survey recorded extensive surviving remains ranging from slight earthworks such as anti-glider trenches to upstanding buildings such as pillboxes and section posts. Also recorded was evidence of Cold War activity, and the surface remains of the entrance to a surviving Royal Observer Corps bunker was also recorded. The most obvious surviving features are the well-preserved pillboxes and section posts that can bee seen in the area surrounding the creek. These can be distinguished by their distinct shapes, pillboxes being square or rectangular structures built out of thick concrete, while section posts are of less substantial concrete and are chevron shaped. Three section posts (143, 144, and 145) can be seen along a former strategic track (146) just south of seal sands. Although now partly buried by a dyke that forms the edge of the nature reserve they are still excellently preserved with several internal features surviving. These section posts were used to house a section of infantry who could fire out of the rifle loop holes. Pillboxes are different in that they would house machine gunners and larger calibre guns. It is likely that the presence of so many section posts in the area of Greatham creek can be attributed to the fact that the marshy ground prohibited the successful excavation of fire trenches. More section posts can be seen in the area of open fields to the north of the surviving bombing decoy site.



Fig 5.4.9 Partly buried section post (145) showing their distinctive chevron shape in plan.

The front of these section posts is no longer visible but the interiors show that the rifle loop holes are still open and sand is beginning to fill these structures. Internal features may in time be lost as a result of this. Section posts can also be seen on the railway embankment (152) described in section 5.3.10 and here they appear to from a "fortlet" (Fig 5.4.12) defending the creek itself, presumably to stop any incursion up this waterway.

The remains of partially demolished pillboxes (155, 156, 154) were located on the North Gare itself and these correspond to known locations relocated using Phase 1 data. The remains of a concrete wall with loopholes and foundations can clearly be seen. These demolished remains are being reused to reinforce the collapsing sections of the North Gare pier (Fig 5.4.10)



Fig 5.4.10 The remains of pillboxes used to reinforce the North Gare pier



Fig 5.4.11 remains of loop holes in concrete wall fragments from demolished pillbox

Several smaller defensive features were also recorded, these included a previously recorded spigot mortar (141) (Fig 5.4.13) now buried by the track leading to the nature reserve. Only the mortar pivot is still visible protruding from the track. This would have been one of several such spigot mortar along the Creek placed in key firing positions to allow rapid repositioning. The others have now been lost.



Fig 5.4.12 'Fortlet' 152 overlooking Greatham Creek made up of two large section posts.



Fig 5.4.13 Spigot mortar base 141 partially protruding from track edge.

Also visible along the edge of the Creek were two possible positions for pillboxes facing inwards to protect against incursions from the channel. These rectangular concrete platforms are the typical shape and size and are made from concrete comparable with other defensive structures in the area (Fig 5.4.14). However, there is no way to definitively verify these as pillboxes as they are not identifiable on period photography.



Fig 5.4.14 possible remains of demolished pillbox 142.

Several earthwork remains were also observed; these primarily consisted of antiglider trenches (177, 178, 183, 182, 189). Although many of these were identified by the Phase 1 mapping the Phase 2 fieldwork found many of these were no longer visible. Several were seen in the vicinity of North Gare and others were well-preserved within the Hartlepool Golf club. The development and reclamation of these areas has led to many of these trenches having been destroyed. Field survey also found that many of the anti-glider trenches identified by Phase 1 in the area recorded as 184 were not in fact anti-glider trenches but either natural channels or drains, possibly related to salt production in the medieval period.



Fig 5.4.15 Recording remains of a surviving anti-glider trench 182.

Also visible in the dunes to the north of North Gare were extensive lines of antitank blocks. These are visible in dunes as a double row all the way from the Gare for at least 2km. They are buried in places, but the same continuous line can be re-established (165, 160, 164, and 161) and the line followed in dune banks which now cover the anti-tank blocks. Also seen in and around these lines of anti-tank blocks were the remains of barbed wire being exposed in small dune blow-outs. These remains are extremely fragile and are being effectively blown away in strong winds. The lines of barbed wire can be seen as stained areas of sand with fragmentary remains of barbed wire, and occasionally picket wire, surviving.



Fig 5.4.16 Anti-tank blocks at North Gare.



Fig 5.4.17 Remains of barbed wire and picket wire at North Gare.

The surviving military remains of a First world war seaplane base were also recorded, although they had been reused later. The site of the former pier has been used as the foundation for a Second World War pillbox. Phase 2 identified the location and surviving extent of the First World War seaplane base at North Gare first identified as part of Phase 1. There are buildings surviving directly next to the power station that date to 1915 (204), which are presumably related to the seaplane base. A slipway and pier (203) survive close to the power station, these were built for the seaplane base, and it is on this pier that the Second World War pillbox (202) was constructed. An area of raised flat ground between the slipway and 1915 buildings is clearly visible, this platform is evidently the remains of the sea plan base itself, the buildings having subsequently been removed.



Fig 5.4.18 Remains of a WWI seaplane base (203) at North Gare.

As well as this earlier element of defensive archaeology the survey revealed a much more recent phase of military activity. In the area of saltern mounds (184) the survey identified the remains of a hatch and other surface features belonging to a Royal Observer Corps monitoring post. These were built all over the country to monitor the conditions and resulting situation in the event of a nuclear attack (Cocroft 2008). The surface features would have been a hatch, several equipment antennas, and a ventilation system. The field survey identified the probable locations of a hatch (185) and ventilation shaft (186) protruding from a modified saltern mound (187). The hatch was sealed and there is no way of gaining access easily so there is no way to ascertain the internal condition of this monument.



Fig 5.4.19 Remains of sealed hatch into ROC Cold War monitoring post.

5.4.9 Threat from erosion

The Shoreline management plan indicates that the typical erosion rate for this type of undefended coastline is 0.1m per year, although at Greatham there is also the additional risk of flooding. The main types of recorded archaeology relate to structural remains in locations where they are not under direct threat from erosion. The section posts and pillboxes at Greatham Creek are not under immediate threat unless sea levels rise significantly in the future. This would then lead to increased impacts upon them. Similarly the lines of anti-tank blocks at North Gare are actually helping to protect the coast as they are accumulating sand and creating artificial dunes around them. Further inland the greatest threat posed to the medieval remains other than sea level rise is that of ongoing trample from cattle which, over time, could significantly change their profile and remove medieval evidence.

There is a significant threat to the remains of the seaplane base at North Gare as it is exposed at high tide and the slipway is in a significant state of decay. On the Gare pier the remains of Second World War defences have evidently been bulldozed away recently to reinforce the dilapidated pier structure. Although there can be no way of preserving these remains now this could be seen as an exemplar of what not to do. A significant proportion of the story of the Second World War defence of this area has been lost as a result of this action. Ephemeral remains, such as the barbed wire entanglements being revealed in the dunes, are not just threatened by erosion but pose a threat themselves. They could be a hazard for dog walkers or children as they are still sharp and dangerous. This means they should be a priority for further management as they have minimal archaeological interest, other than as a record of their presence, but pose a problem to the areas amenity.

5.4.10 Summary and conclusions

The main threat in the Greatham Creek area is flooding due to ongoing sea level rise and increased threat of dune blowouts from erosion. The SMP2 policy for this area is No Active Intervention or Managed Realignment of the coast (see Table 5.4). The exception to this is the area around the North Gare and Seaton Carew. Here, the developed areas and existence of the power station mean that policy is to protect the area from any further erosion and defend against flooding. The impact of these on archaeology is that any flood defence works or implementation of sea defences could inadvertently destroy or damage some areas of archaeological interest and importance. For example, the seaplane base west of North Gare.

Management		Policy	Policy			
Area		unit	Plan			Comment
			2025	2055	2105	
					ΗT	Consider planned
MA13	Seaton Carew	13.1	HTL	HTL	L	realignment
						Possible future feed with
MA13	Seaton Sands	13.2	NAI	NAI	NAI	dredged material
					ΗT	
MA13	North Gare	13.3	HTL	HTL	L	
	North Gare					controlled by structure to
MA13	Sands	13.4	NAI	R	R	south
						Investigate use of dredged
MA13	Bran Sands	13.5	NAI	NAI	NAI	material

Table 5.4 Shoreline Management Plan 2 policies for the Greatham Creek area.

The Archaeology in the area of Greatham Creek and North Gare represents one of the best-preserved examples of a completed Second World War defence area along the North East Coast. It also has the added First World War and Cold War elements giving a chronology of defensive activity within the same area. Much of what was originally constructed still survives in some form allowing a picture of the various stop lines and key defensive points to be built up. This archaeology is clearly under threat from both active erosion and demolition in the short term and rising sea levels in the long term.

The medieval remains, while not under immediate threat, must also be considered in the long term. Rising sea levels would see a large percentage of the land reclaimed by man in the 19th and 20th centuries reclaimed by the sea in the 21st century. This would not just affect the military archaeology and produce a landscape more akin to that depicted on the 1st edition Ordnance Survey mapping, but also increase flood risk further inland, leading to the area of medieval salterns being directly affected.

Given this context there is benefit in maximising understanding of the significant elements of the wartime archaeology followed by a detailed survey of the medieval salterns. Such surveys would then provide a more complete record and gain in understanding for both sets of what are currently exceptionally wellpreserved remains.





5.5 Crimdon Dene and Blackhall, County Durham.

5.5.1 Background

The possible location main flint scatter site at Crimdon Dene is close to the MHWS limit while the sites at Blackhall are on the cliff edge. Within the SMP2 these sites are in an area of the coast where the recommended policy is one of 'No Active Intervention'. The former is vulnerable to the effects of sea level rise and the latter to the erosion of the cliff. The first aim of the survey was seek to identify the location of the scatters. A detailed check of the area was made to identify whether there were further visible flint scatters in the vicinity, as flint scatters are rarely found in isolation.

An organic layer buried below a layer of sand was identified as a possible land surface potentially related to that described as containing the flint scatter (Coupland 1936). This possible land surface was sampled as part of the palaeoenvironmental sampling and is discussed further in Chapter 6. What also became evident during the course of the field survey were the extensive surviving Second World War remains surviving around the mouth of the Crimdon Beck. The survey area was therefore expanded to include and assess the extent of these remains. A similar exercise was undertaken at Blackhall focusing on the surviving remains of the colliery.



Fig 5.5.3 Crimdon Dene in full flow in winter.



5.5.2 Location and geology

Crimdon Dene is located on the Durham coast north of Hartlepool (NGR NZ 48913 36566) (SMP PU11.1). The geology of the area is Magnesian Limestone overlain by a clay till which supports a grassland cover. The coastal cliffs are broken up with small valleys winding inland between low rolling hills. These denes and valleys have formed foci for settlement during the prehistoric period, and for the Mesolithic in particular (Waddington 2007). There are large sand dunes to the south of Crimdon Dene and these are eroding rapidly as they are being cut back by wave action, but also by the flow of the Crimdon Beck which is cutting onto the dune system from the landward side. To the north the dunes run into limestone cliffs with a rock cut platform and intermittent sand beach.



Fig 5.5.4 One of the large dunes to with the Dene in the foreground, showing the effects of ongoing erosion.

5.5.3 Previous research

Extensive research into Mesolithic activity on the Durham has been carried out since the 1920s. Trechmann first identified the Mesolithic flint scatter at Crimdon Dene in 1912. He retrieved a scraper and several flakes (Raistrick and Coupland 1936). Further investigation revealed Mesolithic flints exposed on the surface of the till but situated below the overlying sand, a situation witnessed elsewhere on the North East coast at Nessend, Holy Island and Newbiggin for example. Raistrick and Coupland identified cores and flint chips and compared the Crimdon Dene site to that further up the coast at Newbiggin (Raistrick and Coupland 1936). Coupland records that 12,000 flints were recovered from the visible exposures. It is not currently known if these artefacts or any archive survives from these early investigations.

The location of the site is described by Raistrick and Coupland (1936, p84):

"The site is situated at the mouth of Crimdon Dene, about 2 miles north of West Hartlepool, where the small burn makes a southerly band as it approaches the sea, forming on the north side of a longish spur, between the shore and the burn. This spur is actually a small cliff of boulder clay covered in part by blown sand."

This is the only archaeological investigation that has taken place at Crimdon Dene, although various attempts to relocate the scatter have taken place. Further work was undertaken at Blackhall were a second scatter was located and limited excavation undertaken by Coupland. This site is proximal to the large former colliery at Blackhall.



Fig 5.5.5 Investigating the landscape containing the flint scatter recorded by Raistrick and Coupland.

5.5.4 NERCZA Phase 2 Archaeological Investigation

5.5.5 Prehistoric

The Phase 2 survey of Crimdon Dene did not identify any definite prehistoric features. However the probable location of the Crimdon Dene lithic scatter as described by Trechmann was thought to have been identified. Using the description of the site (see above) the possible location of this layer was identified (se fig.5.5.1). Buried below sand on a spur of land several organic-rich layers were observed that represent old land surfaces. These organic horizons were sampled (Fig 5.5.6) for radiocarbon dating and the spur of land and sampling points surveyed. However the organic samples have subsequently returned dates of cal 1956 AD, ruling out these deposits as prehistoric. The location of the Crimdon Dene flint scatter has therefore not been identified and the implications of this are discussed in more detail in Chapter 6.



Fig 5.5.6 Uncovering the buried organic layer on the spur of land at Crimdon Dene.

Several pieces of flint were retrieved from the spur of land during the survey although these were out of context and also heavily beach rolled (98). It is possible that they are remnant cores as one in particular seemed to have a series of planned blade scar removals. As flint occurs naturally here in beach pebble form it is not always certain as to whether a piece has been worked. The evidence for surviving prehistoric remains at Crimdon Dene is suggestive but not yet conclusive. The organic remains do not represent a prehistoric layer and the location of the flint scatter remains to be definitively located. Further investigation and a more detailed survey would enhance the understanding of this site and it's potential.



Fig 5.5.7 Location of the "spur" of land described by Coupland which is thought to be the land on the right of the picture.

At Blackhall no surface evidence of the lithic scatter reported to be there was observed.

5.5.6 Other pre-medieval evidence

No Romano British or early medieval features were identified during the course of the survey in this area.

5.5.7 Medieval

Remnant earthworks of broad medieval ridge and furrow ploughing were observed within Crimdon Dene caravan park (87). These remains are preserved on two plots of land directly to the north of the park and as well as being broad have a distinct reverse S bend to them indicating that they were formed by ox and not horse drawn ploughs. This type of ridge and furrow is associated with the medieval period. The graded nature of the earthworks suggests that the field was ploughed again in the post-medieval period thereby truncating the earlier ridges.



Fig 5.5.8 Location of the remains of heavily graded broad medieval ridge and furrow at Crimdon Dene Caravan park.

5.5.8 Post-Medieval

The main post-medieval evidence comes from the remains of the colliery at Blackhall. Earthwork and structural remains survive on the east side of the railway. Part of the site is now an industrial estate, however to the south of this extensive but fragmentary remains of the former works can be seen. This site could be recommended for a full detailed Level 3 survey (Ainsworth *et al* 2007) to fully record all components of the site. The current survey recorded only the extents of the visible remains. Building platforms, tracks, and other earthworks, as well as fragmentary structural remains, can all still be identified within the extent of the former works.



Fig 5.5.9 The Blackhall colliery site.

5.5.9 20th Century

The main element of 20th century archaeology were military features dating to the Second World War. At the mouth of Crimdon Dene the Phase 1 recorded extensive defensive features visible on aerial photography, presumably defending the easy access up the Crimdon Beck. Unsurprisingly most of these more ephemeral features have since been removed, however the remains of pillboxes, section posts and an anti-tank battery can still be seen.

The well preserved anti-tank battery (83) is defended by a pillbox (82) and a communications trench (105) is also still visible running up the hill between them. Interestingly, although the Phase 1 study picked up features such as anti-tank cubes across the base of Crimdon Beck and the military features on the beach and cliff top in the vicinity of this battery, they did not pick up any trace of features 82, 83 or 105. This perhaps highlights the limitation of aerial survey as these unrecorded features are now some of the most visible in the landscape and are exceptionally well preserved. The anti-tank battery was most likely heavily camouflaged during the Second World War and this may account for it not having been picked up on contemporary aerial photography.



Fig 5.5.10 Anti-tank battery at Crimdon Dene, showing access on the roof.

The anti-tank battery is exceptionally well preserved, probably due to the partial burial by sand of most of the structure which has served to protect it. Both the pillbox (82) and battery (83) are on the edge of an area of stable dunes, below the golf course and overlooks a commanding position over the beach and mouth of Crimdon Beck.



Fig 5.5.11 Pillbox (82) built to defend the anti-tank battery (83).

To the south of these structures two areas of collapsed concrete rubble (104) and (106) can be seen eroding out of the dune cliff and slumping down onto the beach. These areas of rubble relate directly to earlier structures recorded in Phase 1. Also visible on the beach are the remains of three pillboxes (108, 109, 110) and a section post (512), all relatively well-preserved and partially buried by beach sand. It is also possible to say from comparison with the 2009/2010 GPS position with the 1940s AP transcription data that all of these remains have shifted several metres from their original position, probably due to slumping and long shore drift. A second section post was visible until very recently and is even depicted on the latest Ordnance Survey mapping but has since either eroded or been removed as it was unsafe, and was gone by the time of the survey.



Fig 5.5.12 One of the exposed pillboxes (106) on the beach at Crimdon Dene.

Another surviving Second World War element at Crimdon Dene is the earthwork remains of trenches. There is an extensive network of these directly overlooking Crimdon Beck forming what can be described as a hilltop redoubt. Trenches can be seen forming a zig zag fire trench pattern, with communication trenches extending back from the main fire trench to the edge of what is now a footpath. Although now heavily overgrown, the survival of these trenches is exceptional with some over 1m deep.

During the wartime period beach defences comprising anti-tank blocks and barbed wire were deployed at Crimdon Dene. These are no longer visible on the beach but many anti-tank blocks can be seen removed from the beach in caches behind the dunes (112). At the same location *in situ* anti-tanks blocks (111) have been used as the foundation for a modern bridge.



Fig 5.5.13 Top: Anti tanks block utilised as a foundation for a modern bridge. Bottom: Cache of anti tank cubes probably removed from their original location.

There are other military remains visible in the vicinity that cannot be positively identified due to the level of ongoing erosion. These concrete structures (114) (116) are at immediate risk of collapse and also relate to features identified during the Phase 1 survey.



Fig 5.5.14 eroding unidentified military remains at Crimdon Dene

5.5.10 Threat from erosion

There is an ongoing risk of erosion at Crimdon Dene, especially to the south of the beck. The erosion and destabilisation of the dune cliff, combined with the erosion caused by the cutting back of Crimdon Beck have led to a high rate of erosion. By comparing aerial photography from the 1940s and the 2009 survey an estimated rate of erosion over a 60 year period can be worked out. It can be seen by comparing the 1940 and 2009 position of the same military features that some parts of the cliff have retreated as much as 40m in 60 years. That works out at a localised rate of erosion of 0.67m per year. Although not the most accurate method due to the shifting of the archaeology as a result of long shore drift, this is a useful way to estimate the erosion rates at a local level using the surviving archaeology (Fig 5.5.15).

5.5.13 Summary and conclusions

There is significant archaeological potential at Crimdon Dene for both the surviving Mesolithic and for substantial surviving Second World War remains. Not only is there a large amount of surviving military remains preserved, but there is the potential for preserved Mesolithic archaeology. The project has identified the possible location of the Mesolithic flint scatter first identified in 1912 that has produced 12,000 flints (Raistrick and Coupland 1936). It has been seen elsewhere that such sites could preserve further evidence of prehistoric settlement as was case with the Mesolithic house at Howick (Waddington 2007).

Management		Policy	Policy			
Area		unit	Plan			Comment
			2025	2055	2105	
	Crimdon					Local management to beck may be considered. Possible beneficial use
MA11	Valley	11.1	NAI	NAI	NAI	reasons

Table 5.5 Shoreline Management Plan 2 policy for the area of Crimdon Dene

The SMP policy for this stretch of coast is 'no active intervention' in the shoreline management plan PU11.1. Taking the rate of erosion seen to the south of the beck, the potential importance of any buried archaeology and the preservation of Second World War archaeology, it is clear that the site faces a high level of threat. The military archaeology preserved at Crimdon Dene is not the best-preserved along the North East coast; however the amount which survives within a small area does make the site interesting. For this reason further investigation of the location of this Mesolithic site and a more structured programme of field walking and excavation would be helpful and crucial to increasing our understanding of the survival of the Mesolithic at Crimdon.



5.6 Trow Point, Tyne and Wear.

5.6.1 Background

At Trow Point an erosion rate of 0.2m/per year has been recorded (SMP2 for North east England 2008) and the SMP2 policy recommendation is to allow the cliff face to retreat. Any remaining evidence of the Trow Point barrow will soon be lost at this rate of erosion. The survey aimed to:

- Ascertain whether the barrow is suffering active erosion.
- Evaluate both the present level of threat to the barrow and the nature of those threats.
- Examine whether there are other visible features associated with the barrow and the extent to which such features are subject to, or at risk from, erosion.
- Record the present extent and condition of the barrow.

Upon initial walkover survey of the site it became clear that it was unlikely that any evidence of the barrow survived. Although there are numerous earthworks surviving on the top of Trow point it is unlikely that any of these relate to the prehistoric period. As with many of the sites surveyed, the majority of features identified relate to Second World War defences and the main exception to this was the late nineteenth century gun emplacement on the point itself.



Fig 5.6.3 The site of the experimental late Victorian "disappearing gun" on Trow point.


5.6.2 Location and geology

Trow Point is a promontory located at NGR NZ 383 665, (SMP PU 2.3 - 3.2) flanked by two sandy beaches, with low cliffs to the south and dunes to the north. The landward side of the promontory has been mostly removed by the extensive workings of Trow quarry which can be seen directly to the west and south of Trow Point. The solid geology of the area is Upper and Lower Magnesian Limestone of the upper Permian age. The Upper Magnesian Limestone is shown to be present beneath the majority of the site (Cooper *et al* 2006). Trow Quarry is in the lowest division of the stratigraphic unit, the Concretionary Limestone. This consists of finely laminated dolomitic limestone. It is grey/brown crystalline limestone with subordinate creamy yellow oolitic or finely granular dolomite.

The limestone exposed at Trow Point is from the middle of the Concretionary Limestone and is characterised by varied calcite concretions (BGS). The area around Trow Quarry lies on the Upper Magnesian Limestone over Durham Lower Boulder Clay. This is overlain by the Tyne-Wear Complex. In addition to the natural geology around Trow Quarry, the quarry itself has been in filled with demolition waste and burnt domestic waste. A layer of Made Ground is therefore overlying the Upper Magnesian Limestone.

5.6.3 Previous research

There has been no detailed research into the remains at Trow Point, however antiquarian investigation recorded the existence of a barrow on the point which was eroded and contained a Cist burial this apparently has subsequently completely eroded away or been removed by continuing quarrying.. The exact location of this burial has now been lost although there is the recorded find of a bronze socketed axe found on the shore below Trow Point in 1867 (Tolan-Smith 2008). This is now stored in South Shields Central museum.(Hart 2004). There has also been summary of the archaeology and history of the coastline between South shields and Whitburn undertaken by the Arbeia Society and this deals with the barrow, disappearing gun, and quarry remains although no field survey was undertaken as part of this exercise (Hart 2004).

5.6.4 NERCZA Phase 2 Archaeological Investigation

5.6.5 Prehistoric

The survey revealed no definitive evidence of prehistoric activity. There was one sub-circular earthwork recorded (132) on the southern edge of the point, however this was highly truncated by military earthworks surrounding and could relate to quarrying activity or earlier military activity rather than the remains of a barrow. This earthwork is only a tentative proposal for the location of the barrow recorded in the 19th century as a more likely explanation is that the location of the barrow is now completely lost.

The 1st edition Ordnance Survey map shows the location of an OS trigonometric point (trig point). These were often constructed on the highest flattest point, and upstanding prehistoric barrows were ideal candidates for these structures, indeed OS trig points are often seen constructed on top of prominent barrows or cairns. The location of this trig point is now lost, and so it is possible that the barrow is also lost, this is as the trig point represents a likely candidate for the barrow's

former location. Another possibility is that the barrow has been removed by the quarrying in the early 19th century.



Fig 5.6.4 Truncated sub-circular earthwork, the only surviving candidate for a possible barrow on Trow point.

5.6.6 Romano British onwards

No Romano British, early medieval or medieval features were identified during the course of the survey

5.6.7 Post Medieval

The most prominent feature on Trow Point is the Victorian gun emplacement (119) (Foster 2004). The gun that can be seen at Trow Point now is a later replacement of an experimental Victorian gun known as the disappearing gun. The gun was initially hidden from view in an underground chamber, this would then be pumped full of water in the event of an attack, causing the gun to rise up as a result of the water pressure. This system would have required a system of water pipes and pumping system to control it and although most of this has now been removed, in one area a short section of piping is visible (131) which could relate to this pumping system. The disappearing gun system was found to be impractical and during the Second World War the defences were moved to Frenchman's battery just to the south. However, Trow Point was used as a hilltop redoubt with pillboxes, weapons pits and trenches constructed (see section 5.6.11). The limestone cliff immediately in front of the disappearing gun is eroding at a rate of 0.1m per year and the site as a whole will become increasingly threatened over the coming years.



Fig 5.6.5 The Location of the disappearing gun at Trow Point.

The other component of post-medieval archaeology visible in the area is Trow Quarry. This large limestone quarry has left a huge scar in the landscape, making Trow point even more prominent in the landscape (fig. 5.6.1). The survey recorded the extent of the quarry earthworks (700) and also a possible related feature, rock cut steps (Fig 5.6.6) (137) that lead from the quarry base up to the top of Trow point. There is no definitive way to date these but it is logical to associate them with the quarry as they lead from the quarry base. An exact date is not possible at this point. The steps are heavily worn and have clearly been used extensively over a long period of time, excluding the view that they are a more recent feature.



Fig 5.6.6 Rock Cut steps at Trow Point.

5.6.8 20th Century

On Trow Point there are numerous well-preserved earthworks dating to the Second World War. These comprise weapons pits, fire trenches and the remains of several fortified gun positions. There is also evidence for two other bases for large calibre guns and these are rapidly being exposed. The main visible component of the earthworks consists of four trenches (121, 126, 128 and 136) (Fig 5.6.14 and 5.6.8) and six weapons pits (120, 124, 125, 127, 133 and 134) (Figs 5.6.14 and 5.6.7). These are well-preserved and can clearly be seen across the top of the point. The concentration of defensive earthworks here is probably intended to defend a key point on the promontory, although this does not relate directly to the disappearing gun as this was not in use during the Second World War, but a later mounting for a First World War coastal defence gun, possibly still in use during the Second World War (122 and 123) (Fig 5.6.9). The coastal gun position consists of two surviving mountings, 122 and 123. These are circular concrete bases 4m in diameter with central fixed pivots to rotate the gun. The proximity of the two positions means that two guns could not effectively be operated at the same time as they could not fully rotate. It is likely that these were used as reserve positions onto which a gun could be moved to get a better arc of fire where it was needed. An earthwork 'pit', now filling with windblown sand which is stabilising and supporting grass, can still be seen surrounding these two mountings slowly covering the surface area of concrete.



Fig 5.6.7 The position and depth of one of the weapons pits with one of the WW1 gun positions visible in the foreground.



Fig 5.6.8 Recording one of the trenches on Trow Point



Fig 5.6.9 One of the two gun mountings 122 and 123 on Trow Point.

The other component of military defences surviving at Trow Point is the remains of four small pillboxes and other earthwork evidence of their original positions (129, 130, 131, 135, 138 and 139). These are not true pillboxes as seen elsewhere at key defensive points but more defended and reinforced defensive positions. They are small and built to defect small arms fire and not large blasts or direct hits. It is unlikely that all of them were originally roofed. They are rapidly constructed defences built on a prominence along the coast to defend the beaches, but also to defend from a flanking manoeuvre behind Trow Point, through the old quarry towards Frenchman's battery. One of these positions (139) has slumped down onto the beach and is being eroded, but its original position (138) can still be seen on top of the cliff (Fig.5.6.14)



Fig 5.6.10 One of the eroding reinforced gun positions (129) overlooking the beach at Trow Point.



Fig 5.6.11 The gun position 139 on the beach having fallen from the top of the cliff on to the beach.

The other large surviving military components dating to the Second World War are the extensive earthwork remains of Frenchman's Battery (140). Here, the main gun positions and a series of subsidiary buildings can all be seen preserved as earthworks, although heavily overgrown. This was effectively the replacement of the single gun at Trow Point and the sister to the Tynemouth and Blyth batteries further north. The site would benefit from further detailed survey to fully understand the extent of what remains prior to its eventual loss to erosion.



Fig 5.6.12 The surviving overgrown earthwork remains of Frenchman's Battery south of the view looking north towards Trow Quarry

5.6.9 Threat from erosion

The base line rate of erosion at Trow Point has been calculated at 0.2m per year.SMP2 for North East England) However, the increase in periodic storm activity means that over time there are significant erosion events that lead to the destabilisation of the cliff edge at Trow Point (Fig 5.6.13). This can be seen by the loss of the pillbox, trig point and possible barrow on the south east corner of the point. The ongoing erosion can also be seen by the recent collapse and slumping of another gun position on the north western edge of the point. The wave action is battering the front of the point, however the most serious and rapid erosion seems to be on the north western and south eastern edges close to where the point joins the mainland. This can also be seen by looking at the coastline as mapped by the Ordnance Survey first edition, which shows a retreat of between 5 and 11m in these areas since they were first mapped on the OS 1st edition 1:2500 map in 1858.

5.6.10 Summary and conclusions

The long term policy for Trow Point is 'managed realignment' (Table 5.6), as little can be done at present to significantly stop the ongoing erosion. The site of the Victorian disappearing gun could be considered significant as it is a unique and nationally significant structure. However, the gun on site now is not the original and the mechanism controlling the gun does not survive, so the survival of original remains is limited. The other earthworks and remains surrounding the gun are not in themselves exceptional but combined together they present a wellpreserved and virtually complete collection of Second World War defences. The two gun positions that have already been lost to erosion do not detract form the fact that this is an exceptionally well-preserved prominent redoubt and a story of the development and reasoning for these defences can be built up by the study of these remains.

Management						
Area		Policy unit	Policy Plan			Comment
			2025	2055	2105	
						Maintain longer term control
MA2	Trow Point (north)	2.3	R	MR	HR	function
MA3	Trow Point (south)	3.1	R	MR	HR	As required for MA2
MA3	Trow Quarry	3.2	HTL	MR	HR	Subject to detailed appraisal

Table 5.6 The SMP 2 policy for Trow point and its environs.

When considering in Frenchman's Battery and the now lost observation post at Graham's Sands into the story a complete picture of the development of coastal defence in this area can be built up. The Phase 2 survey has highlighted for the first time many features not previously recorded in the HER. A detailed Level 3 survey of the remains of Frenchman's Battery would preserve it through record although it is not currently directly threatened by erosion.



Fig 5.6.13 The cliff directly underneath the gun emplacement at Trow Point.



5.7 Robert's Battery, Northumberland.

5.7.1 Background

The remains of Robert's Battery represent an example of a multi-period gun battery, situated at Hartley, just south of Seaton Sluice on the Northumberland coast (NGR. NZ 3425 7612) (SMP PU 24.1-24.2). The site consists of a preserved encampment and command post, which dates from the First World War (now a private residence known as Fort House) and a later Second World War battery which was never fully finished. The battery itself was mostly buried underground with apertures for the guns protruding just above the cliff. The remains face ongoing threat from cliff retreat and collapse but this is disguised by the fact that the only visible remains of the emplacement, which date to the Second World War emplacement, are those that remain on the surface in pasture field. The subterranean element of the complex is now inaccessible as it has been sealed with backfilled rubble for safety purposes.



Fig 5.7.3 The former Officers Quarters and Observation Post at Robert's Battery.

The site is a rare example of an exceptionally well-preserved First World War facility with the latrines, water tower and storage blocks all still maintained by a sympathetic owner. Fort House itself retains many original period features and this, along with the other buildings present on site, represents a chance to investigate the original function and use of the site.

Further survey work was undertaken to the south of the battery, where the remains of a firing range were clearly visible, and also to the north in the vicinity of Seaton Sluice. This allowed further military features to be recorded and additional context to be added to what survived in the fields surrounding Fort House.



5.7.2 Location and geology

The site of the battery is located in southern Northumberland and the area is generally characterised by gently undulating land incised by river valleys and tributaries. Carboniferous Coal Measure rocks create this landform, stretching from south-east Northumberland through to Co. Durham, which comprises of shale and soft sandstones with numerous coal seams. Permian rocks overlying the Coal Measures then outcrop as cliffs at the coast. There are also glacial lake deposits of fine silts and clays.



Fig 5.7.4 View south to St Mary's Island from the Observation Post surviving on top of Fort House

5.7.3 Previous research

No formal investigation of Roberts's Battery has been undertaken; however, the site has been subject to research by the Fortress Study Group and a local history society. Both have investigated the history and remains of the battery although there has been no formal dissemination of the information. Reference has been made to the site in an overview of the North-East coastal defences undertaken by the Fortress Study Group (FORT 12, 1997 pp97-104.) The site has also been opened up regularly for Heritage Open Days, by the current owner, so that people could visit the site and look around.

5.7.4 NERCZA Phase 2 Archaeological Investigation

5.7.5 Prehistoric onwards

No prehistoric, or early medieval features were identified during the course of the survey

5.7.6 Medieval

The extent of medieval archaeology was contained within one field, which contained heavily graded, broad, medieval ridge-and-furrow. This is currently

occupied by horses and is set directly inland from the field containing the upstanding remains of the battery.



Fig 5.7.5 Earthwork remains of medieval ridge-and-furrow ploughing.

5.7.7 Post-Medieval

The only visible post-medieval archaeology is the field system which surrounds Fort House. This represents an enclosure field system with later alterations and additions as the settlement at Seaton Sluice grew and developed over time.

5.7.8 20th Century

The archaeology of the First World War appears to be rare along the North East coast for two reasons; it has been lost to erosion and the sites of First World War defences were redeveloped in the lead up to and during the Second World War. The surviving remains at Fort House represent the best preserved example of a First World War facility surviving on the North East coast. The water tower (438) and store buildings (439 and 441), are exceptionally well preserved and, although the original barrack blocks have long since been demolished, the current owner has built two modern garages to the original design and on the location of the former barracks, although they are aligned differently. The site therefore retains some of its original feel and character.



Fig 5.7.6 The WW1 water tower and store rooms underneath.

The associated buildings, such as store facilities (441) and a brig (439), sit within a larger surviving enclosure (436), which has been modified in some parts but as a whole retains the original outline of the camp. At the north west corner of this enclosure there is the only surviving example of a defended First World War latrine block (440). According to the owner of the property and military historians from the local history society the only other known surviving example is thought to be located in the Sudan, although this has not been confirmed. This makes this structure potentially extremely rare and significant, which is something the owner is aware of and he advocates its preservation.



Fig 5.7.7 The defended latrine block (440) with blocked up loopholes now visible.

Fort House (443) itself, although now a private dwelling, was originally the Officers and Non-Commissioned Officers quarters, the kitchens, the Officers mess, Commander's Office and Battery Observation Post. Much of this is still preserved in the internal layout of the building. The mess room and main corridor boast elaborate original features, an echo of earlier British army traditions, and the mess still has a large service hatch to allow food to be passed through which shows the massive thickness of the structure's walls.



Fig 5.7.8 Exterior of Fort House looking north.



Fig 5.7.9 Internal features including the chandelier, indicating the higher status of the Officer's Mess and a view of the corbelling in the main corridor.



Fig 5.7.10 The serving hatch, showing the thickness of the interior walls.

The house also has some other interesting visible features inside. On the upper floor, inside a recently constructed cupboard, an original wartime map of the United States can be seen on the original wall. It is possible that this was a larger map of the world, as not the entire map is visible. It could have been placed as a reminder of home for American troops who were stationed here during the Second World War. The original layout of the upper level and Observation Tower is harder to ascertain as later partition walls have been erected to make the area more suitable as bedrooms. This is something the current owner is considering returning to its original open plan form.



Fig 5.7.11 Wartime map visible inside a modern cupboard. The back wall is the original wall.



Fig 5.7.12 View north from the Observation Tower.

Other internal features include a basement level, presumed to originally have been a storage area for supplies and ammunition. There is a suggestion by the owner that there was a connecting tunnel between the house and the battery itself, although this is unconfirmed and no evidence of such a tunnel was seen during the survey. It would also be impractical to access such a tunnel due to the low ceiling level in the basement. The basement is accessed through a hatch underneath the entrance in the seaward side of the property.



Fig 5.7.13 Hatch leading into basement level with concrete stairs.



Fig 5.7.14 The basement level.

Other miscellaneous features can be seen in the garden of Fort House and these are presumably related to the original First World War camp. A ventilation pipe (437) of unknown function can be seen against the south wall of the garden. This would appear to be contemporary with or to pre-date the wall of the compound but its precise function remains unknown. Elsewhere in the garden there is evidence of low graded earthworks which could be the position of former features within the camp. The most prominent of these is (442).



Fig 5.7.15 Possible ventilation pipe 437 against the inside of the south wall.

Another outcome of the survey of Fort House was the discovery of documents held by the owner which include plans, aerial photographs, records and historical information relating to the site. One particularly interesting document is the elevation drawings and plan of the original building at Fort House showing all of the described features and entitled "Tyne Defences Hartley".



Fig 5.7.16 The original plans and elevations of Fort House as held by the owner.

Other features can also be seen in the area surrounding the battery. Along the cliff-top path several trenches, weapons pits and other features can be seen. These were clearly constructed to defend the perimeter of the battery. One trench in particular (444) is eroding as the cliff retreats and slumps. Other features include metal fittings and fixtures (447 and 528) which could relate to military activity but remain uncertain in their interpretation.



Fig 5.7.17 eroding trench (444) on the cliff edge with Fort House in the background.

Further to the south of Fort House are the earthwork remains of a firing range (503). Although the range itself is now lost within the current field system, the three large earthwork banks that held targets still remain (fig). The current Post Office long-distance transmitting masts are situated next to the former firing range, however concrete bases for at least 3 more masts, now removed, can also be seen. These could be military in origin, although there is no definitive evidence for this.



Fig 5.7.18 One of the three earthwork mounds surviving on the firing range.



Fig 5.7.19 The concrete bases of one of the removed radio masts.

The survey also investigated locations further to the north at Seaton Sluice. Here there is significant evidence for the Second World War defences Comprising two upstanding pillboxes (506 and 505), two weapons pits (507 and 508) and also the remains of a foundation platform for a pillbox (509). All of these are concentrated on Sandy Island in Seaton Sluice, presumably to defend the bridge and the sluice or harbour.



Fig 5.7.20 Circular pillbox (505) on Sandy Island, partially buried in sand.



Fig 5.7.21 Circular pillbox (506) showing brick-blocked aperture.

The upstanding remains of the Robert's Battery (451) are currently fragmentary and collapsing. The plan of the battery can still be observed, as well as graded earthworks that may represent subsidiary buildings. The underground remains are reportedly well-preserved, although these are presently too dangerous to access safely and so were not surveyed.



Fig 5.7.22 Location of Robert's Battery collapsed structures and earthwork remains.

5.7.9 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.2-0.4m per year (SMP2 for north East England). The most threatened and fragmentary remains are those of Roberts's Battery itself. The concrete rubble can still be seen protruding from a field north of Fort House, but these remains are collapsing on themselves. They still indicate the original layout of the battery but are currently unsafe to access. The erosion rate here is 0.2m per year and the upstanding remains will be at risk within 100 years. However the underground remains will be directly threatened much sooner. The underground facilities associated with the gun battery, including the gun emplacements will start to erode out of the cliff face as it retreats, as is already occurring on the cliff top, where the original defensive emplacements for the battery are already being destroyed.

5.7.10 Summary and conclusions

The SMP2 preferred policy for the Robert's Battery area is No Active Intervention and, although the erosion rate is not as severe as elsewhere on the Northumberland coast, there is a long term risk to the archaeological remains. Robert's Battery presents an opportunity for investigation of a unique military complex that has its origins in the First World War and was developed for use in the Second World War. However, it was never finished completely and was abandoned as it outlived its usefulness. This means that all of the remains of the battery date to a very specific period and would allow greater understanding of the early defence of the North East coast, particularly during the Second World War.

Manageme		Policy	Policy			
nt Area		unit	Plan			Comment
			2025	2055	2105	
MA24	Collywell	24.1	HTL	HTL	HTL	
	Bay					
MA24	Crag Point	24.2	NAI	NAI	NAI	Crag point
	to Currys					headland to
	point					remain
						undefended, local
						intervention to
						replace Hartley
						Cove steps for use
						as an emergency
						access from the
						beach and allow
						access to view the
						unbroken coal
						measures.

Table 5.7 The Shoreline Management Plan 2 policy information for Seaton Sluice and Robert's Battery.

The area to the south is threatened by the lack of sea defences and it may be that the firing range is also at least partially lost within 50 -100 years. The surviving military archaeology will retain its intrinsic value as a single entity. If any one part is lost, or threatened with loss, the understanding of the whole site will suffer. This includes the earthwork and structural remains of pillboxes found at Sandy Island. The rapid survey has recorded the location and condition of these defences and this is probably enough information to inform policy but the main battery would benefit from more detailed earthwork and architectural survey if possible safely.



Fig 5.7.23 The eroding cliffs at Hartley with St Mary's Island and lighthouse in the background.





5.8 Druridge Bay, Northumberland.

5.8.1 Background

The area of Druridge Bay (PU 17.1 - 17.5) is experiencing significant changes to its coastline. The central and southern areas of the bay are experiencing an accumulation of dunes and, while the northern end of the Bay is experiencing the effects of severe erosion. Both of these processes are causing rapid changes to the coastline. The dunes and cliff between Hadston Carrs and Low Hauxley are retreating and eroding at an extremely rapid rate. Behind the dune system is an area of low-lying agricultural land interspersed with wetland. These areas are former open-cast mining areas and this activity, combined with the ongoing dune and cliff erosion, means only a narrow strip of preserved archaeology survives.

The area has rich and varied archaeology which includes the medieval and Knights Hospitaller Preceptory Chapel at Low Chibburn, a tower house at Cresswell, varied multi-period industrial remains and also the significant prehistoric site at Low Hauxley, which is detailed in section 5.9. Druridge Bay was also one of the key sites identified as a possible invasion landing beach during the Second World War. Due to this fact there is a large quantity of wellpreserved military archaeology, much of which is eroding and under imminent threat of destruction. As this was one of the key areas identified by the Ministry of Defence which resulted in the subsequent extensive and elaborate defensive emplacements being planned for construction Druridge Bay was highlighted for further rapid survey as part of Phase 2 of the NERCZA.



Fig 5.8.3 Druridge Bay viewed from Cresswell looking north towards Low Hauxley



5.8.2 Location and geology

Druridge Bay is located on the Northumberland coast and falls between Low Hauxley to the north and Cresswell at its southern end (Fig 5.8.2). It is 9.3km in length and the current foreshore in front of the dune system is comprised of interbedded sandstone, mudstones and coal, all of which outcrop in the intertidal and foreshore area. To the rear of the dune system a huge swathe of land has been exploited for open-cast coal extraction which has meant that a narrow strip of sand dunes is the only surviving band of archaeological remains in the central and northern part of Druridge Bay. This is under active and rapid erosion from the seaward side.

5.8.3 Previous research

There has been much previous research into the archaeology of Druridge Bay and the surrounding area, mostly focused on the Bronze Age remains around Low Hauxley. These are discussed in detail as part of section 5.9. Other research has been undertaken into the palaeoenvironmental and palynological evidence available at Druridge Bay, in particular at Cresswell (Innes 1988). This is also discussed further in section 5.9 as the Phase 2 NRECZA survey covered the Low Hauxley area in greater detail.

5.8.4 NERCZA Phase 2 Archaeological Investigation

5.8.5 Prehistoric

The only surviving prehistoric evidence form Druridge Bay that the survey recorded is localised to the Low Hauxley areas surrounding the outflow of the Bondicarr burn. This will be discussed in detail in section 5.9.



Fig 5.8.4 The dune and till cliff at Low Hauxley.

5.8.6 Romano British – Early Medieval

No Romano British or Early Medieval features were recorded during the course the survey.

5.8.7 Medieval

The only medieval feature recorded by the survey at Druridge Bay was the Preceptory Chapel at Low Chibburn (454). This was originally a small priory chapel, later converted into a dower house for Widdrington Castle, and then became a small farmstead in the post-medieval period. Subsequent to this the chapel as seen further alteration when it was converted to a Second World War defensive structure and even used as shelter by those working on the nearby open cast mining operations in the post war period.

The structure is designated as a scheduled ancient monument (SAM no 1106493) and is set 0.75km back from the dune system. It lies outside the Environment Agency flood zone (EA 2007) and is not under any imminent threat from coastal erosion.

The tower house at Cresswell (Listed Building number 238155) is a second medieval feature in this area but it is not at threat from erosion. The survey did not record this as part NERCZA Phase 2 as it has already been accurately mapped as part of the Phase 1 process.



Fig 5.8.5 Part of the original Preceptory at Low Chibburn, with Second World War alterations to windows.

5.8.8 Post-Medieval

At Cresswell the earthwork remains of at least four, and possibly five, postmedieval enclosures can be seen (258). The stone wall which faces the road still has the gateways to these visible. This part of the village was still occupied in the 1860s and was abandoned shortly after; these can be seen on 1st and 2nd edition of Ordnance Survey mapping. This small scale abandonment could be as a result of local people leaving small rural settlements to move to larger urban areas to find work in the later 19th and early 20th century.



Fig 5.8.6 One of post-medieval enclosures for a small house at Cresswell.

Other post-medieval structures can be seen eroding out of the cliff at Low Hauxley although these are considered to be military in origin (318, 332 and 360.) The preceptory chapel of Low Chibburn (454) was also altered in the postmedieval period, as discussed above.



Fig 5.8.7 The later Dower house at Low Chibburn

5.8.9 20th Century

Druridge Bay contains a large quantity of surviving Second World War military archaeology, even considering the recent open-casting activity and the ongoing erosion. Almost every aspect of coastal defence is covered between Creswell and Low Hauxley including observation posts, gun emplacements, pillboxes, and antitank defences. The most obvious and common military archaeology is the remains of lines of anti-tanks defences. These defences can be seen intermittently along the full length of Druridge Bay. They are less concentrated in the southern and central areas, but what can be seen is mostly *in situ*, whilst to the north more survive, but they have often been moved off the beach. An example of this can be seen at the outflow of the Bondicarr Burn.

Pillboxes frequently survive along the length of the Bay and the Phase 2 survey recorded the partial remains of 12 pillboxes, including a well-preserved, disguised pillbox on the road behind the dunes (253). Three structures previously thought to have been pillboxes (269, 266 and 300) were found to be observation posts for a former bombing range out in Druridge Bay itself. They can be distinguished by the very large entrance and apertures in the front of the structure. Pillboxes always have their entrance at the rear as they are easily defensible.



Fig 5.8.9 A line of 13 anti-tank blocks exposed in the mouth of a small burn.

In the dunes, especially in the area of Druridge Bay country park, there are substantial surviving earthworks of large anti-tank ditches (322, 296, and 299 as

well as others). These large ditches are preserved at the road side between Ladyburn Lake and Hadston Carrs. These are over 1m deep in places and at least 2m wide. They were designed to slow or trap a tank sufficiently so that infantry could place an explosive charge on the front of the vehicle.



Fig 5.8.10 One of the observation posts. Note the large blocked entrance and aperture at the front, distinguishing it from a pillbox.



Fig 5.8.11 Disguised pillbox located north of Blakemoor Farm.

As well as many of these larger structural elements, many earthwork components survive, particularly where the dunes have stabilised or around small settlements. In Cresswell there are a series of weapons pits preserved both on the village green (241) and along the roadside next to the caravan park (239). Such features are less well-preserved as most of the landscape is dunes, so ephemeral features such as weapons pits and trenches do not survive particularly well. There are also the remains of a small military camp at Cresswell (244), close to the fragmentary remains of a battery located to its north (264). These are both at high risk of future erosion and are in locations were the dunes are slightly less stable than elsewhere in the Bay area.



Fig 5.8.12 Earthwork remains of military camp at Cresswell

The most surprising surviving remains were the earthwork remains of at least two bombing range markers (291 and 280). These features are unusual because what survive are not the structures themselves but the scars left from their removal. They would have original been a simple series of concrete blocks surrounding a painted, raised, wooden arrow indicating the direction of the bombing range. What survives now is the removal scars representing the position of the blocks and the base of the arrow. When recorded as part of the Phase 2 survey they corresponded exactly with what had been transcribed as part of the Phase 1 aerial photography transcription.

Several other structures were recorded including a searchlight base and a possible range finder base (Fig 5.8.13). The current survey represents a comprehensive catalogue of what survives from the Second World War at Druridge Bay.



Fig 5.8.14 Part of the bombing range marker foundation (291)

5.8.10 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.1m per year (SMP2 for north East England). Coastline alteration by natural processes are occurring throughout Druridge Bay although the erosion at the north end of the bay is clearly having much more serious impacts upon the
archaeology than the sand dune formation to the central and southern areas. Comparison of the position of military archaeology recorded by Phase 2 with the position of the same features as recorded by Phase 2 was undertaken, as was done at Crimdon Dene. This showed that although serious erosion is ongoing in the north, the southern and central areas of Druridge Bay show accumulation of sand deposits. An example of this is the observation posts that originally overlooked the Bay are now located behind extensive dunes which obscure the view they would have originally commanded.



Fig 5.8.15 Pillbox exposed in dune in central Druridge Bay.

Localised destabilisation of the dunes is revealing various military features all the time and this places them at risk of slumping and collapse. The changeable nature of the dune landscape also means that the Phase2 survey may not have recorded all military archaeology that survives. New features are often revealed while others hidden by dumps of sand and dune blow outs, meaning that what was recorded by NERCZA is only a current condition statement for what was visible in Winter 2009.



Fig 5.8.16 eroding military structure.

5.8.11 Summary and conclusions

As discussed above the erosion at Druridge Bay is serious and ongoing. Apart from Low Hauxley and Cresswell, which are designated as Hold The Line the rest of Druridge Bay has a policy of Managed Retreat in SMP2. Little will be done to protect the exposed dune land archaeology and there is even investigation into managed tidal flooding to reduce impacts on other stretches of coast. These processes will have a serious detrimental impact on the heritage of this stretch of coastline

Manageme		Policy				
nt Area		unit	Policy Plan			Comment
			2025	2055	2105	
						Develop a progressive transitional approach to defence inline with erosion pressure to
MA17	Beacon hill links	17.1	MR	MR	MR	main village and its access
MA17	Low Hauxley	17.2	HTL	HTL	HTL	With the probable need to realign the southern end
MA17	Druridge Bay north	17.3	MR	MR	MR	Develop drainage plan and access the southern end
MA17	Druridge Bay south	17.4	MR	MR	MR	Examine potential for tidal flooding inland
MA17	Cresswell	17.5	HTL	HTL	HTL	

Table 5.8 Shoreline Management Plan 2 policy plans for the Druridge Bay area.

Should this policy be adopted further monitoring of the condition of the military archaeology should be undertaken, particularly as it is not clear if the results of this survey includes all of the military features in the Bay area, as such features are being constantly exposed and hidden due to dune movement. If invasive coastal defences are implemented to flood the hinterland of the dunes further archaeological investigation should be undertaken to evaluate the resulting effects on the military archaeology of Druridge Bay as well as the prehistoric archaeology at Low Hauxley.



Fig 5.8.17 Druridge Bay looking south from Low Hauxley





5.9 Low Hauxley, Northumberland.

5.9.1 Background

The Low Hauxley coastline has been identified as a high risk area for archaeological and palaeoenvironmental remains to be impacted upon by coastal erosion. As a result of earlier work (see previous work section below) and the North East Rapid Coastal Zone Assessment Phase 1, Druridge Bay, and particularly its northern end around Low Hauxley, has been highlighted as requiring an urgent archaeological response.

The main archaeological site consists of a Beaker period-Early Bronze Age cairn cemetery and underlying Mesolithic site (HER number 5604), although other archaeological features are known to the immediate north and south of this site. The cliffs and foreshore at Low Hauxley are also designated as a Site of Special Scientific Interest (SSSI) on account of their importance to Quaternary studies represented by the exposure of bedrock, glacial till, peat and dune within the eroding cliff. The archaeological remains are not themselves designated in any way. However, any impacts on the archaeological remains will also cause an impact on the SSSI.

5.9.2 Location and geology

Low Hauxley is situated at the north end of Druridge Bay in mid-Northumberland 2.5km south-east of Amble. The Low Hauxley area has been severely affected by 'coastal squeeze' as open-cast coal mining has taken place over very extensive areas on the landward side of the coastal strip with on-going coastal erosion taking place on its seaward side. The sites at Low Hauxley occupy a localised natural high point which is flanked to the north and south by separate organic sediment units, usually described as 'peat', that appear to have started to form as wetlands during the Neolithic in the 4th millennium cal BC. The strip of surviving dune system in this area varies between 15m and 50m wide.

The geology of the Low Hauxley site consists of interbedded Carboniferous deposits of sandstone and coal measures overlain by glacial till. Immediately overlying the till is a sealed ancient land surface that dates to the Mesolithic period and upon which a Mesolithic occupation site survives. This ground surface appears to have accumulated in depth over time so that, by the Beaker period, burial pits were cut down through the land surface into the till below with cists constructed and cairns raised above them to form a cemetery. Subsequent to the on-set of dune formation in the Bronze Age the burial cairns and land surface were covered by 3-4m of wind blown sand. However, this accumulation did not take place as a single event as organic lenses can be noted at various heights in the sand dune cliff sections and these represent episodes of stability when a vegetation cover developed before further dune accumulation took place. Currently the dune system is stable and overlain by a thin soil with marram grass cover.

On the lower lying ground to either side of the high point, peat formation took place from at least as early as the Neolithic period in what appear to have been areas of lagoon. Dating samples from the top and base of each of these peat beds have been recovered for scientific dating as part of the Phase 2 NERCZA project. The peats contain an important palaeobotanical resource that includes a variety of plant remains, invertebrates and pollen evidence together with flint tools, worked wood and, potentially, other archaeological material. Several hewn timbers have been reported from the peat beds further north (Low Hauxley A) where axe or adze marks have been noted (e.g. Jim Nesbitt pers. comm.). Other areas of discrete peat beds have been noted at Low Hauxley and the north end of Druridge Bay, including the new exposure noted at a lower elevation to the south of the cemetery site in the inter-tidal zone. This deposit hosts preserved human and animal footprints (see below).



5.9.3 Previous research

The various archaeological interventions are summarised in the following list:

- 1. 1983 Excavation of an eroding cist and single inhumation by Steve Speak of Tyne and Wear Museums Service. One page text summary produced.
- 1983 Excavation at the cliff face by Clive Bonsall (Edinburgh University) of Bronze Age burials and Mesolithic flint scatter, bone and shell material. The burials came from below the same cairn as the burial excavated by Speak. Nothing published apart from a very short notice in Proceedings of the Prehistoric Society (Bonsall 1984).
- 3. 1993 Excavation of two stone burial cists found eroding from the cliff face, also from beneath Bonsall's 'Cairn 1'. One cist contained a cremation and one an inhumation and each was accompanied by a very fine and well-preserved Beaker. Undertaken by Tyne and Wear Museums Service (TWMS). Stratigraphy report was produced.
- 4. 1994 Evaluation excavation behind and off-set from the eroding face of the Bronze Age cemetery by Lancaster University Archaeological Unit. Detailed Archive Report and Appendix produced. Little archaeology was recorded apart from a flint assemblage, as well as more modern material. An assessment of the peat was also undertaken together with radiocarbon dating of the skeletal material from the Bonsall excavation.
- 5. 2007 Photographic recording of an eroding stone-built structure sealed by the dune sand *c*.35m to the north of the cemetery site and photographic recording of rectangular rock-cut pits on the foreshore in front of the cemetery site by Jim Nesbitt. Photographs held by Northumberland HER and by Archaeological Research Services Ltd.
- 6. 2009 Excavation of a small, badly-eroded stone cist holding a few fragmentary remains of a cremation in a newly eroded section of cliff face 2m north of the TWMS cist excavations, and therefore presumably from below part of 'Cairn 1' along with excavation of a second, separate, badly-eroded cremation in a pit burial located in a newly eroded section of cliff face 5m south of the TWMS cist excavations. The work was undertaken by Archaeological Research Services Ltd and an Archive report produced. Radiocarbon date obtained on cremation 2 and one for cremation 1.
- 7. 2010 Photographic and rapid survey of the Druridge Bay coastline which has identified many previously unrecorded Second World War sites together with an area of human and animal footprints brought to ARS Ltd's attention by Jim Nesbitt. The footprints were found within a newly exposed peat bed 25m to the south of the Bondicarr Burn outflow. Precision survey of the extent of each peat exposure, sampling of the top and base of each for radiocarbon dating also took place and radiocarbon dates are awaited. A rapid photographic survey along the length of the Low Hauxley cliff line has also been undertaken.

A series of palaeoenvironmental studies have been undertaken on the various organic deposits visible in the Low Hauxley cliff sections on the soils and sediments. These include the published work of Frank (1982), Innes and Frank

(1988) and Farrimond and Flanagan (1996) and the unpublished work undertaken as part of the Bronze Age cemetery investigations (Huntley 1995; Issitt *et al.* 1995; Payton and Usai 1995; Tipping 1994). At Amble Bay and Cresswell Ponds, both in Druridge Bay, Shennan *et al.* (2000) have cored for dating samples to provide past sea level index points whilst Wilson *et al.* (2001) have made a study of Late-Holocene dune development along the Northumberland coast including the dune system at Druridge Bay.

5.9.4 NERCZA Phase 2 Archaeological Investigation

5.9.5 Prehistoric

Excavation of two Bronze Age burial features was undertaken by Archaeological Research Services Ltd. in 2009 as part of the NERCZA Phase 2 work. This small-scale excavation recorded two graves. Burial 1 was a small stone-built grave box, or tiny cist, made from small sandstone slabs wedged into a pit that had been cut into the glacial till and then covered with a low stone cairn. A depth of 3.5m of sand dune accumulation has since built up above the cairn. Inside the grave box, or small 'cist', had been a cremation, traces of which still survived in the stone-lined cavity. This material was collected for analysis and dating. At the foot of the cliff immediately below the grave box was a small pile of cremated human bone and it is reasonable to assume that this is material that has fallen out from the grave box. However, this had been intermingled with the beach sand as successive tides had washed up to the cliff face. This material was not collected as its true provenance could not be ascertained. However, because the grave box was starkly visible in the cliff face the position of this cremation debris below the grave box is also consistent with an inverted ceramic vessel having been removed from the grave box by a light-fingered passer-by and the cremation material falling to the floor on removal. Although this is not known with certainty, the fact that a stone had been recently wedged across the cist to hide it from view led the excavators to believe this to be a likely scenario.



Fig. 5.9.2. The small cist containing a human cremation that also lay under Bonsall's Cairn 1.

Burial 2 was a grave comprising a pit burial that had partly eroded from the cliff face. This pit was not located underneath any observable cairn although there is a cairn to the immediate north of this burial which is probably Bonsall's 'satellite cairn' or Cairn 2. A pit had been cut into the glacial till and a plain Beaker had been placed inside containing a human cremation, together with a dump of the pyre debris that had been scraped up. This pyre debris was very black and contained much charred debris and grey ash that was probably still hot when it was deposited as the heat has turned part of the Beaker pot a pale grey colour. A few Mesolithic flints had been scraped up with the pyre debris and deposited in the pit with this material which implies that the funeral pyre was situated on the ground and the gathering up of the remains included the scraping up of material from the underlying Mesolithic ground surface. Single-entity, long bone fragment, radiocarbon dating samples were submitted for each burial. Cremation burial 2 has returned an Early Bronze Age date of 1890-1690 cal BC at 95% confidence while Cremation burial 1 has returned an early Bronze Age date of 2140-1890 cal BC at 95% confidence (see Table 5.9.1)..

Intervention	Sample	laboratory	$\delta^{13}C$	Radiocarbon	Weighted	calibrated date
		couc	(/00)	age (DI)	Wicall	(95%)
						confidence)
Bonsall 1983	Burial 1	OxA-5553	-	3615 ±45	3621 ±34	2140-1890 cal BC
			20.6			
Bonsall 1983	Burial 1	OxA-5553	-	3630 ± 55		
			20.8			
Bonsall 1983	Burial 2	OxA-5553	-	3410 ± 55	3420 ± 38	1880-1640 cal BC
			20.5			
Bonsall 1983	Burial 2	OxA-5553	-	3430 ±55		
			20.6			
ARS Ltd	Burial 2	SUERC-	24.7	3470 ± 30	-	1890–1690 cal
2009	[011] -	27330				BC
	cremation					
ARS Ltd	Burial 1	OxA-22476	-	3569 ± 28	3569 ± 22	2010–1875 cal
2009	[009]-		25.3			BC
	cremation					
ARS Ltd	Burial 1	SUERC-	-	3570 ± 35	-	-
2009	[009]-	28741	24.5			
	cremation					

Table 5.9.1. Summary of radiocarbon dates from the various archaeological interventions at Low Hauxley.

During the survey work at Druridge Bay amateur archaeologist Jim Nesbitt drew to the attention of the project team a freshly exposed inter-tidal peat lying immediately south of the Bondicarr Burn outflow. In a tightly defined area of this peat an abundance of human and animal footprints could be observed pressed quite deeply into the peat surface. This peat lies at a lower level than the peats that flank the cemetery site and so was thought to be potentially earlier. Samples from the top and base of this thin peat horizon were taken and the following radiocarbon dates were returned. 5330-5210 cal BC (basal) and 5220-4990 cal BC (top), a detailed discussion of these dates is contained in section 6.3 of this report. Archaeological Research Services Ltd surveyed the extent of the footprint area during a rising tide that was depositing sand back over the peat. The sand has now accumulated to a depth of *c*.0.5m and the peat is currently buried with no surface remains visible except for the occasional tree stump that protrudes through the sand. Accurate recording of these footprints remains an urgent priority next time the surface is scoured clean after a storm event. A sample of wood was also retrieved from this deposit and upon cleaning and further analysis was observed to have cut marks, potentially made by a stone tool (Taylor 2010). This sample is also discussed in section 6.3 and an assessment of its significance is made in Chapter 7.



Fig. 5.9.3 Human footprints filled with sand and pebbles deeply impressed within the peat that survives as a thin layer within the inter-tidal zone.



Fig. 5.9.4. Examples of some of the animal footprints that can be seen amongst the human footprints.

A full photographic survey of the cliff face was undertaken (Fig 5.9.5), providing a full visual record of the eroding peat in one continuous section that can provide a bench mark against which future erosion can be compared and assessed.



Fig 5.9.5 An excert from the full section of photographed peat layers.



Fig 5.9.6 Location of the Tyne and Wear interventions at Low Hauxley

The recording also included dune accumulation over the possible wider extent of the Low Hauxley burial ground. These included the earthwork remains of 6 possible locations for cairns (344, 345, 346, 347, 348 and 349). This interpretation is however uncertain and confirmation of this would have to be established through evaluation trenching. Several pieces of worked flint were also collected from the top of areas of exposed peat. It is unclear whether they were eroded out of the cliff face or simply exposed from the erosion of the lower peat deposits (326 and 330).



Fig 5.9.7 Mapping the visible peat at Low Hauxley

5.9.6 Romano British period onwards

No Romano British, early Medieval, or Medieval features were recorded during the course of the survey along this stretch of coastline.

5.9.7 Post-Medieval

Several post-medieval features were exposed in the cliff face, including a stone wall revealed on the cliff top edge (360). Very little of this is currently revealed and the wall appears to be of hand-hewn sandstone. It could be a wall, or possibly a small structure, but so little is extant that a definitive interpretation is impossible.



Fig 5.9.8 recording the extent of sandstone wall 360.

The other features identified at Low Hauxley that are presumably of postmedieval date are a series of square and rectangular rock cut features. These are of unknown date or function but are probably related to the fishing industry. They could be Bratt holes or Hulleys (see section 5.3), used for the storage of bait and used to keep a catch fresh. Alternatively, they could be fishing traps of some description or used to store fishing equipment, although the latter is unlikely due to their positioning in the inter-tidal zone.



Fig 5.9.9 One of the square rock cut features at Low Hauxley

5.9.8 20th Century

The twentieth century archaeology visible at Low Hauxley, as with many of the other sites surveyed, mostly dates from the Second World War. The majority of visible surviving remains are anti-tank blocks. They survive both *in situ* and moved out of context, most notably to form part of the outflow of the Bondicarr Burn (322). Many of these anti-tank blocks are heavily eroded and barely recognisable, apart from their basic shape and construction material (316). This is in sharp contrast to those visible at the southern end of Druridge Bay where they survive intact and *in situ*.



Fig 5.9.10 Anti-tank blocks used as part of the outflow of the Bondicarr Burn.



Fig 5.9.11 A heavily eroded anti-tank block in the make up of a storm beach at Low Hauxley.

A series of more ephemeral Second World War earthwork features were also seen at the top of the cliff. Although not as extensive as elsewhere along the coast several weapons pits were visible (350). Second World War archaeology could also be seen eroding on to the beach, where pieces of picket wire and barbed wire could be seen (361).



Fig 5.9.12, Section of picket wire found eroded onto the beach.

5.9.9 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.4m per year (SMP2 for north East England). The coastal strip at Low Hauxley, where the eroding archaeological remains have come from, lies in Policy Unit 17.3 'Druridge Bay North' (formerly 32 under SMP1) of DEFRAs Shoreline Management Plan 2 (SMP2). The preferred Policy Recommendation for this policy unit is "Managed Realignment" up to the years 2025, 2055 and 2105. The term 'Managed Realignment' is defined in SMP2 as "Allowing the shoreline to realign, landwards or seawards, sometimes with management to initiate and control change" (Guthrie *et al.* 2009, ii). Given that the land to the rear of the existing dune system has been removed by opencast coal mining there is limited scope for dune roll back to take place in this section of the coastline, and such roll back would itself give rise to the destruction of the archaeological and palaeoenvironmental deposits.

The SMP2 report also contains estimates of baseline erosion rates at various points. These are based on existing data and may be expected to increase with sea level rise. Accordingly, the figures presented in Table 5.9.2 below should be taken as a minimum. The erosion over the last year alone at Low Hauxley as recorded by the NERCZA survey is considered to vary between 0.5m-1.5m as the erosion has been particularly acute over the last 18 months and there is no sign that this rate is decreasing. The eroding cemetery area is situated in the Bondi Carrs section according to the table below.

Location	NGR	Rate per	Over 100
	(approximate)	year	years
Low Hauxley	NU287028	0.4m	85m
Bondi Carrs	NU286020	0.5m	80m
Hadston Carrs	NU280010	0.5m	70m
Druridge Bay	NZ277960	0.1m	15m

Table 5.9.2. Rates of coastal erosion taken from SMP2 report (Guthrie et al. 2009, 190).

The rates of erosion produced for this table are based on assumed seal level rise rates of 0.05m to year 2025, 0.26m to year 2055, 0.8m to year 2105. However, the latest minimum sea level rise estimates forecast in the official UK Climate Projections published by DEFRA are for a rise of 0.5m – 2m by 2050 for Northumberland. If this new estimate is accurate, and it is widely acknowledged as a minimum, then the annual erosion rates need to be revised upwards by at least twice as much. On such a basis the cemetery site and peat exposures can be expected to erode at a rate of around 1m per year and this is in keeping with the observations made during the course of the NERCZA project during 2008-2010. The threats to the site can be characterised as:

- Direct erosion by wave action of the exposed peat and Mesolithic settlement/Bronze Age cemetery site
- Direct erosion of the Mesolithic Peat and the remains of Mesolithic human and animal footprints and a large worked assemblage
- Indirect erosion by wave action caused by undercutting of the dunes and then slumping from above. This is how the earlier cists were exposed and all deposits are at risk from this form of erosion.
- Many people know the exact location of the cemetery site and both the author and amateur archaeologist Jim Nesbitt, the latter of whom monitors the site regularly, are convinced that some remains have been robbed from the site. This was suggested by the circumstances of discovery of ARS Ltd's Burial 1 where a cist stone that had fallen to the ground had been clearly wedged back in to cover the void where the cremation had been to hide it from view. A deposit of cremated material immediately below the cist appears to have been dragged out or to have fallen out of an inverted pot that was extracted from the cist.

None of these threats face any real prospect of being able to be controlled and so evaluation of what still survives on the site followed by a programme of area or targeted excavation and/or further monitoring remain the most viable options. However, the weakness of the monitoring approach is that it leaves costs and commitments open-ended, creates a highly fragmented and limited archaeological record, as well as missing material removed by erosion and robbing of the site.

5.9.10 Summary and conclusions

Mesolithic activity is documented along the entire length of the North-East coast but the evidence consists mostly of surface flint scatter sites, and no Mesolithic ground surface hosting a lithic scatter, animal bone and marine shell has been found anywhere else apart from at Low Hauxley. Such sites are also rare nationally. The survival of in situ archaeological deposits of this period make Low Hauxley nationally important on account of the Mesolithic archaeology alone. It is possible that Mesolithic structural remains may also survive, such as hearth pits or evidence for a structure, and, having been sealed by the Beaker period and Bronze Age burial cairns and subsequently the dune deposits, the Mesolithic archaeology remains relatively undisturbed and in situ. The Mesolithic deposits are covered by calcareous dune sand which has an alkaline bias and which the various interventions have already demonstrated allows for good preservation of organic material, including fish bone. Thus, the site can be ascribed national importance on the grounds of period, rarity, preservation, condition, vulnerability as well as the diversity of the site, given that it also contains a pristine Beaker/Early Bronze Age cemetery and stratigraphically associated organic sediments to either side.

The Beaker/Bronze Age cemetery comprises a group of stone cairns of unknown number overlying cist and pit burials that have already exhibited a range of mortuary practices including crouched inhumation and cremation. Burial monuments of this period are relatively common in the archaeological record of the region, although they are usually heavily robbed and many have been disturbed or robbed by antiquarian and illicit diggings. The cemetery at Low Hauxley is of special importance because the cemetery survives as an intact group on what appears to have been a small island or headland raised up above a surrounding wetland/marsh. Moreover, as the cairns have been completely sealed by wind blown sand of up to 4.5m depth, the cemetery that survives is preserved in pristine condition. Not only do the structural features appear intact but the calcareous nature of the sand has meant that survival of bone and other organic material is excellent. The Beaker pots also show a remarkable condition of preservation. To find such a well-preserved and sealed Bronze Age cemetery is a discovery of national importance on the grounds of preservation, condition, vulnerability and potential, as well as the diversity of the site given that it also contains a Mesolithic site and is stratigraphically associated with contemporary organic sediments to either side.

As far back as 1995 the Lancaster University Archaeological Unit made the following point in their report on the site in the opening paragraphs of their Recommendations section,

"The point should be forcibly made that the constant and escalating threat to this site, namely coastal erosion, will not lessen without human influence, and that the site will be destroyed in the near future."

This statement is even more prescient now than it was in 1995.



5.10 Alnmouth, Northumberland.

5.10.1 Background

The HER records that a quantity of slaggy material has been noted eroding out of the sides of an embanked promontory known as 'Pan Close' beside the River Aln at Alnmouth. A number of mounds to the east have been identified as sleeching tips and the area is marked as 'Saltings' on the 1:10,000 OS Map. This site is being eroded by the river and the SMP2 data show the whole area to be at risk of flooding. The 'Preferred Strategic Option' here is to 'Selectively Hold The Line' and as a result these sites could be at risk from construction work. The oyster beds which lie adjacent to the Alnmouth salt working site are subject to the same threat. It was the aim of the rapid survey to:

- Photograph, measure and record the oyster ponds.
- Ascertain whether the sites are suffering active erosion.
- Ascertain the nature of any damage to the features by current processes of erosion.
- Evaluate both the present level of threat to those sites and the nature of those threats.

Also the environs and wider landscape was investigated to add context to the survey. The scheduled enclosure (SAM number 8180) on the hilltop to the north of Alnmouth were investigated, along with the south bank of the Aln and additional Second World War archaeology including a guano storage building converted into a section post/pillbox.



Fig 5.10.3 Recording the location of one of the oyster beds



5.10.2 Location and geology

The town of Alnmouth is located on the North Sea coast at the mouth of the River Aln, 31 miles south of Berwick-upon-Tweed and 29 miles to the north of Newcastle (NU 243099) (PU 13.1-13.9). The settlement lies along a spur of land between the estuary of the River Aln and the North Sea, at the very southern end of a series of low hills. The local superficial geology is predominantly boulder clay which overlies solid geology of Carboniferous limestone. This forms the hilly, rolling landscape which runs down the North Sea coast from the Anglo-Scottish border and includes Alnmouth. The estuary of the Aln has been historically prone to change. The most notable alteration to the course of the river in historic times occurred during a great storm on Christmas Day 1806, when the river broke through to the North Sea in an area of low fields to the south of the town and north of Church Hill. The shape of the estuary has fluctuated ever since. On the south side of Alnmouth the dune systems are also shifting as wind-blown sand creates and destroys high dunes behind the beach. Both wind and sea are significant threats to archaeological remains in the area. (Willaims, 2007).



Fig 5.10.4 Modern cross on top of Church Hill, overlooking Alnmouth. In the foreground is the foundation of a removed hexagonal pillbox.

5.10.3 Previous research

Much research has been undertaken into the town of Alnmouth itself and its evolution over time. This is particularly significant when considering the fluctuating course of the River Aln. Local history groups and several antiquarian articles have been published on the development of the town along with the oyster beds. However no modern, detailed study of the archaeology has been undertaken.

The surviving traces of these 'oyster beds' (ponds) were first noted in the modern era in the unpublished research work of G. Bettess 1994, whilst brief descriptions, survey data and a basic chronology were subsequently published under the authorship of G. and F. Bettess, 2004.

Bettess (2004) noted that, of the five 'Oyster Ponds' recorded as extant in 1865, the remains of only the two could still be detected around a century later. No accurate date for the installation of these surviving ponds was established, although they clearly pre-date the 1st edition O.S. maps published in 1865. It was recorded that the operations there 'did not last long, and the project had to be abandoned because silt and mud built up around the beds'. This conclusion seems to have been based upon the remarks of George Tate (1805-1871) published by the Berwickshire Naturalists Field Club (BNFC) in 1877 (Osler 2009). There is also a record of an 'Oyster house' overlying one of the ponds, although the exact form and location of this building is unknown (Osler 2009). There has also been investigation into the hilltop enclosure known as the Nightfold on Beacon Hill. This unusual hilltop enclosure is a Scheduled Ancient Monument (SAM/OCN No Northumberland 31) and was subject to geophysical survey in 1993 and, prior to this, limited excavation in 1969. Both of these failed to find any definitive evidence of occupation.

The main piece of work has been an extensive urban survey undertaken by English Heritage for Northumberland County Council. The report summarised the development of Alnmouth from documentary, cartographic and archaeological sources. It also assessed the detailed archaeological potential of the town and how any future development could impact on the significant archaeological resources. The project did this with reference to the national and local planning process and its regard to archaeology.



Fig 5.10.5 The enclosure bank at Beacon Hill.

5.10.4 NERCZA Phase 2 Archaeological Investigation

5.10.5 Prehistoric

The NERCZA survey did complete a rapid assessment of the enclosure at Beacon Hill, previously thought to be Iron Age. This enclosure has been heavily disturbed by later activity, including the construction of at least one bunker and a green for Alnmouth golf course. However the shape and location of the enclosure do not seem to relate to an Iron Age settlement. A more likely interpretation is that there are 3 separate elements to the earthworks. There is a track (220), still in use that relates to a quarry to the east (221), which is presumed to be post-medieval in date. However the quarry track forms the southern bank of the Iron Age enclosure and it does not appear to cut or ride over the eastern edge of the enclosure (218), which appears to end naturally. This means that the southern edge of the enclosure as previously defined is incorrect.

Also the enclosure is in the incorrect position for a traditional Iron Age or other prehistoric enclosure as the hilltop would make a more suitable location rather than the base of a slope, where it is located. It is more likely that this enclosure (218) is a medieval or earlier boundary that actually continued to the north-west following the same contour, but has now been lost to the north of the road due to later ploughing. This could be the boundary to different land holdings or possibly to medieval parkland. If this was an enclosure a significant boundary would not be needed on the south side as there is a steep drop into a stream valley. The internal features of the enclosure (351 and 219) represent a golf course green and bunker respectively and are much later additions.



Fig 5.10.6 The termination of the boundary (218)

The clearest evidence for 218 representing an earlier field boundary is on Thomas' Wilkins map of 1794 (Williams 2007). This shows the northern boundary of Shepherd's Hill and the eastern boundary of Innt Close (also marked on Wilkin's map) could be linked with the boundary formed by 218. This could indicate that this is not a prehistoric boundary but a medieval or early postmedieval field boundary. Further work needs to be undertaken to establish the true date and function of these earthworks.

5.10.6 Romano British onwards

No Romano British or early medieval features were identified during the course of the survey

5.10.7 Medieval

Proximal to the Alnmouth oyster beds, enclosed by the original post-medieval retaining wall (209), is an area of graded broad ridge and furrow (210), visible from the path alongside the retaining wall. Another area can also be seen on the west side of the Aln Bridge. The visible remains have a clear reverse 'S' alignment and can be broadly dated as medieval as a result. Here the salt marsh encroachment respects the former field of medieval ploughing and pronounced ridges can still be seen. This demonstrates the significant effect the change in course of the River Aln has had on the surrounding landscape.



Fig 5.10.7 Remnant medieval ridge and furrow still clearly visible in a field behind the postmedieval wall.

The evidence of saltings can also be seen on the Ordnance Survey mapping in the place name evidence and it seem likely that the surrounding earthwork mounds represent medieval or post-medieval saltings. Other features in the vicinity, such as the chapel at the foot of Church Hill, may have their origins in the medieval period but what survives now is the remains of the post-medieval rebuild after the catastrophic change in the course of the River Aln in the 18th century. The position of the original church is visible Thomas' Wilkins map of 1794.



Fig 5.10.8 The post medieval chapel at Church Hill.

5.10.8 Post-Medieval

The most significant post-medieval remains are the upstanding posts that represent the remains of the oyster beds. Two complete sub-rectangular beds (223 and 224) are visible and a third is partially exposed (225). Also visible are two more fragmentary post alignments (226 and 227) which are less coherent. All of these relate to oyster beds mapped on the 1st edition Ordnance Survey map and their interpretation as such is certain (Osler 2009). The oyster beds probably relate to industrial activity during the early 19th century as they are not visible on Thomas' Wilkins map of 1794, which pre-dates the realignment of the River Aln and shows other detail, such as the former church at Church Hill. The oyster beds also most likely post-date the storm event which realigned the river as they are visible on the 1st edition Ordnance Survey map of 1860. This broadly dates these structures to having been implemented between 1794 and 1860 and having a relatively short lifespan.



Fig 5.10.9 The extant Oyster Beds

Other post-medieval archaeology was also recorded by the survey, most of which was structural in nature. The dilapidated, roofless Chapel at the foot of Chapel Hill is now set within National Trust. The architectural detail and main structure (232) survives along with a hollow way (234) that leads up from what was formerly the river bank prior to the River Aln's realignment.



Fig 5.10.10 The chapel and hollow way viewed from the east

Other post-medieval structural remains included that of a Victorian gun battery (214) on the hill overlooking Alnmouth golf course. This structure survives exceptionally well in its original form. There are some obvious later Second World War modifications but the core original structure still survives. Other earthworks (215 and 216) also survive and could relate to other element s of the battery or could be Second World War features. A precise date for the battery can be gained from the preserved dedication stone which cites:

"This battery was erected by His Grace, Algernon, Duke of Northumberland K.G for the use of the Percy Artillery Volunteer. Completed 12^{th} March 1861"



Fig 5.10.11 The dedication stone on Alnmouth gun battery

Local stories recount the competitive nature of gunnery practice between Alnmouth battery and the Amble rocketery battery. One specific tale says that upon defeat by the Alnmouth gunners the Amble team rioted in the streets of Alnmouth.



Fig 5.10.12 Alnmouth battery showing original features and later loop holes.

To the south of the River Aln a long, narrow barn-like structure survives (237) and is thought to be a storage building for imported guano. The building survives along with the earthwork remains of a former access road (238). As with the battery above Alnmouth links there are Second World War alterations to this building. Loopholes have been punched into this structure to make it more defensible.



Fig 5.10.13 Former guano storage shed south of the River Aln.

5.10.9 20th Century

As with elsewhere along the coast the largest percentage of surviving archaeological remains in the area of Alnmouth relate to the Second World War defence of the coast. Anti-tank blocks (212, 222 and 235), remains of pillboxes (233 and 215), a road block (213) and alterations to earlier structures (237 and 214) can all be seen. The earthwork remains of military buildings also survive (236 and 215).



Fig 5.10.14 A Second World War loophole placed in the wall of the post-medieval guano shed.

All of these features are in stable condition with the exception of the anti-tank blocks which are mostly exposed on the beach. However, a dilapidated gun emplacement below Chapel Hill (230) is facing a much greater degree of threat. This structure is in a poor condition, at risk from flooding and could represent a disguised pillbox or gun emplacement. A positive identification of the structure has not been possible as a result. It is able to discern that the structure was constructed out of concrete and there is also evidence for loopholes and other military features. However, it appears that there has been some attempt to disguise the windows as they have been created to look like those of the chapel, although they are made from cast concrete. If this structure was a disguised pillbox or anti-tank battery it could potentially be an important lost military feature.



Fig 5.10.15 Fragmentary remains of a military building, possibly disguised, south of the River Aln.

5.10.10 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.1m per year (SMP2 for north East England) although the make up of the dunes makes this figure variable. The threat from erosion at Alnmouth is considerable in certain areas, especially with the added risk of flooding and the fluctuating course of the River Aln. This places the oyster beds at high risk of further degradation. They are well within the flood zone mapped out by the Environment Agency in 2007 (SMP2 Northumberland). However, the remains on the south bank of the Aln are also under threat from flooding and all recorded archaeology in this area, with the exception of the pillbox on top of Chapel Hill, are within the Environment Agency flood zone. To the north of the river, as most of the recorded remains are on higher ground, the threat is less in the short term. However, increased tidal levels and storm events will also increase the rate of coastal retreat and place monuments such as the Alnmouth battery under threat in the longer term.

5.10.11 Summary and conclusions

The archaeology of the Alnmouth area is mostly post-medieval and later in date. This is certainly true of the most threatened archaeological features in the intertidal zone and in the flood zone projected by the Environment Agency in 2007. The oyster beds are particularly at risk as they will be covered by the River and eventually be completely silted up. At the moment they are exposed and eroded by the tide so perhaps this is the best way to preserve them. They have now been accurately mapped and photographed so there is some preservation through record of these features.

Management		Policy	Policy			
Area		unit	Plan			Comment
			2025	2055	2105	
						Maintain and adjust
MA13	North Links	13.1	MR	MR	MR	bank with groynes
						Re-shape frontage
MA13	Golf Links	13.2	MR	MR	MR	to retain sediment
	Alnmouth					To maintain estuary
MA13	Corner	13.3	HTL	HTL	HTL	shape
	Estuary Outer					Maintain flood
MA13	North	13.4	HTL	HTL	HTL	defence
	Bridge					
MA13	frontage	13.5	HTL	HTL	HTL	
MA13	Estuary Inner	13.6	MR	MR	MR	Local flood defence
	Estuary Outer					
MA13	South	13.7	NAI	NAI	NAI	
						To maintain estuary
MA13	Church Hill	13.8	HTL	HTL	HTL	shape
MA13	Buston Links	13.9	NAI	NAI	NAI	

Table 5.10 Shoreline Management Plan 2 polices for Alnmouth area

The SMP2 preferred policy at the key point where the oyster beds are located is Hold The Line, although other than maintenance of the flood defence there are no further specifics on how this is to be achieved. Also this presumably refers to the retaining wall set back 100m to the landward side of the oyster ponds so any flood defences do not protect them. Again where slaggy material has been seen eroding out of the shore the policy is to Hold The Line although the SMP refers to the bridge frontage and not to the area where archaeological remains are eroding. It is difficult to see how further mitigation could be undertaken without directly disturbing archaeological remains so perhaps a more detailed programme of recording is necessary to monitor these as they erode and record any new features that become visible.




5.11 Amble, Northumberland.

5.11.1 Background

The Amble hulks lie in the zone between LAT and MHWS and are vulnerable to every tide. They lie within SMP2 Policy Unit 15.2 for which The 'Preferred Strategic Option' is 'Managed Retreat'. The hulks lie in the inter-tidal zone within an area of special conservation and are clearly visible at low tide. The hulks are clearly under threat from natural processes and could be further threatened by construction works designed to protect the coastline. The information gathered as a result of this survey provides the necessary level of detail for decisions to be made such as designation, and will provide a baseline against which future damage from erosion can be measured.



Fig 5.11.3 One of the larger hulks visible at Amble

5.11.2 Location and geology

Amble is located at the northernmost tip of the south-east Northumberland Coastal Plain (NGR NU 264049) (PU15.2). This area is characterised by sedimentary limestone, shale and Carboniferous sandstones including local deposits of coal. This has resulted in a gently undulating plain with the occasional sandstone outcrops along the sandy coast and modest sandstone cliffs threaded with beds of coal and shale, where the rivers and streams have cut through them. The Amble hulks are located on the north bank of the River Coquet in the area of the former harbour of Warkworth. This has since silted up and much of the former harbour is now salt marsh, with the Amble marina now being the principal harbour serving both Amble and Warkworth. The hulks are thought to be contemporary with or date to just after the abandonment of the old Warkworth harbour and may relate to the former coal industry, being used as inshore shipping vessels for the transport of coal.



5.11.3 Previous research

Little archaeological research has been undertaken in the Amble area although the hulks have been subject to previous investigation and are recorded in the NMR (907646 – 907649). Although not comprehensive there has been some limited sketched survey by Adrian Osler (2009) and his notes depict the form and position of the hulks, although they do not present a full measured survey. There has also been the suggestion that these may be abandoned herring boats (Parry 2006) although the local view is that they were coal lighters. The NERCZA Phase 2 survey did not undertake a plank-by-plank survey of the hulks but did go further than previous work by recording basic measurements of the hulks and general photography.

5.11.4 NERCZA Phase 2 Archaeological Investigation

The Hulks

The Amble hulks can be split into two broad categories, comprising three smaller boats (357, 358 and 359) and five larger, broader hulks (352, 353, 354, 355 and 356). Also recorded by the survey was a short stretch of what was presumably part of the former Warkworth harbour, comprising a line of protruding wooden posts (453). Looking at the general form of the larger vessels, it would superficially appear that they are all similar to a type of vessel called a wherry, a ship commonly used for transporting coal and other goods up and down estuaries and along the coast. Similar vessels have been seen along the Tyne at Newburn, as recorded by Alan Williams and Paddy Taylor (Taylor 2009). It would seem likely that similar vessels could be seen in the Amble area relating to the nearby colliery and that, upon the decline of Amble as a coal port, they were simply abandoned on the north shore of the River Coquet.

The larger vessels are all between 18 - 19m in length and between 7 - 8m in width at the broadest point, although they are also mostly fragmentary so these may not be the exact original dimensions. The only exception to this is vessel 352, which is 21m in length and 10m wide, and would seem to be of slightly different construction. The four largest hulks (353, 354, 355 and 356) are all clinker-built vessels however the largest and easternmost (352) is carvel built. This slightly different construction method could suggest that this vessel is slightly earlier than the others and not a typical wherry.



Fig 5.11.4 The rudder of vessel 352



Fig 5.11.5 The hold of the largest vessel 352

It is possible that all the larger vessels where abandoned upon the decline of the coal port at Amble / Warkworth. However the presence a possible earlier vessel

could indicate that the hulks where abandoned from different owners at different periods and that they all had individual lineage. Therefore they will all date to slightly different times, demonstrating abandonment over a longer period. Despite this it would appear that the hulks where all more or less abandoned during the same period due to the consistent rate of decay and the spatial positioning of the two vessels recorded as 354 and the position of 355 butting up against the former harbour wall 453.



Fig 5.11.6 Examining the details of carvel built hulk 352

The smaller vessels (357, 358 and 359) seem to be have been more recently abandoned, although again no precise date can be confirmed. They are all small fishing vessels, averaging 10m long and 2m wide. Their construction is small scale clinker as they are not for longer journeys like the larger hulks and they are presumably later than the hulks abandoned on what was formerly the harbour edge but is now inter-tidal mud. They are generally located closer to the shore and are more likely to relate to the fishing industry unlike the wherries and carvel. A 115m long line of upright wooden posts, right on the edge of the lowest extent of the inter-tidal zone, is also visible. Hulk 355 butts against this and a probable interpretation is that this is the edge of the old Warkworth harbour wall that went out of use in the 19th century.



Fig 5.11.7 Vessel 355 lying against the former harbour wall 453



Fig 5.11.8 One of the smaller, and presumably later, fishing vessels

The hulks at Amble represent an easily accessible assemblage of roughly contemporary vessels, presumed abandoned as a result of the decline of Warkworth and Amble as a coal port. Although the survey has recorded basic measurements and photographed the site a detailed survey, not possible within the scope of the NERCZA Phase 2 would increase the understanding of the origins of these vessels. Alan Williams and Paddy Taylor have undertaken a similar survey of the inter-tidal hulks at Newburn on the River Tyne (Taylor 2009). However the Amble hulks represent a better preserved example at a higher level of risk than the Newburn hulks.

5.11.12 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.1-0.3m per year with variation over 100 years leading to a loss of between 30 and 40m (SMP2 for north East England). The Amble hulks face a significant threat from both the direct wave action of every tide as they are almost completely submerged twice day. Also the hulks are located well within the Environment Agency flood zone, outlined in 2007. They face a direct threat from ongoing erosion and rising sea levels which will eventually destroy them completely.

5.11.13 Summary and conclusions

The SMP2 preferred policy in the area where the Amble hulks lie is Managed Realignment, or Managed Retreat. The area behind the hulks between Castles Dikes and the caravan park is currently wetland and a special conservation area. This will be the buffer for further flooding and rising sea levels, effectively creating a lagoon behind the dunes. This policy assessment does not seem to take the hulks into account as being of historical significance. They are not referenced in SMP2 which only considers Scheduled Ancient Monuments as significant.

Management Area		Policy unit	Policy Plan			Comment
			2025	2055	2105	
	Inner					Maintain and enhance
MA15	Estuary	15.2	MR	MR	MR	habitat

Table 5.11 The Shoreline Management Plan 2 policies for the area the hulks lie in.



Fig 5.11.9 The hulks as they appear at low tide.

As previously stated the potential significance of the Amble hulks as a collection of contemporary vessels means that their loss to rising sea levels should be prevented. The NERCZA Phase 2 survey, whilst undertaking photography, basic measurements and providing a preliminary identification, has not answered fundamental questions regarding the vessels age and function. Additional and more detailed survey to produce accurate measured drawings of all key features should be undertaken in order to gain a positive identification of the vessels form and function. The differences between the hulks, with three types of vessels identified, should also be further explored to ascertain the function of the different vessels, their relationship to the harbour at Amble and the dating of their abandonment.



5.12 Scremerston, Northumberland.

5.12.1 Background

At the lowland fort of Scremerston the eastern side of the site has been lost to erosion of the cliff by wave action and the remaining, surviving crop marks are bisected by the East Coast Mainline. The NERCZA Phase 1 desk based assessment has identified both this fort and the fort at Fenham (see section 5.13) to be of 'high regional significance'. These features were again targeted for survey with a view to:

- Photograph, measure and record any visible earthworks.
- Ascertain the nature of any erosion.
- If the erosion has revealed a section of the earthworks record a section of the feature and collect any eroding artefacts that will assist in characterising and dating the features.
- Ascertain whether the sites are suffering active erosion.
- Evaluate both the present level of threat to those sites and the nature of those threats.

The threat to the archaeology at Scremerston is the same as with the palisaded enclosure at Fenham, namely ploughing and further coastal erosion (see section 5.13). However the presence of the main line railway at this location makes it more likely that there will be further sea defence work in order to preserve the railway line. The railway line will also have caused considerable damage to archaeological deposits of the enclosure.



Fig 5.12.3 The field containing the crop marks (right side of photo) of the Scremerston enclosure, demonstration proximity to railway cutting and coast (left side).



5.12.2 Location and geology

The fort at Scremerston is located at NGR NU 0183 4968 (PU 3.1) on a lowlying, rolling plain running down to cliffs in the east. The site is approximately 1.8km south of Spittal, south of Berwick upon Tweed. The land surrounding the site is heavily-ploughed arable fields. The geology is glacial till overlying limestone of the Alston formation, with sandstone, siltstone and mudstone deposits. The East Coast mainline runs very close to the cliff top along this stretch of coast, passing within 10m in some places. The crop marks themselves are located in arable fields on either side of the East Coast mainline. Almost the entire eastern side of the fort has been lost to erosion with only a small fragment surviving in the fields to the east of the railway.

5.12.3 Previous research

The Scremerston multivallate fort was known prior to the NERCZA phase 2 investigations; however no detailed study has been made of it. In 1922 a cist burial containing flints, a female skull and beaker fragments was found at Scremerston (Tolan-Smith 2008) and in 1948 a second cist burial was found containing two Beakers (Craw 1919-1922, 383-384). Other than these small-scale investigations no recorded archaeological work has been undertaken in the area and nothing at all in the vicinity of the multivallate fort.

5.12.4 NERCZA Phase 2 Archaeological Investigation

5.12.5 Prehistoric

The investigation of the fort did not reveal any surface evidence of archaeological remains. Unlike the field at Fenham (see section 5.13) there are no remnant earthworks or other features visible. There are also no features visible in the cliff face and, although there must be some remains surviving given the results of the aerial photographic survey, nothing is visible in section due to slumping of the cliff top. No archaeological evidence for the crop marks recorded by aerial photographic transcription could be seen on the ground and the survey produced no other evidence for prehistoric activity in the area.



Fig 5.12.4 The field containing the crop marks. No surface evidence of the enclosure was visible.

5.12.6 Romano-British onwards

No Romano-British, Early medieval, or medieval features were identified during the course of the survey

5.12.7 Post Medieval

The survey did record several post-medieval features in the landscape around Scremerston. The proximity of the railway to the edge of the cliffs mean that a number of features associated with it can be seen. Most obvious of these is the former route of the railway along the edge of the cliff (469) which was clearly visible surviving as an earthwork. This is cut by the modern route of the railway but can be traced west, along field boundaries and a footpath.



Fig 5.12.5 The former railway cutting, showing the modern route of the line behind the stone wall.

There are further remains approximately 1km to the south where extensive earthwork and structural remains of a limestone quarry (465) and an associated kiln (458) are located. The kiln is eroding. A second kiln is overlain and altered by a Second World War gun emplacement (465). This structure is in a dilapidated state and at extreme risk of collapse in the near future. It is exposed to wave action at high tide and is being undermined, adding to the destabilisation.



Fig 5.12.6 Kiln 458 on the cliff edge, is being eroded.

5.12.8 20th Century

The archaeology of the Second World War is also visible at Scremerston. Here a previously unidentified radar station with structures (466) identical to those of the well-known Craster radar station were identified. Phase 1 of the NERCZA project had not highlighted these buildings, although it did record the position of a radio antenna just to the north. The buildings have been previously recorded as military but their exact function had not been identified.



Fig 5.12.7 The Transmitter/Receiver block at the Scremerston radar station.

The gun emplacement on top of the lime works south of Scremerston (465) was also recorded. Although well-known this structure had never been examined in detail before. Upon closer inspection it was found to be a potentially important structure and is one of only two 6 inch gun emplacements built on this scale in the country, the other being at Budle Bay to the south. The internal fittings and scale of the build is impressive and it may even be based on German designs, although there is no direct evidence for this at present.

Another possibly defensive structure made up of concrete sandbags was also recorded on the foreshore (459). This was eroding out of the cliff face, and its precise form and function was not apparent due to the level of damage to the structure.



Fig 5.12.9 eroding concrete structure built utilising concrete sandbags, possibly of Second World War date.

5.12.9 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.1-0.2m per year (SMP2 for north East England). The threat of erosion at Scremerston is significant and the remains recorded by the survey are under serious threat from erosion at the base of the cliffs. This causes slumping and rock fall and will eventually lead to the loss of significant archaeological material associated with the multivallate fort. All the cliff top remains, including that of the enclosure are at risk from eventual collapse and the retreat of the cliff.

5.12.10 Summary and conclusions

As the East Coast mainline runs in close proximity to the cliff edge there is significant reason to maintain the current cliff alignment. The financial implications of redirecting this line weighed against investment in coastal defence mean that this stretch of coastline is more likely to be eventually protected.

Management						
Area		Policy unit	Policy Plan			Comment
			2025	2055	2105	
MA3	Scremmerston Cliffs	3.1	NAI	NAI	NAI	

Table 5.12 The Shoreline Management Plan 2 policy for Scremerston.

Despite the high level of threat to the known and potential archaeological remains and the current and planned SMP2 policy of No Active Intervention it is unlikely that future management policy will move to further protect this area if needed. This said, the lack of knowledge about the state of what survives beneath the ground at the Scremerston enclosure; means that further investigation, such as geophysics or limited excavation would be preferable and add to our understanding of these lowland multivallate sites. It would also preserve what is left through record prior from any further change along the coast.



Fig 5.12.10 The foreshore and cliffs to the south of Scremerston





5.13 Fenham, Northumberland.

5.13.1 Background

The paisaded enclosure identified from crop marks at Fenham, Northumberland has already been partly destroyed by ploughing and erosion. The surviving remains are very close to a low cliff, about five metres high, immediately above MHWS. This site is likely to be further damaged as the cliff continues to erode. Approximately 50% of the enclosure has already been lost and there is evidence of recent and ongoing landslips on the cliff below. Although the cliff base is protected by salt marsh and wetlands, the cliff is vulnerable to tidal surges and storm events and faces a long-term threat from increased erosion occurring due to a rise in sea level.

The remains of a former medieval grange of Lindisfarne priory, later a manor house complex, also survive in the hamlet of Fenham and there are several other features surrounding this which the survey recorded. There is also evidence of surrounding military features dating to the Second World War, particularly extensive anti-tank defences to the north of Fenham, between the hamlet and the Holy Island causeway.



Fig 5.13.2 One of the cliff face slumps below the Fenham enclosure



5.13.2 Location and geology

Fenham is a small hamlet located on the mainland opposite Holy Island approximately 2.1km south of the Holy Island causeway (NGR NU 08664 40812) (PU 4.3). The hamlet is built around the remains of a former medieval grange of Lindisfarne priory, which later became a manor house. The crop mark remains of a multivallate enclosure have been noted a further 0.7km to the south on the cliff edge. The geology at Fenham is glacial till overlying Yoredale Group limestone with subordinate sandstone and argillaceous rocks (BGS). The rolling hills descend to the coast where they form low till cliffs with limestone outcropping at the base. The coastal environment is mostly wetland and salt marsh, formed by the construction of the Holy Island causeway in the 1950s.

5.13.3 Previous research

The enclosure at Fenham was recorded for the first time as part of Phase 1 of the NERCZA project. It had been previously recognised on aerial photography but no HER or NMR record existed for it. The focus of previous research has been into the earthworks of the grange at Fenham, protected as a Scheduled Ancient Monument (number 6502), although this has not been extensive either. There has been a basic level of investigation as part of the English Heritage scheduling process. Other interesting features in the surrounding landscape have not been investigated in detail.



Fig 5.13.3 The field containing the crop marks of a multivallate enclosure at Fenham

5.13.4 NERCZA Phase 2 Archaeological Investigation

5.13.5 Prehistoric

The focus of the NERCZA Phase 2 investigation was the enclosure identified on aerial photography as crop marks by Phase 1 of the project. The field survey aimed to identify the nature and extent of any surviving remains on the ground or exposed in the cliff face. Upon examining the field containing the crop marks

it was clear that it had been extensively ploughed, probably from the medieval period onwards, considering the proximity to a known grange and the lack of extant ridge and furrow.

Despite the long period of ploughing a closer examination of the exact position of the crop marks revealed possible earthworks and the enclosure is located on the natural high point along the cliff edge (NGR NU 09154 40128). Standing on this point looking inland it is possible to see a slight but regular break of slope set into the natural fall of the land. This regular break of slope (472) was recorded and found to follow almost precisely the line of the inner ditch recorded on aerial photography (Fig 5.13.14). This survival of a slight earthwork (Figs 5.13.5 and 5.13.7) could indicate that below ground survival or archaeological deposits could be very good despite the heavy ploughing. This can be seen elsewhere, such as the promontory fort of Boltby Scar, where the eastern edge of the rampart was completely destroyed but a similar break of slope could be traced and was found to relate exactly to the line of the former ditch (Oswald and Burn 2009, Powesland 2009). The survival of below ground archaeology here was also found to be very good and a complete ditch section survived (Powesland 2010).



Fig 5.13.4 The view from the centre of the multivallate enclosure looking south



Fig 5.13.5 View north from the prehistoric enclosure at Fenham. Ranging pole marks the centre of the surviving remains.

The cliff face is destabilising below the enclosure and has slumped considerably (fig 5.13.7). However this has hidden the upper layers of the exposed cliff face so no archaeological deposits can currently be seen in section. However, with each slump and the associated ongoing erosion, archaeological deposits associated with the fort must be being lost.



Fig 5.13.6 The cliff face below the enclosure. The ranging pole is roughly parallel to the centre of the enclosure.

5.13.6 Romano-British onwards

No Romano-British or Earl medieval features were identified during the course of the survey

5.13.7 Medieval

The remains of a medieval manor, dating to the 14th century, and an earlier, 13th century monastic grange where also observed. However, there was not the time or resources to undertake a detailed survey of the earthworks as part of this project. The earthworks appear to be in excellent condition and would be suitable for a Level 3 earthwork survey (Ainsworth *et al* 2007). They have been accurately mapped from the air but other features, not visible on aerial photography, will be present and further investigation would be beneficial. However, this is not an immediate concern as the remains are not under immediate threat from erosion or rising sea levels, as they are located 50m inland from current MHWS. The NERCZA Phase 2 survey recorded the extent of earthworks still visible (470).



Fig 5.13.7 The earthwork remains of the manor and grange at Fenham looking south from the hamlet.

5.13.8 Post-Medieval

The post-medieval archaeology at Fenham, other than that associated with the development of the hamlet itself, is negligible. Two shooting butts (474 and 471) are visible on the foreshore and these have clearly been rebuilt recently, possibly as a result of collapse encouraged by tidal surges. The main area of post-medieval archaeological remains is a small lime kiln and associated quarry, preserved as earthworks (475), to the north of the settlement. Although much of the original area of the quarry has been ploughed away the northern most extent survives and has been mapped by the survey.



Fig 5.13.8 One of the two shooting butts on the foreshore.



Fig 5.13.9 The remains of a small quarry and associated kiln preserved as earthworks north of Fenham

5.13.9 20th Century

As with elsewhere along the coast the most regularly encountered surviving remains date to the Second World War. Most surprising of these can be seen along the cliff edge in the vicinity of the Fenham enclosure. Here barbed wire and picket wire can be seen all along the upper cliff edge for a length of approximately 100m (473).



Fig 5.13.10 Picket wire and barbed wire, seen on the left of the picture not the fence in the foreground, exposed on the cliff edge at Fenham.

Also visible to the north of Fenham is a long, double line of anti-tank blocks that run all the way to the Holy Island causeway. These blocks are generally wellpreserved and present an interesting, possibly wartime, modification. Between the gaps in alternate blocks (every second gap) small brick walls have been constructed. These could have two different explanations. They could be wartime modifications to give firing positions for troops between the cubes. This is possible as they appear to be contemporary with the blocks. However they could also be post-war additions to create shooting butts overlooking the wetlands as they would not afford a soldier much cover.



Fig 5.13.11 Anti-tank blocks at Fenham with a small brick built shooting butt or defensive wall clearly visible.

The other clearly visible defensive structure is the anti-glider poles visible on Fenham Flats and Goswick Sands. These rows of posts (496) were designed to prevent gliders landing on the wide, open sands, which were conducive to this form of attack in wartime. This area was mostly open tidal sands as the causeway had not been constructed at the time and the current area of salt marsh was not as extensive.



Fig 5.13.12 Row of anti-tank blocks north of Fenham



Fig 5.13.13 The anti glider poles on Goswick Sands and Fenham Flats

5.13.10 Threat from erosion

The Shoreline Management Plan estimates the cliff retreat along this part of the coastline at 0.2-0.4 m per year (SMP2 for north East England). The threat from erosion to these sites is ongoing, especially in the cliffs at Fenham were slumps and collapse are clearly visible. Here, although the cliff is protected by the wetlands, rising sea levels will increase the impact of tidal surges. SMP2 projected coastline shows that a further 15% of the Fenham enclosure will be lost to erosion within 100 years. This will leave relatively little scope for further investigation of the monument.

5.13.11 Summary and conclusions

The situation at Fenham may improve, although the policy of the SMP2 is No Active Intervention, with the encouragement of natural defences to take place. This encouragement of the wetland environment will help to protect the site in the intermediate term; however long-term threat from rising sea levels as well as the immediate threat of slumping will continue to erode the site.

			Polic			
Managemen		Policy	у			
t Area		unit	Plan			Comment
			2025	2055	2105	
						Encourage the development of
	Fenham					inter-tidal natural defence to rising
MA4	flats	4.3	NAI	NAI	NAI	hinterland

Table 5.13 Shoreline Management Plan 2 Policy for the area of Fenham.

Although as much as 50% of the fort has been lost to coastal erosion what survives is potentially in a very well-preserved condition. There is an immediate

threat to the remains from the ongoing effects of slumping of the cliff face, which will be directly affecting the archaeological deposits in the ditches and internal features of the fort. However it could be argued that as 50% of this monument has already been lost to erosion it may be seen as already too damaged to provide further archaeological information. There are much better preserved examples of this type of enclosure, one of which was highlighted by Phase 1 of this project for the first time at NU 105 374 in Northumberland of which crop mark remains of a hut circle set within three ditches and a palisade slot are visible (Tolan Smith 2008). This example could be a much better target for further investigation as it is in a better state of preservation set back 1km inland and is not currently threatened by coastal erosion.



5.14 Nessend, Holy Island, Northumberland.

5.14.1 Background

The Mesolithic lithic scatter at Nessend Quarry on Holy Island is significant as finished tools only account for 1.7% of the assemblage (O'Sullivan and Young 1995). The remainder represents debitage and manufacturing waste. This analysis suggests that the site was primarily used to locate and collect material to produce tools that were then used elsewhere. The assemblage is made up of nearly 2000 items and there is also the presence of bevelled pebbles, usually associated with the processing of seal skins, so other activities may have been undertaken at the site.

The NERCZA survey aimed to:

- Record the visible extent of the lithic scatter
- Identify any visible worked flints
- Quantify any erosion taking place
- Assess the level of threat the site faces

The survey also widened the survey to include the chapel on St Cuthbert's Isle, due to the high level of threat from erosion faced by the surviving remains. Other features already visible on the rest of the island were also recorded in the course of the walkover survey. This allowed a wider assessment of threatened sites allowing the ongoing erosion at Nessend to be placed into a wider context.



Fig 5.14.3 The earthwork and structural remains of the chapel on St Cuthbert's Isle



5.14.2 Location and geology

Holy Island is a tidal island, 1.7km offshore from the mainland, and is connected via an artificial causeway constructed in the 1950s. The quarry at Nessend is located on the north side of Holy Island (NU 12935 43669) (PU4.7-5.2). The site of the Mesolithic flint scatter at the quarry, recorded in 1987, is being eroded on the surface by storm event erosion and run off into the quarry. The geology is glacial till and wind-blown sand overlying limestone of the Alston formation, with sandstone, siltstone and mudstone deposits.

5.14.3 Previous research

The Ness End lithic scatter was the target for archaeological field walking and accurate survey of its extent in 1987. This work catalogued 2000 flints, predominantly debitage, although there were some flint tools and some bevelled pebbles possibly intended for the working seal skin (O'Sullivan and Young 1995). The site has not been further investigated since the initial study. Other sites on Holy Island have been the focus of detailed archaeological investigation, including Lindisfarne Priory, Lindisfarne Castle, the early medieval settlement of Green Shiels and the town of Lindisfarne itself.

5.14.4 NERCZA Phase 2 Archaeological Investigation

5.14.5 Prehistoric

Upon investigation the extent of the flint scatter was found much as described in the Book of Lindisfarne: Holy Island (O'Sullivan and Young 1995). Several pieces of flint debitage were seen across the area of the original survey, which comprised an area of exposed clay just south of Nessend Quarry. The area of exposed clay is being eroded by wind blown sand scouring the exposed clay surface and collapse of the land surface into the quarry. In the long term the area will also be affected by coastal retreat. The survey identified an area of exposed clay (518) 155m x 139m in area with at least 2 worked flints (518;1 and 518;2) recorded and retrieved and several more possible flints observed. The area would benefit from continued monitoring and regular field walking of the freshly exposed areas. This could be implemented by involving motivated local archaeology groups under the supervision of a professional archaeologist to regularly visit the site and make notes on its condition. The value of a local motivated individual or group can be seen at Low Hauxley where Jim Nesbitt has identified many important archaeological features as a result of regular monitoiring.



Fig 5.14.4 Part of the exposed area of the lithic scatter

5.14.6 Romano-British

No Romano-British features were identified during the course of the survey

5.14.7 Early Medieval

There is a well-known, previously recorded example of an early medieval settlement located at Green Shiels (408) (NU 12154 43642) (SAM no 7787). The site is made up of a series of four longhouses set within an irregular-shaped enclosure. This site has been extensively excavated and surveyed and was recorded rapidly as part of the NERCZA Phase 2 survey in order to gain a current condition statement. The site is well-protected in a stable dune system, and signage interprets the site for visitors. The structures themselves are well-preserved with internal features like hearths and doorways surviving.


Fig 5.14.5 One of the early medieval long houses at the Green Shiels settlement.



Fig 5.14.6 A stone hearth surviving in the centre of one of the long houses at Green Shiels.

5.14.8 Medieval

St Cuthbert's Isle is a small tidal island located 194m offshore and to the south of Lindisfarne. The island contains the earthwork and structural remains of a medieval hermitage and chapel (SAM no 7797), which survived well enough to enable an accurate plan of the structures to be recorded. The chapel (386) comprises two rooms, forming a T-shaped layout with the larger room to the west. The other structure (388), possibly accommodation or storage, is on the

east end of the island and forms a rectangular structure with two (or possibly three) rooms. These structures are directly threatened by coastal erosion and this is especially visible at the west end of the chapel were the eastern wall has been partially lost to wave action. The site is threatened every high tide in all but the calmest weather. Archaeological deposits are clearly visible in the section as a result of the erosion and these will continue to disappear as the erosion continues.

Also visible next to the remains of the chapel are several earthwork features, the most distinct of which is a stone lined drain (387), which may be cut into rock on the east side of the island. A possible spoil heap was also recorded, which probably relates to the excavations of the remains in the 19th and 20th centuries. This sub-circular mound (391) is located just to the north of the chapel and could represent the remains of a monastic cell, although this seems unlikely given the form of the earthwork. Also recorded where the remains of a slipway (392) and several later mooring fittings (389 and 390).



Fig 5.14.7 The eroding section of the chapel (388) on St Cuthbert's Isle



Fig 5.14.8 The main standing remains of the chapel, with Lindisfarne Priory visible in the background.



Fig 5.14.9 The earthwork remains of a drain for the chapel and associated building.

The remains of another possible medieval long house were recorded, underlying the railway embankment connecting Castle Point with Ness End Quarry. This feature (405) is a rectangular earthwork protruding from beneath the embankment and may be the remains of a previously un-recognised longhouse. Detailed survey and or geophysics could provide more certainty to this interpretation.



Fig 5.14.10 Possible earthwork remains of a medieval or later longhouse.

5.14.9 Post-Medieval

There is extensive evidence of pos- medieval industrial activity on Holy Island, such as the famous Lime Kilns (403), which are situated below Lindisfarne Castle. These were recorded as they are under threat from wave action on the highest tides, being located on the foreshore. These are the most visible industrial features on the island, although there are also the remains of a holloway (399), an industrial railway (404), at least four quarries (406, 407, 519 and 410) and an extremely fragile and fragmentary pier to the south-west of Lindisfarne Castle (398). This pier is in very poor condition and exposed to erosion by wave action at every high tide.



Fig 5.14.11 The impressive Lime Kilns (403) below Lindisfarne Castle



Fig 5.14.12 The remains of a wooden and stone pier (398) south of Lindisfarne castle



Fig 5.14.13 The holloway (399) leading to pier (398), the posts seen in fig 5.14.12 can just be seen over the fence.

Substantial quarries are located at Ness End (519), Emmanuel Point (407), and at Castle Point. However, extensive remains can also be seen in the centre of the island where two 19th century lime kilns are located inside a large quarry (410). Here a line of earlier kilns, surviving as earthworks (411), can also be seen.



Fig 5.14.14 Two lime kilns preserved within the quarry (410)

There are also several defensive features dating to the post-medieval period surviving on Holy Island. The most well known of these features are Lindisfarne Castle and the Tudor period fort overlooking the harbour. The castle was not recorded as it has already been subject to topographic survey, undertaken on behalf of the National Trust. However the fort (396) is positioned on an eroding promontory and is in a poor condition and so was surveyed and photographed as part of the NERCZA Phase 2 survey.

Also visible in proximity to Lindisfarne castle was a small raised fire step and ramp (401). This could be a later industrial feature, related to the loading of material fro the quarry or lime kilns, however its position makes this function less likely. This feature is more likely to be part of the original fort at Lindisfarne Castle, as it is connected with a bank that encloses the rock outcrop.



Fig 5.14.15 The stone built fire step, or loading bay and raised bank for the lime kilns.



Fig 5.14.16 The keep of the Tudor fort (396), overlooking the harbour

A further longhouse (409) was also recorded on the north side of the island, 200m west of the Green Shiels settlement. However this longhouse was occupied until the early 20th century, and was rebuilt in the post-medieval period. Despite this it is possible that this structure has medieval or even early medieval origins. This structure is gradually being buried by windblown sand.



Fig 5.14.17 Partially upstanding gable end of Longhouse (409)

5.14.11 20th Century

The archaeology of the Second World War is notable by its absence on Holy Island. The defensive features built along the shore on the landward side of the island seem to suggest the island was not considered defensible or practical to reenforce and the bulk of the defences were built on the mainland.

5.14.12 Threat from erosion

The Shoreline Management Plan estimates the retreat along this part of the coastline at 0.1m per year (SMP2 for north East England). There are two key areas that face serious threat from erosion on Holy Island. The most obvious of these is the ongoing damage to the remains on St Cuthbert's Isle. The ongoing erosion here is seriously impacting upon known significant remains at every high tide and during every storm event. The effects on the lithic scatter at Ness End are harder to quantify but no less serious. Here the effect of wind blown sand is scouring the surface of the clay and eroding the land surface containing the flint. The area is not presently at threat from direct coastal erosion but there is evidence of slumping of the land surface into the 19th century quarry at Nessend.

5.14.13 Summary and conclusions

Holy Island is a site of archaeological, historical and environmental interest, with seven Scheduled Ancient Monuments on the island (including St Cuthbert's Chapel). The north shore of Holy Island is designated by SMP2 as No Active Intervention. However, there is a plan to maintain the natural dune system which protects this stretch of coast meaning it is not immediately under threat by direct coastal erosion. The site is however facing a threat from windblown sand and effects of run-off erosion.

Management						
Area		Policy unit	Policy Plan			Comment
			2025	2055	2105	
MA5	North Coast	5.1	NAI	NAI	NAI	Maintain natural dunes
MA5	East Coast	5.2	NAI	NAI	NAI	
MA4	Holy Island Cliff	4.7	NAI	NAI	NAI	
						Maintain back defence to
MA4	Holy Island Harbour	4.8	HTL	HTL	HTL	harbour

Table 5.14 Shoreline Management Plan policy for the area of Holy Island

On the south coast of Holy Island, and for St Cuthbert's Isle the SMP2 policy is No Active Intervention. This means that there will be no policy implemented to protect the remains at St Cuthbert's Isle. As a result, further archaeological investigation is crucial to preserve this monument through record. The only area where Hold The Line will be implemented is the harbour area, where the Tudor fort is located.



5.15 Bamburgh, Northumberland

5.15.1 Background

The most notable visible landmark on this stretch of coastline is that of Bamburgh Castle itself. However, there are well-known and partially recorded remains of an early medieval cemetery to the south of the castle, at a location called the Bowl Hole. Here early medieval and prehistoric remains have been observed, revealed in a large dune blowouts. The Bamburgh Research Project have excavated and investigated the area of the burial ground for 9 years. The results of the excavations are now in the process of being analysed and a publication will follow.

The site is under direct threat of erosion from dune blowouts. The continued investigation into the area by the Bamburgh Research Project means that much work has already been undertaken at the Bowl Hole, and for this reason the principal aim of the survey was to quantify the effects of erosion in that area and also to investigate the area around Bamburgh to record any other features that may be visible in the surrounding landscape.



Fig 5.15.3Bamburgh Castle viewed from the village, looking east.

An extensive network of military features, including trenches, pillboxes and antitank defences are located around Bamburgh. As these were observed to be quite extensive it was decided to record them so they could be incorporated into the wider survey of the military archaeology of the Northumberland Coast AONB (see section 5.16).



5.15.2 Location and geology

Bamburgh is located on the north Northumberland coast north of Seahouses. The landscape comprises low rolling hills with occasional basalt rock outcrops, such as the one on which Bamburgh Castle sits. The coastal setting is one of wide sandy beaches with dunes behind; to the north of Bamburgh a rock outcrop forms a shore platform to the south of the beach of Budle Bay. The geology is comprised of till and windblown sand overlying the basalt which occasionally outcrops.



Fig 5.15.4 The beach and dunes with Bamburgh Castle in the background, looking south.

5.15.3 Previous research

Extensive research has been undertaken around Bamburgh by the Bamburgh Research Project (BRP). They have investigated the burials in the dunes, history and archaeology of the castle and the surrounding environs. For the purposes of this rapid examination of the archaeology some of the results are summarised below.

The first excavations within the castle occurred in the 1960s and 1970s undertaken by Dr Hope-Taylor, who discovered the gold plaque known as the Bamburgh Beast, as well as the Bamburgh Sword. The results of these excavations were never published. In 1996 the Bamburgh Research Project was formed, with the aim of investigating the site using modern archaeological techniques to continue to research the archaeology of the castle and town (Young 2006).

The Bowl Hole burial ground was initially lost in the later medieval period but was revealed by a violent storm in 1817 (Young 2006). It subsequently became the subject of various antiquarian investigations. More recently the site has been excavated by the Bamburgh Research Project between 1998 and 2007. In June 2006 a three year project began between the Bamburgh Research Project and

Durham University to fully analyse the bone from the Bowl Hole excavations, the results of which are due to be published shortly.

5.15.4 NERCZA Phase 2 Archaeological Investigation

5.15.5 Prehistoric and Romano British periods

No Prehistoric or Romano British features were identified by the phase II survey

5.15.6 Early Medieval

The Bowl Hole burial site has been investigated in detail over the past 9 years by the Bamburgh Research Project. Although the area excavated is overlain by windblown sand the extent of the burial ground seen so far comprises a generally flat topography. However the NERCZA Phase 2 work has surveyed several low mounds which, although made up wind-blown sand, do not fit into the rest of the pattern of the dune build up surrounding them. It is possible that these low mounds are dunes "artificially" created by wind-blown sand collecting around and overlying pre-existing mounds. These could therefore represent burial mounds relating to the early medieval cemetery.

The site is in an area considered as being geomorphologically stable, however there is still the possibility of serious dune blow out, which could destabilise a large area of dunes and expose more of the burial ground to the elements. The area is monitored as part of the Bamburgh Research Project and so if any serious threat arises and places archaeological remains in danger of erosion they will be able to respond.



Fig 5.15.5 One of the possible burial mounds at the Bowl Hole.

5.15.7 Medieval

The castle and other medieval features were photographed but not accurately plotted with the GPS as part of the rapid survey as they are already well-known

and understood monuments, not at imminent risk of coastal erosion. The sites are also covered in some detail as part of the NERCZA Phase 1 survey report.

5.15.8 Post-Medieval

No industrial or obvious post-medieval features were observed during the course of the NERCZA Phase 2 survey.

5.15.9 20th Century

A concentration of Second World War military features survive in Bamburgh. These survive to the south of the castle and can be seen along the beachfront as well as being set back in the dune system. All are exposed to various forms of weathering and the effects of coastal erosion. These are discussed as part of the Northumberland Coast AONB military archaeology survey discussed in detail in section 5.16. However the remains will be discussed in brief here.



Fig 5.15.6 A hexagonal pillbox in the dunes at Bamburgh.

The concentration of upstanding military remains at Bamburgh includes three well-preserved pillboxes (370, 372, and 375). These pillboxes, combined with anti-tank defences and trenches, form a defensive pocket around a low-lying beach with a small knoll behind on which the trenches are positioned (see below)



Fig 5.15.7 The front aperture of one of the pillboxes.



Fig 5.15.8 The rear entrance of the same pillbox.

Pillbox 370 is actually a 6 inch gun emplacement, set in the same dune system, and this can be distinguished because of its large aperture and the fitting for a much larger gun mount. Pillboxes did not have these features and were of a much simpler construction. This is a very well-preserved example of such a gun position and is set within a more stable area of the dunes.



Fig 5.15.9 The front of the 6 inch gun emplacement (370)



Fig 5.15.10 The same emplacement looking head on at the gun aperture.



Fig 5.15.11 The concrete mounting for a 6 inch gun which would have been overlain by a metal mounting for the gun to swivel on.

As can be seen elsewhere on the coast there are earthwork remains of trenches and weapons pits around the defences at Bamburgh. Weapons pits (373) can be clearly identified around the central pillbox (375). These features occupy a low knoll defending the flanks of the pillbox (375) creating a natural defensive redoubt. Three additional trenches (376, 377 and 457) are brick lined, a very unusual construction type. This could simply be due to a surplus of bricks and some skilled labour locally. Alternatively, this could be a post-war training feature, for example re-enforcing the trenches as the Cold War began.

Another surviving element of the Second World War archaeology is anti-tank defences, which can be seen along the beach at Bamburgh. The wide open beaches were obviously conducive to amphibious assault and so the concentration of defences here is not surprising. However, to the south of Bamburgh Castle many have been removed, while on the beaches to the north larger quantities survive. Some have also been painted as dice (fig 5.15.13) and this re-working of the emplacements, while a commonly seen activity in the south of England, is not a common occurrence on the Northumberland coast.



Fig 5.15.12 One of the brick lined trenches (376) found around Bamburgh.



Fig 5.15.13 Anti-tank blocks painted as dice to the north of Bamburgh Castle.

The survival of a larger number of defensive structures to the north of the castle in comparison to the south could be explained by a number of factors. The land to the south is owned by the Bamburgh Estate and so may have been cleared to improve the area for holiday makers. The northern part was perhaps not considered as important, or may have been in different land ownership at some point since the Second World War. Also the defences to the south simply may not have been as extensive as the defences to the north during the Second World War, although given the nature of the wide beach here this seems unlikely.

5.15.10 Threat from erosion

The Shoreline Management Plan estimates the retreat along this part of the coastline at 0.3-0.4 m per year (SMP2 for north East England). The threat faced by these remains from erosion is not as serious as elsewhere on the coast, although ongoing erosion of the dune system is constant. The wide sandy beaches and dune system protect most of the archaeological remains recorded from the most of serious effects of erosion at present. The anti-tank blocks seen on the beach to the north of Bamburgh Castle represent some of the most exposed archaeological remains but seem to be stable as there is little evidence of active erosion, such as has been seen elsewhere along the coast.

Elsewhere the main threat to archaeological remains comes from severe dune blowouts such as the one seen in 1817. There is no evidence to suggest that there is a high likelihood of future blowouts, although if they were to happen, either in the dunes where the military archaeology survives or the vicinity of the Bowl Hole, then archaeology could be exposed and placed at risk of erosion.

5.15.11 Summary and conclusions

The archaeology at Bamburgh is varied and incredibly significant, ranging from an early medieval Northumbrian royal stronghold, through to extensive Second World War military archaeology. It has a long history of research and investigation by various organisations and archaeologists. There is no doubt about the importance of the resource which survives at Bamburgh and it will be the focus of future research, not least by the Bamburgh Research Project. However the SMP2 policy for the area is No Active Intervention, so no plans are in place to prevent further erosion. The only comment in the SMP is to potentially realign the road if it becomes necessary.

Managemen		Policy	Policy			
t Area		unit	Plan			Comment
			2025	2055	2105	
	Bamburgh and St					Potential realignment of
MA6	Aidens Dunes	6.1	NAI	NAI	NAI	road in the long term

Table 5.15 Shoreline Management Plan 2 policy for Bamburgh

For this reason continued monitoring of the long-term effects of erosion on the archaeological sites at Bamburgh should be maintained. This will most likely be done through the continued work of the Bambrugh Research Project which could monitor erosion events and respond effectively to any increased threat to archaeological remains.



Fig 5.15.14 Anti-tank blocks exposed on the beach at Bamburgh.

5.16 NORTHUMBERLAND COAST AREA OF OUTSTANDING NATURAL BEAUTY MILITARY ARCHAEOLOGY SURVEY

5.16.1 Summary

In January 2010 a rapid archaeological survey and field investigation of surviving Second World War defences within the Northumberland Coast Area of Outstanding Natural Beauty was undertaken by Archaeological Research Services Ltd. The aim of the survey was to better understand the nature of the surviving earthwork and structural remains and record any previously unrecorded or unknown military or defensive remains within the AONB. This data could then be used to implement better interpretation of what survives and inform future management. A secondary aim of the project was to suggest possible ways of carrying out this interpretation. The survey utilised rapid walkover methodology with extensive photography, notes and accurate positional information utilising mapping grade Global Positioning System equipment.

The survey provided detailed condition statements for over 100 surviving remains along with photography, descriptive text and an assessment of any threat to the monument. This was collated within a GIS database along with additional data from Phases I and II of the North East Rapid Coastal Zone Assessments for this area of Northumberland. This allowed a comparison with what was known to previously exist with what remains are currently visible. This produced a complete record of Second World War remains within the AONB along with the current condition statements for each individual monument.

The survey revealed beach defences such as pillboxes and anti-tank blocks in situ and moved from their original positions, and these are clearly visible due to their size, frequency and construction. Several civil defence structures such as air raid shelters were also identified. Many of the identified remains were also reused later and Cold War Observer Corps monitoring posts were also recorded due to their Second World War components. The survey also uncovered more subtle and ephemeral features such as weapons pits and fire trenches still surviving as earthworks along the coast. Many of these features were recorded for the first time developing the knowledge of what survives of the defences for this part of the country during the Second World War.

The extent of surviving remains within the AONB has provided key areas that could be considered for further interpretation to the public. These include guided walks, self-guided walk leaflets, signage, Heritage Open Days, community workshops, oral history projects, and possible guide publications. All of this could help to promote and explain the surviving archaeology within the AONB to the public. Key sites such as Dunstanburgh, Craster, Boulmer, Budle and Goswick were identified as being prime targets for interpretation due to the concentrations of surviving military archaeology and the range of monuments that can be seen from the Second World War

By targeting these sites and utilising the detailed survey data from the fieldwork an effective program of interpretation of the Second World War military remains that survive within the AONB could easily be undertaken. This will develop local understanding and also help visitors to the area understand the military archaeology they can see around them.

5.16.2. Introduction

In January 2010 Archaeological Research Services Ltd was commissioned by the Northumberland Coast Area of Outstanding Natural Beauty to undertake a rapid survey of archaeological remains dating from the Second World War that survive within the AONB, the results of which are incorporated into the wider NERCZA Phase 2 report here. The survey methodology was the same as that employed for the larger scale survey (see section 4.2), although it only focused on the military remains. The main objective of the survey was to record and interpret any surviving Second World War remains within the AONB. The results of this would then provide possible areas and sites for interpretation of the remains for those visiting the AONB.



Fig.5.16.1 Pillbox constructed of concrete sandbags south of Dunstan Steads.

The project area encompassed the whole of the AONB meaning that the total area to be surveyed was 138km². The ultimate aim of the project was to provide potential targets for further interpretation to the public so, due to the large size of the project area, those locations with public access already in place were prioritised. Despite this the survey achieved excellent coverage of the AONB and 141 records of surviving military and related features from the Second World War were entered into the project GIS. Additional data from Phase 1 of the North East Rapid Coastal Zone Assessment was also incorporated, adding a further range of Second World War sites that were already known to have existed.

The compilation of these records, along with detailed condition statements and photography, provide a useful land management tool for the AONB. This will allow for a more informed knowledge base for managing these important assets in the future. This data can then also be used to plan public interpretation of the remains, based on location, access, condition and risk to the remains.

5.16.3. Location

The Northumberland Coast AONB extends from the River Coquet, to just south of Berwick at Scremerston. It encompasses the settlements of Bamburgh, Seahouses, Dunstanburgh, Craster, and Alnmouth and is 138km² in area (Fig 5.16.2). There is considerable geological variation within this area, although the bedrock is primarily limestone and sandstone. The AONB also contains a range of different landscape types including salt marsh, dune systems, wetlands and hard rock cliffs as well as arable fields and a range of foreshore types. Land use is mostly arable agriculture with some areas surviving as pasture. There is also evidence of former industrial activity including creation of quarry complexes and open cast coal mining.



5.16.5 Previous Archaeological research

Although much archaeological research has been undertaken within the Northumberland Coast Area of Outstanding Natural Beauty, none of this work has been undertaken to look specifically at the surviving Second World War remains. This project has collated the available evidence from documentary sources and aerial photography, together with extensive field survey, for the first time.

Other work, undertaken at a national level, includes the Defence of Britain Project commissioned by the Council for British Archaeology. This project produced a database of 20,000 military sites nationwide. However it was far from a complete record of what now survives on the ground as both aerial photographic survey and field investigation of the North East coast as part of the NERCZA project has revealed that much more survives than was recorded by the D.O.B project.

Between 2003 and 2006 English Heritage undertook a multi-disciplinary investigation of Dunstanburgh Castle and its environs (Oswald *et al.* 2006). This study involved historical investigations, as well as architectural and earthwork surveys; it also involved study of high-quality aerial photography from 2003. Although the investigation covered all aspects of the archaeological narrative of the area, a particular focus was made on the evidence surviving from the Second World War. Pillboxes, beach defences, weapons pits and a mine field were all recorded in the area surrounding the castle and local people's testimony was also incorporated to build up more social history of the area during the Second World War. The wider area was also investigated to put the Second World War defences into context which recognised the depth and complexity of known defences, but not all existing earthworks and structures were recorded due to the scale of investigation.

English Heritage undertook a detailed earthwork survey of Craster Radar Station between 2003 and 2005, producing a Research Series report in 2006 (Hunt and Ainsworth 2006). This report covered the nature and extent of upstanding remains within the area of the former radar station and also provided a detailed historical background to the site and the development of radar technology. The Second World War military archaeology of the Northumberland coast has been enthusiastically investigated by highly motivated amateur groups and individual over the years and several pieces have been published in recent editions of the Archaeology in Northumberland magazine produced by Northumberland County Council.

5.16.6 Description of Surviving Remains

The survey recorded many types of monument. For ease of interpretation and reference they have been grouped here into nine main groups. Detailed typologies have not been included in favour of a more interpretive approach which also considers management issues and any threats faced. Although the distribution of surviving sites across the AONB is fairly regular there is more survival closer to the coastline. Agriculture, industry and settlement have removed some of the military archaeology further inland where as sites survive along the less well-developed coastal area. Several key locations with concentrations of military archaeology were identified, such as RAF Boulmer,

Dunstanburgh Castle, and the area around the Holy Island causeway including Goswick Sands. (Fig. 5.16.27). The exact nature of what remains at these locations is described below.

5.16.6.1Pillboxes

Pillboxes are the most easily recognizable surviving structures from the Second World War. The combined NERCZA Phase 2and AONB survey recorded 37 pillboxes surviving in some form within the AONB (Fig. 5.16.5), all of different construction and size. Although the Ministry of Defence produced many different standard types of pillbox during the Second World War (Fig. 5.16.4 and 5.16.6) in reality there are many more as they were constructed to suit the requirements and setting of each individual location in an *ad hoc* manner by many different groups around the country. For this reason a type 23 pillbox in Northumberland will differ considerably from a type 23 pillbox in Suffolk. The range and complexity of pillbox types has been covered in detail by other publications (e.g. Osbourne 2008) and so will not be discussed here. The current report will not attempt to distinguish each individual type recorded as this is not of immediate practical use, except to the interested military researcher, but will concentrate on the survival of the remains and management issues which are of more benefit to the AONB.

Pillboxes are located at weak points in the coastal defence. The mainstay of the defences would have been barbed wire entanglements, anti-tank blocks, fire trenches, and weapons pits. The pillboxes are often thought to be the main part of the defence when in fact they were merely reinforcing the defence infrastructure which is now less visible. The pillboxes today represent key indicators of areas where more surviving remains could be found. This is especially true in the area around Dunstanburgh and Embleton where the pillboxes (of various types) are all located in proximity to surviving earthworks of trenches and weapons pits.



Fig 5.16.3 Pillbox located in dunes at Dunstanburgh (scale = 2m)



Fig 5.16.4. Pillbox constructed of concrete sandbags south of Dunstan Steads

Most pillboxes extant within the AONB are very well-preserved, with most surviving along the coast or at road edges. Examples at Dunstanburgh and Craster (633. 635, 636, 637,640, 645, 658, 657) are exceptionally well-preserved although those on the edge of the dune system face the threat of erosion from the sea. Other less well-preserved pillboxes can be seen, such as one in the dunes between Bamburgh and Seahouses (672) (Fig. 5.16.21), which has been deliberately destroyed by explosives in the post-war period. The blast damage is still in evidence and the pillbox is now at risk of collapse, both as a result of the original blast and due to natural slumping.

Management of these structures appears to be straightforward as most are set within agricultural land, form part of field boundaries or are next to tracks. Therefore they do not generally interfere in the current land use and face little threat of demolition or removal. Those along the coast are at more threat from natural causes and several are eroding or slumping as a result of erosion. These historic assets are protected in line with the prevailing Shoreline Management Plan policy unit and little can be done at present to change the level of threat. Despite the level of threat to the pillboxes along the coast, these structures are all along public rights of way or permissive paths. These paths make them easily accessible and these pillboxes therefore lend themselves to exploration by the public.





5.16.6.2 Trenches and Weapons pits

As mentioned above, although pillboxes are perhaps the most commonly visible surviving defensive structure from the Second World War the bulk of the manned defences would have been hand-dug trenches and weapons pits, with pillboxes reinforcing weak points. This defence infrastructure was extensive and the extent of trenches and weapons pits in existence during the war was plotted by the aerial survey element of Phase 1 of the North East Rapid Coastal Zone Assessment. This produced detailed transcription for the Northumberland Coast AONB area and this was utilised during the Phase 2 NERCZA field survey and the AONB survey.



Fig 5.16.7. Weapons pit preserved as an earthwork, directly south of Dunstanburgh castle.

The temporary nature of these defences has led to the common assumption that they have either been filled in or no longer exist. Field survey has revealed that these monuments frequently survive as earthworks at several locations along the Northumberland coast. Within the Northumberland Coast AONB the best survival can be seen around Dunstanburgh Castle. The strip of stable dunes between Dunstanburgh Castle and Embleton contain some of the best-preserved examples. These are located in a narrow strip between the golf course and the beach where the lack of development and agriculture have left these earthworks untouched since the Second World War. Other locations with surviving remains of earthwork defences include Budle, Bamburgh, Benthall, Boulmer and Alnmouth.



Fig 5.16.8. Recording a Second World War Fire Trench, preserved as an earthwork north of Dunstanburgh Castle

Surviving earthwork defences have not been recognised inland from the coast within the AONB as part of this survey (Fig.5.16.9), and this is not surprising. The combined factors of open-cast mining and arable agriculture will have obliterated any trace of these sometime ephemeral features. This makes the examples that survive on the coast potentially more important as they form the main surviving remains from the defensive infrastructure and they are at a greater risk from active erosion and long-term coastal retreat.



5.16.6.3 Anti-tank Defences

Anti-tank blocks are another common sight along the beaches of the North East coast. They represent another common surviving form of Second World War monument due to their size, construction and the scale of deployment (Figs 5.16.10 & 5.16.11). Designed to impede tanks progressing inland from landing beaches they are sometimes known as Dragons Teeth, however this term usually refers to the more triangular blocks, of which none were recorded along the Northumberland coast. The types seen in Northumberland are commonly referred to simply anti-tank cubes or blocks.

Originally the area of anti-tank blocks deployed would have been much more extensive and this can be seen from 1940s aerial photography. The anti-tank blocks that are visible now (Fig. 5.16.10) are only a fragment of the original deployment. This can be attributed to several key reasons. Firstly, many of the anti-tank defences were removed in the immediate post-war period; they cluttered up beaches and access routes and had outlived their usefulness. Secondly, they were moved from their original position to form new anti-erosion defences and this can be seen around Boulmer where the anti-tank blocks have been incorporated into a sea wall defence a distance from their original position. The third reason is natural; many anti-tank blocks have actually encouraged dune growth as they have gradually trapped sand around them. This can be seen in areas around Fenham and Goswick were new dunes have built up around the anti-tank blocks which have helped to stabilise the area surrounding them, accumulating sand and protecting the coast from erosion. This has meant that many lines of anti-tank blocks have disappeared from direct view but the former line of them can be traced in unusually straight and angular dune systems, with the occasional block visible.





Fig 5.16.11. Anti-tank blocks on the beach directly north of the mouth of the River Coquet.

The main threat to anti-tank blocks is from removal by those responsible for managing the land. Due to their large number they are often not considered to be important, however the key role they played in defending the coast should not be underestimated as they are an important part of the story. They are also often highly exposed to coastal erosion on beaches and eventually face being broken up by wave action. This is the case at Boulmer were the few *in situ* anti-tank blocks are at risk from erosion.

5.16.6.4 Gun Emplacements

There are two main types of gun emplacement still existing within the Northumberland Coast AONB. Large coastal battery sites, of which two still exist, and machine gun emplacements which are more numerous. The coastal battery sites within the AONB, which originally housed large calibre guns for anti-naval defence (Fig.5.16.12), are located at Budle (500) and Goswick (463). These two sites are very similar in construction and location, both being situated on a high point and defending a potential landing beach. The site at Budle differs slightly from that at Goswick in that there are also well-preserved subsidiary structural remains of unknown function associated with the gun battery (502).



Fig 5.16.12. Coastal battery gun emplacement covering the potential landing beach at Goswick (463).

The two coastal gun emplacements at Goswick and Budle are clearly impressively built structures. They were built to house 6 pound guns, something evident from their size and the remaining fittings on the floor of each emplacement However 6 pound emplacements are usually simpler structures, similar to those seen at Bamburgh, made with thin concrete walls. The Goswick and Budle emplacements also have interesting internal features such as storage for shells built into the walls utilising drain pipe sections. These drain pipe sections are angled outwards and have scratch mark evidence of stoppers used to prevent the ammunition sliding out. This, along with the scale of construction, is very unusual and there is nothing known which is directly comparable nationally. It is even possible that they are based on German designs although there is no direct evidence for this except the visual similarities.


Fig 5.16.13 Budle battery (500), clearly demonstrating scale of construction.



Fig 5.16.14. Machine gun emplacement exposed in dunes south of Beadnell.

The more numerous type of gun emplacements are the machine gun sites. These are easily mistaken for pillboxes but can be distinguished by their larger apertures and the existence of a mounting base for a machine gun (Figs 5.16.13 and

5.16.14). Machine gun emplacements were usually placed to defend key areas on landing beaches and good examples can be seen at Beadnell Bay (657) and Bamburgh (370). The machine gun emplacements face the same management issues as pillboxes due to their similar nature and position. The larger coastal batteries are less at risk due to their larger size and more massive construction.

5.16.6.5 Radar Stations

Within the Northumberland Coast AONB only one Second World War Radar Station is extant. This is located on the Heugh at Craster (634) and is already well-known and recorded. This site was part of the Chain Home Low sites which were positioned all around the East and South coast during the Second World War. The two principal buildings, the transmitter/receiver block and the standby set block, survive. There are also numerous earthwork remains relating to the defence of the station and also the later use of the site as a Prisoner of War camp. The site was subject to a Level 3 detailed survey from 2003 – 2005 and this recorded the earthwork and structural remains in great detail. This survey also suggested the use of the site as a POW camp from 1944 onwards. This was previously based on local testimony only but the survey revealed some physical evidence including cultivation terraces and the existence of a possible exercise area (Hunt and Ainsworth 2006).



Fig 5.16.15. Surviving buildings at Craster Radar Station, later used to house Prisoners of War.



5.16.6.6 Airfields

The only Second World War Airfield surviving within the AONB is RAF Boulmer (612). This is still used by the RAF as a base for operating the Sea King search and rescue helicopters and as a Radar Station. This site occupies two separate sites, the Ops site located to the west and the Main site to the north of what was the former RAF Bomber Command station during the Second World War. The site originally was a bombing decoy, used as a decoy for RAF Lesbury, but when this smaller station closed the decoy at Boulmer became a real airfield.



Fig 5.16. 17. The surviving remains of the former runway at RAF Boulmer looking north-west from the road.

The site grew and the current main site of RAF Boulmer currently occupies the bulk of what was the Second World War air base. Several Second World War structures can still be seen in the trees opposite the main entrance to RAF Boulmer and a pillbox is visible to the east of these. On the ground much of the infrastructure surrounding the old runways is still visible in pastoral and arable fields. Most striking is the line of the old runway, obviously unsuitable for arable farming due to the concrete construction, left as pasture in a long, linear field (Fig. 5.16.18).

Several buildings are visible in the fields surrounding the old runway but these are not easily accessible as they are set within privately owned fields with little or no access. These surviving structures are set on field boundaries or next to tracks so, as with the pillboxes seen elsewhere; they face little threat of removal due to their location. The structures that would be under threat in most cases have already gone. This can be seen by a comparison of the field survey data with the data from aerial photographic survey of NERCZA Phase 1 (Fig. 5.16.19).



5.16.6.7 Civil defence structures

Civil defence structures such as air raid shelters do not often survive in urban areas due to development since the Second World War. However, in more rural areas they are less frequent due to the lower threat from bombing. In Boulmer itself, two air raid shelters survive. These brick built structures can still be seen in the yard of the church (629) (Fig. 5.16.19) and the front garden of the former coastguard watch building (630). These are fairly unusual structures to survive in such a location and could be related to the initial use of RAF Boulmer as a bombing decoy. There is evidence from Second World War aerial photography that there were bombs dropped in the vicinity, as bomb craters can be seen north of Boulmer itself.



Fig 5.16.19. Air Raid Shelter (629) in the churchyard at Boulmer. The sunken entrances are still visible at both ends.



Fig 5.16.20. Stanton air raid shelter in garden of the Old Rectory, Howick.

The explanation for the location of the two air raid shelters can be seen from the use of the proximal buildings. The church was originally used as a school house as well and this is exactly the kind of civilian building that would have had an air raid shelter despite a rural location. The other building was a coastguard lookout house and probably would have had Royal Observer Corps connections as well, again necessitating an air raid shelter.

Another type of air raid shelter was seen to the rear of the Old Rectory at Howick (Fig. 5.16.20). This type of shelter is a Stanton shelter identifiable by its curved concrete roof. It is now used as a storage building and has had a window broken through at the rear.

5.16.6.8 Removed / Rubble remains

Some of the sites recorded consisted of defences that had been removed, destroyed or eroded by the sea. The remains of these usually survive as concrete rubble eroding out of dunes or cliffs (Figs. 5.16.22 & 5.16.23) and the original form and function of these structures has been lost. Some interpretation can be applied to these sites by utilising NERCZA Phase 1 data which has already positively identified military sites by using period aerial photography. This data can be interrogated to apply a possible interpretation to fragmentary or rubble remains and this has been done within the GIS. Although in most cases definitive identification is not possible it does highlight that there are still many possible sites buried in dune systems that may survive in some form.



Fig 5.16. 21. Destroyed pillbox (672) in the dunes between Seahouses and Bamburgh.



Fig 5.16.22. Concrete of Second World War date eroding out of the cliff base on the beach near Boulmer.



Fig 5.16.23. Probable pillbox eroded onto the beach near Seahouses; there are many such examples along this stretch of coast.

5.16.6.9 Miscellaneous

Many more ephemeral and unusual earthworks and structures were recorded, some of which have not yet been positively identified. Earthwork remains of three minefields were recorded, at Birling, Dunstan (Fig 5.16.24) and also at Dunstanburgh beneath Scrog Hill. The last of these had previously been recorded by English Heritage in 2003 – 2005. Most other earthwork features were identified as anti-glider trenches or as possible platforms for buildings. Other structures visible around the Holy Island causeway have been identified as anti-glider poles deployed during the Second World War. These simple lines of wooden posts seen to the north of the causeway were used to prevent an enemy glider landing on the sands at low tide and deploying infantry. They survive today in fragmentary form although some clear lines can still be seen.



Fig 5.16.24. Evidence of the former minefield between Dunstanburgh and Beadnell. This is one of several craters created by the removal of the mines

Unusual concrete structures at Budle (502) (Fig 5.16.25) have previously been associated with the battery structure. Although clearly associated with the coastal battery (500), this feature is actually part of an earlier industrial complex associated with a quarry several kilometres to the south. The Budle battery (500) is constructed on top of an embankment for bringing quarry tucks filled with stone to the site. The miscellaneous concrete structures are actually the remains of winding gear and machinery to transfer the stone to the pier to the east.



Fig5.16.25 Concrete base for winding gear for the quarry 2km to the south.

5.16.7 Discussion and Conclusions

The Northumberland Coast Area of Outstanding Natural Beauty has a large range of archaeology within it. The archaeology of the Second World War represents the period with the most commonly surviving evidence yet it is one of the least understood. The defences of the coastline represent temporary structures and earthworks which, unless preserved, promoted and interpreted will pass out of living memory very soon. This will make it harder to interpret and to understand in the future once the last generation surviving from the 1940s is gone.

The fact that these remains are now mapped and interpreted allows strategies to be devised for interpreting what remains to the public and also for their future management. Now that the importance and fragility of the more ephemeral earthwork remains such as trenches and weapons pits has been recognised land management polices can take these monuments into consideration as significant archaeological remains.

Building on this, the interpretation of the Second World War remains to the public will help to inform management. Once land owners and the public begin to understand the significance and function of the Second World War remains in their area they can begin to get more involved in protecting them. This is true of the Second World War more than any other period of archaeology as the events that caused the construction of these defences are still within living memory. This means that the connection to the recent past, through family members and friends, is still there which provokes an interest. This is something that was evident talking to many landowners and local residents during the course of the field survey. Further interpretation can only be beneficial and lead to local interest growing and the attraction of new visitors to the AONB.

Many of these military remains are under direct threat of erosion, preserved between the former area of open-cast mining and the ongoing erosion along the coast. Although one of the most recent periods of archaeology observed during the course of the wider NERCZA project they represent some of the most threatened and poorly understood remains. In many cases this is a direct relationship, the lack of understanding from land owners and managers cause these important remains to be dismissed due to their relatively recent origins. However this does not take away from their potential importance, especially when considering the unique nature of remains such as the Budle and Goswick gun emplacements which are singularly important regionally and potentially nationally.



Fig 5.16.26 The track for the quarry leading to the winding gear base at Budle













5.17 Summary of Archaeological survey results

5.17.1 Supplementary sites

The NERCZA Phase 2 survey also covered the site at Whitburn south of Newcastle (NZ 41026 61257). Here a full survey was not undertaken but one site, identified from observation on Google Maps and highlighted by English Heritage's Regional Science Advisor, Jacqui Huntley, was added to the record. This was initially identified as a possible break water, however further investigation of the shape and location of the feature suggests it is a large, stonebuilt, fish trap. The construction is very similar to those seen at Budle Bay, and it is currently thought that those structures are related to a monastic grange of Linisfarne Priory and date to the medieval period. The feature at Whitburn was recorded and added to the record but no detailed assessment of the surrounding environment was undertaken as with the other sites surveyed within NERCZA Phase 2. Other supplementary sites have also been included within the other NERCZA surveys and are discussed in the previous sections.



Fig 5.17.1 The Whitburn fish trap as seen on the ground from the west.

5.17.2 Summary of results

The NERCZA Phase 2 survey has covered 142km of coastline and identified 609 archaeological sites. However as the foreshore, dunes, cliff tops, wetlands and hinterlands surrounding each survey area have also been investigated a larger landscape context around the individual sites has also been covered. A huge variety of coastal landscapes with many different, specific management issues were covered, including SSSIs, AONBs, and nature reserves. Each of these has a current record highlighting the nature and extent of the remains at each site but also their current condition and any threat from erosion. The survey therefore has provided much more than a simple statement on what remains and how it is threatened at fourteen locations, it has added depth to what was already known and also provided detail of new archaeological sites never before recorded.



Fig 5.17.2 Second World War pillbox at the First World War seaplane base at North Gare, Teesside.

The majority of these new sites have been military in origin and this is for two main reasons. The archaeology of the Second World War has only recently started to be added to the historic record. It has been known about previously but no large scale work has been undertaken to include the data in either the NMR or local HERs For this reason a lot of previously known, but unrecorded, sites have been added to the record for the first time. Many more have been recorded by surveys such as the Defence of Britain project and the location or interpretation of sites identified has subsequently been modified by Phases 1 and 2 of the NERCZA.

It is important to note the Phase 1 aerial photographic survey has meant the recognition of many of these military for the first time. However, Phase 2 has also identified a large quantity of more subtle features associated with these sites, such as trenches and weapons pits, that show them to survive much more extensively than previously thought. As a result of this the majority of new sites recorded by the Phase 2 survey are Second World War military features proximal to or set within the boundary of these previously known locations.



Fig 5.17.3 The quarry at Loftus Alum works, North Yorkshire, viewed from above.

Although the project has identified many military sites and these make up the largest proportion of newly discovered sites, all other periods have been represented as well. The NERCZA Phase 2 field survey has recorded sites from the Mesolithic, Iron Age, Romano-British, Early Medieval, Medieval, Post-Medieval, First World War, Second World War and Cold War periods. There is now a database of all records which will allow informed management of these sites based on their current condition, and future threat. This will allow any possible future projects to go back and monitor these sites and inform management policy further according to changing conditions and rates of erosion using the NERCZA Phase 1 and 2 data as a baseline dataset. The outlines and proposals for management of the archaeological sites, plus a tabulated list of the most threatened archaeological sites, are included in Section 7 of this report. This summarises the main management options and issues highlighted by the NERCZA Phase 2 survey.

6. PALAEOENVIRONMENTAL INVESTIGATION

6.1 Introduction

The NERCZA Phase 1 report highlighted four threatened sites that have been previously recorded as the location of inter-tidal peat deposits. These sites were identified at Hartlepool Bay, Whitburn, Cresswell and Low Hauxley. Each of these sites was surveyed during the course of the field survey and all visible intertidal peat deposits mapped using the same methodology as that adopted for the archaeological features (see Chapter 4). Investigation at Whitburn and Cresswell found no evidence of visible peats, which had been recorded in antiquarian records at Whitburn and in recent palaeoenvironmental studies by Ian Shennan beneath the active dunes at Cresswell. This is because the peat layers at these sites are currently covered by a substantial layer of sand. Exploratory coring was undertaken at these locations but no evidence of the previously recorded peats was found.

At Hartlepool, the area of exposed peat at the south end of the bay at Seaton Carew has been recorded, mapped and dated and a detailed report produced (Waughman *et al.* 2005). As a result of this further work on this peat was not undertaken in favour of attempting to map the northerly extent of what was described as a submerged forest by Trechmann (Trechmann 1936). Exploratory coring at the north end of Hartlepool Bay found an organic layer which could be the edge of a desiccated peat layer, located at the western edge of the Hartlepool headland to the east of the docks.



Fig. 6.1 A band of exposed peat in the eroding cliff section at Low Hauxley, Northumberland, at low tide. Wave action is currently undercutting the soft cliff sediment (till) resulting in the collapse of the peat layers and dune sand above. Material is lost on most tides.

The peat beds exposed at Low Hauxley are some of the most exposed and best known on the North East coast. They have already been discussed in the context of the Mesolithic-Bronze Age archaeology associated with them in section 5.9. Although two separate 'peat' beds had been recognised before (e.g. Tipping 1994), this survey has established at least five separate peat beds at Low Hauxley (A-E below), one of which was previously unknown, and the visible bands that can be seen within the cliff section do not form one continuous sediment unit. These different units have been accurately mapped as part of this study and those peats that have not previously been subjected to radiocarbon dating have been dated. The new peat bed identified at low Hauxley has human and animal impressed footprints surviving on its surface and this thin organic horizon has been dated to the Late Mesolithic.

Samples were also taken from an organic deposit initially thought to represent a possible early land surface that was observed at Crimdon Dene, and which appeared spatially related to the position of the prolific flint scatter described Raistrick and Westoll (1933). However this surface ultimately proved to be a modern deposit (see radiocarbon results below).



Fig. 6.2 Excavation and recording of the test pit at Crimdon Dene.

6.1.1 Aims and Objectives

The aim of the palaeoenvironmental survey was to accurately survey areas of inter-tidal peats and organic sediments and to collect and submit material suitable for radiocarbon dating at those sites for which no dating evidence was available, as well as to assess the potential of each peat to contain palaeoecological remains suitable for understanding past environments.

At Hartlepool the aim of the investigation was to establish the depth, extent and date of the peat bed at the north end of Hartlepool Bay. This would help in

understanding the significance of the peat and whether it has the potential to contribute to palaeoenvironmental reconstructions.

At Low Hauxley some dates had already been obtained on peat exposures to the north of the Mesolithic-Bronze Age site but the other peats are of unknown age and so it is currently difficult to assess the relative significance and value of each peat bed and how, if at all, they relate to each other. Furthermore, the earlier dates are from samples with generally large age ranges. A targeted programme of accurate survey and dating was required to disentangle this complex suite of geomorphological deposits.

Crimdon Dene was not initially identified for sampling, however upon identification of the possible buried land surface during the field survey further investigation was deemed necessary. Although no worked flint was retrieved from the layer, if it proved to date from the Mesolithic period this would help not only in identifying Trechman's prolific lithic site but would also help establish the relative significance of this organic deposit. Further investigation to relocate and accurately map the position and extent of the lithic scatter could then be undertaken.

6.1.2 Methodology

At each site samples were collected using a sand auger with an open chamber, and samples were placed directly into plastic finds bags. These were then labelled, double bagged and kept in plastic tubs. Each sample was catalogued and refrigerated until sent for specialist pollen and macrofossil assessment and subsamples taken for radiocarbon dating. Suitable samples from the targeted peat were selected for dating in a meeting with John Meadows from English Heritage's Scientific Dating Team and Jacqui Huntley, the English Heritage Regional Science Advisor.

The samples from Crimdon Dene were collected differently, being sampled by excavation of a test pit through the dune sand (Fig 6.2). The same collection and storage methodology was followed. This was also true of the sampling of the peat layer containing the footprints at Low Hauxley where a larger sample was taken in order to give the best chance for retrieval of datable material, as the peat had been re-covered in beach sand when the sampling took place.

6.2 Radio-Carbon Dating

By John Meadows and Clive Waddington

Each sample, other than OxA-22797 (Table 6.1, 6.2 and 6.3), consisted of a single waterlogged plant macrofossil, identified by Charlotte O'Brien of Durham University. Dana Challinor re-examined the Hartlepool Bay wood fragments to select those with minimal intrinsic age. The samples were dated by Accelerator Mass Spectrometry (AMS) radiocarbon dating at the Oxford Radiocarbon Accelerator Unit (technical procedures are described by Bronk Ramsey *et al.* (2002; 2004), and at the Scottish Universities Environmental Research Centre in East Kilbride (SUERC; technical procedures are described by Vandenputte *et al.* (1996), Slota *et al.* (1987), and Xu *et al.* (2004)). Internal quality assurance procedures at both laboratories and international inter-comparisons (Scott 2003) indicate no laboratory offsets, and validate the measurement precision given.

Deleted: top

The BP results reported in Tables 6.1, 6.2 and 6.3 are conventional radiocarbon ages (Stuiver and Polach 1977), quoted according to the format known as the Trondheim convention (Stuiver and Kra 1986). Their calibrated date ranges have been calculated by the maximum intercept method (Stuiver and Reimer 1986), using the program OxCal v4.1 (Bronk Ramsey 1995; 1998; 2001; 2009) and the IntCal09 data set (Reimer *et al.* 2009), and are quoted in the form recommended by Mook (1986), rounded outwards to decadal endpoints. Fig 6.3 shows the calibration of these results by the probability method (Stuiver and Reimer 1993), again using OxCal 4.1 and the IntCal09 calibration data. The probability that a sample dates to a particular calendar date corresponds to the height of its probability distribution at that date.

Comparison of the radiocarbon results from each peat exposure sampled at Low Hauxley has been undertaken using Ward and Wilson's (1978) test of statistical consistency. This produces a test statistic, T', which should be less than 3.8 in 95% of cases where two samples are of the same radiocarbon age (which they will be, when they are of the same calendar age). Thus the two results from Low Hauxley A (711) are statistically consistent (T'=0.3), as are the two from Low Hauxley C (713) (T'=0.0). In these cases, we have no reason to believe that the two fragments dated are different in date, and we would tend to accept the results as indicative of the date of deposition of the sediment sampled and therefore the date after which peat accumulation commenced.

By contrast, neither the pair of results from Low Hauxley D, 7 (715) (T²=506.2), nor those from Low Hauxley E, 13 (750) (T²=7.8), are statistically consistent, and it is not clear which, if either, result is the better estimate of when the sediment sampled was deposited. Ordinarily we would use the later result as a *terminus post quem* for sedimentation. In the case of the peat with the human and animal footprints, Low Hauxley E, there is only a small difference in date between OxA-22735 and SUERC-30015 and this is probably due to the effects of compression in this thin peat lens, or the time taken for a few cm of sediment in this sample to accumulate. The Late Mesolithic date, in the last centuries of the 6th millennium cal BC, provides a significant new dimension for understanding human activity and natural coastal change at Low Hauxley in a period that did not appear to be encompassed by the previously dated peats. The dating of this peat bed is of further significance as it contains not only human and animal footprints but also worked timbers, one of which has shown evidence for having been worked with stone tools.

The difference between OxA-22734 and SUERC-30008 from Low Hauxley D, 7 (715), at the base of this sediment unit is considerable, perhaps as much as1500 years, and it is probably better to regard the latest of the two dates as a *terminus post quem* for the commencement of sediment accumulation until further dates are available. The stratigraphically later *Iris* seed from the top of the sediment unit (SUERC-30014) dates to the early Iron Age, indicating that peat formation ceased at this time before subsequent dune sand accumulation.



Fig. 6.3 Calibration of the Low Hauxley and Hartlepool Bay radiocarbon results by the probability method (Stuiver and Reimer 1993), using the IntCal09 calibration data (Reimer *et al.* 2009).

The Hartlepool Bay samples (from two points at the top of a buried organic deposit) are both prehistoric and clearly of different date, which suggests that if the samples are more or less *in situ* and the buried land surface between them is continuous, parts of it must have been substantially truncated, perhaps by dredging activities.

The F¹⁴C ('fraction modern') results are from samples with elevated radiocarbon contents, due to the 'bomb spike' in atmospheric ¹⁴C levels caused by atmospheric nuclear testing in the 1950s and early 1960s. Kueppers *et al*'s (2004) calibration data has been used to convert these to calendar date ranges in Table 6.2 (Stuiver and Reimer 1986) and Fig 6.4 (Stuiver and Reimer 1993). The Crimdon Dene peat deposit thus appears to have formed in the late 1950s (or possibly in the mid-1990s). The two results from spit 1, taken for statistical consistency, have not been tested as the 'bomb spike' is so extreme in this period that leaves growing months apart would give inconsistent radiocarbon ages.



Calibrated date (cal AD)

Fig. 6.4 Calibration of the Crimdon Dene radiocarbon results by the probability method (Stuiver and Reimer 1993), using the Kueppers et al (2004) calibration data.

Each of the sites that have been successfully sampled and dated as part of this project are discussed in greater detail below.

6.3 Hartlepool Bay

6.3.1 Location and background

The samples at Hartlepool Bay were taken from the North side of the bay between the headland and the harbour (NGR: NZ 5662 3357). The landscape is a small embayment with a sandy beach overlying the edge of the rock outcrop of the headland.



Fig. 6.5 The small embayment east of Victoria Harbour from where samples were collected.

6.3.2 Previous research

There is a long history of research and investigation of the submerged peats at Hartlepool Bay in the area around Seaton Carew. Samples taken from these peat beds and these have produced two sets of dates dating to the Early Bronze Age (Waughman *et al.* 2005). The report complied by Tees Archaeology details the results of all of these interventions and sampling programmes. Previous sampling was also undertaken as part of a commercial evaluation of Victoria Harbour which revealed similar organic deposits (O'Brien 2006).

6.3.3 Threat from erosion

The area subject to survey is not currently threatened by direct erosion due to a substantial covering of sand. However, during periods of storm activity this could easily be removed, as has been seen elsewhere along the coast, placing the deposits at risk. The deposits could also be threatened by any future development of the harbour entrance.

6.3.4 Pollen analysis

By Charlotte O'Brien

Pollen was poorly preserved in the samples from Hartlepool Bay. A few *Quercus*, *Corylus*, and Chenopodiaceae (goosefoot family) pollen grains, and fungal and fern spores were recorded (O'Brien 2010).

6.3.5 Radiocarbon dating results

Sample	laboratory	δ ¹³ C	radiocarbon age	calibrated date range
_	code	(‰)	(BP)	(95% confidence)
719 top of peat	OxA22798	-28.3	4199±36 BP	2900-2660 cal BC
720 top of peat	OxA22736	-26.5	5901±33 BP	4850-4700 cal BC

Table 6.1 Radiocarbon results from Hartlepool Bay.

6.3.5 Summary and conclusions

Samples were taken from an organic layer identified as a possible desiccated peat from six separate cores. Two of the cores (719 and 720) provided suitable material for pollen analysis and C14 dating from the top of the sample. However, the sample was very wet and the lower portion of the samples had dropped from the chamber so samples from the base of any unit were unable to be obtained.

Sample 719 produced a date ranging from 2900-2660 cal BC and dates to the later Neolithic period. Sample 720 produced a date of 4850-4700 cal BC and dates to the Late Mesolithic. This broad date range comes from two samples of what was initially thought to be the same organic, possibly desiccated, peat layer as both samples were located within 10m of each other. This could indicate differential accumulation of separate organic deposits along this stretch of the coast, as several of the cores produced no material at all.

However, it is possible that these samples represent a continuous peat bed or land surface, and if this is the case it means that the peat that produced the Late Mesolithic date from the top of its profile must have been heavily truncated – having lost the Neolithic material above but that still survives in other locales of the bay as indicated by the other dated core. The truncation could have been caused by the construction of the harbour, the medieval town walls or by subsequent dredging activities. Despite being truncated, these sediments represent a valuable historic asset as they contain material that can inform upon the coastal Late Neolithic and earlier environment. Coastal peats dating to this period have not yet been found elsewhere in the Tees region and so they represent a significant palaeoenvironmental resource that would repay further and more detailed investigation and recording. The sediment is currently protected by a thick layer of sand and as a result is not at any immediate threat of erosion, but may eventually be exposed and placed at risk as a result of rising sea level.



6.4 Crimdon Dene

6.4.1 Location and background

Crimdon Dene is located on the Durham coast north of Hartlepool (NGR NZ 48913 36566). The geology of the area is Magnesian Limestone overlain by boulder clay. The coastal cliffs are broken by narrow, deeply incised valleys, or 'Denes', that wind their way to the coast.

6.4.2 Previous research

Although the lithic scatter site and 'forest bed' at Crimdon Dene has been seen and recorded previously (Raistrick and Westoll 1933), no palaeoenvironmental sampling has previously been undertaken in the area. The details of the previous archaeological research and current field survey are provided in section 5.5 of this report.



Fig. 6.7 Crimdon Dene viewed from the cliff to the south of the estuary mouth.

6.4.3 Threat from erosion

There is an ongoing risk of erosion and destabilisation of the dune cliff, combined with erosion caused by the cutting back of Crimdon Beck and this has led to a high rate of retreat. This is described in detail in section 5.5.12.

6.4.4 Pollen analysis

By Charlotte O'Brien

Pollen was not recorded in Spits 1, 3 and 4 from Crimdon Dene, and the only pollen noted in Spit 2 was a *Pinus* (pine) grain. A few diatoms and fungal spores were noted in Spit 1 (O'Brien 2010).

Sample	laboratory code	δ ¹³ C (%)	radiocarbon age	calibrated date range
Spit 1	OxA-22731	-25.4	1.06020±0.00294 F ¹⁴ C	Cal AD 1957
Spit 1	SUERC- 30007	-26.9	1.1209±0.0045 F ¹⁴ C	Cal AD 1957-96
Spit 2	OxA-22797	-30.6	1.03979 ±0.00328 F ¹⁴ C	Cal AD 1956-7

6.4.5 Radiocarbon dating results

Table 6.2 Radiocarbon results from Crimdon Dene.

6.4.5 Summary and conclusions

The deposit sampled at Crimdon Dene is clearly a modern deposit and is therefore not related to the flint scatter as was initially thought possible. Despite apparently fitting the location, as described by Coupland in 1936, the deposits observed were most likely formed in the 1950s or even as late as the 1990s. This could indicate that the layer observed by Coupland in the 1930s has been subsequently buried by episodes of dune creation and stabilisation. An alternative explanation is that the visible extent of organic material observed as containing worked flint in the 1920s and 30s has now been eroded away through natural processes. Further work could usefully be undertaken to try and relocate and record the potential location of the flint scatter as sea level rise and coastal retreat continues.



6.5 Low Hauxley

6.5.1 Location and Background

The main archaeological site at Low Hauxley comprises an area of locally high ground that forms a small hillock or knoll. The archaeological remains on this knoll include a Mesolithic occupation site and a Beaker-Early Bronze Age period cemetery. Since this period there has been a considerable accumulation of dune sand across the site and this has been subject to a complex sequence of geomorphological processes (Innes and Frank 1988). These processes have meant that the landscape has seen a number of significant changes since the beginning of the Holocene.

To either side of the knoll are 'peat' beds that can be followed along the cliff section for several hundred metres to the north. However, not all the peat layers are from the same sediment unit and so each unit has been carefully mapped and photographed (see Fig 6.1). Some of the units have been investigated before, Low Hauxley A and B, and the priority of this survey was to record and date those peats that had not previously been examined. This included the newly discovered peat at a lower elevation that contained the remains of human and animal footprints (Low Hauxley E).



Fig. 6.7 View along the cliff face at Druridge Bay with a recently eroded block of peat collapsed onto the foreshore from peat formation Low Hauxley B (June 2009).

The site looks directly out on to the North Sea. The sea has evidently cut back into the dune system since the Bronze Age meaning that the cemetery is now a coastal site, although when it was originally in use it would have been set back from the shore on a knoll surrounded in full, or in part, by coastal wetlands or lagoons. The current foreshore in front of the dune system comprises a rocky foreshore with interbedded sandstone, mudtsones and coal, all of which outcrop in the inter-tidal and foreshore area, depending on the amount of beach sand cover at any one time. To the rear of the dune system a huge swathe of land has been exploited for open cast coal extraction which has meant that the strip of sand dunes is the only surviving band of natural surficial deposits, and which seals an extremely rich palimpsest of archaeological remains, especially in the central and northern part of Druridge Bay (see also separate 'Review of archaeological interventions and site condition' by Waddington 2010). Currently this precious and well-preserved resource is now under active and severe erosion from the seaward side (SMP 2).

A Devensian blue-grey weathered till, which varies in depth along the coast, directly overlies the solid geology (Innes and Frank 1988). The cemetery, at which a rescue excavation took place as part of this project (Waddington and Cockburn 2009), is positioned on a localised high point approximately 100m north of the Bondicarr Burn where debouches into the North Sea. The dune sand that seals the prehistoric archaeology and peat deposits along this section of coastline have an average depth of 3.5m, although this varies between 3m and 4m. Within the sand dunes are thin lenses of organic material which represent old land surfaces and turf lines (palaeosols) that have formed during episodes of dune stability since the Early Bronze Age and thus show the potential of the dune system to provide palaeoenvironmental information on later periods as well. These buried soils represent the top of the dune system during earlier periods prior to further accumulation.

Inset within the glacial till, and below the dune system, are organic peaty deposits. These deposits are sometimes described as 'ancient forest bed' or 'inter-tidal peat', though in the case of Low Hauxley they are probably more accurately described in most cases as in-filled lagoons. These thick bands of peat, typically up to 1m in thickness, have been the subject of earlier work (Frank 1982; Innes and Frank 1988; Farrimond and Flanagan 1996 and Wilson *et al.* 2001). They contain the visible remains of old trees and have produced archaeological material including chipped flints from Low Hauxley B (Jim Nesbitt pers comm.). One of the peats close to the Low Hauxley cemetery, Low Hauxley B, is known to span the Neolithic-Early Bronze Age periods (Drury 1995) and the long peat exposure at the northern end of Druridge Bay, Low Hauxley A, has been estimated at having built up over a *c.*1900 year period (Frank 1982), although dating as part of this project suggests the origin of this peat is earlier than previously thought and in parts has accumulated over a *c.*3000 year period.



Fig. 6.8. Area of shell midden, possibly Mesolithic in date, exposed in the cliff face immediately above the till deposit. This had been eroded away by the time of the 2009 excavation.

6.5.2 Previous research

A full review of previous archaeological and palaeoenvironmental research and investigation can be seen in the accompanying report (Waddington 2010).

6.5.3 Threat from erosion

The threats faced by the resource at Low Hauxley are discussed in detail in section 5.9.9 of this report.

6.5.4 Pollen analysis

By Charlotte O'Brien

Pollen was present in all of the samples from Low Hauxley except context (1000). *Alnus* (alder) pollen was abundant in several, for example contexts (706), (709), (711), (713) and (715), while *Sphagnum* spores were predominant in contexts (705) and (708). Other species noted were *Quercus* (oak), *Corylus* (hazel), *Salix* (willow), ferns including *Polypodium* (polypody), Poaceae (grasses), Ericaceae (heathers), *Betula* (birch), *Pinus* (pine) and herbaceous taxa including *Plantago lanceolata* (ribwort plantain), Fabaceae (pea family) and Apiaceae (carrot family) (O'Brien 2010). The various peats at Low Hauxley have all shown good preservation of botanical macro and micro fossils with the collective potential to inform on palaeoenvironmental reconstruction from the Late Mesolithic through to the Early Iron Age, as well as hosting archaeological remains dating from all of these periods.

Sample	laboratory	δ ¹³ C	radiocarbon age	calibrated date range
*	code	(‰)	(BP)	(95% confidence)
711 (Low Hauxley A)	OxA-22732	-26.2	5915 ±31 BP	4850–4710 cal BC
711 (Low Hauxley A)	SUERC- 30010	-28.8	5940 ±35 BP	4930–4720 cal BC
713 (Low Hauxley C)	SUERC- 30009	-28.5	4675 ±35 BP	3630–3360 cal BC
713 (Low Hauxley C)	OxA-22733	-26.8	4674 ±30 BP	3630–3360 cal BC
714 (Low Hauxley D)	SUERC- 30014	-28.9	2505 ±35 BP	790–510 cal BC
715 (Low Hauxley D)	OxA-22734	-27.8	3776 ±29 BP	2290–2050 cal BC
715 (Low Hauxley D)	SUERC- 30008	-28.7	4790 ±35 BP	3650–3510 cal BC
750 (Low Hauxley E)	OxA-22735	-25.5	6296 ±34 BP	5330–5210 cal BC
750 (Low Hauxley E)	SUERC- 30015	-28.1	6160 ±35 BP	5220–4990 cal BC

6.5.5 Radiocarbon dating results

Table 6.3 Radio carbon results from samples collected at Low Hauxley.

6.5.5 Summary and conclusions

The dated samples from the various peat exposures at Low Hauxley reinforce the view of these organic units being separate geomorphological entities, with each formed at a different time period, although in most cases with periods of overlap.
For ease of identification each of the visible peat layers at Low Hauxley has been given a letter A-E (Fig 6.9 and Table 6.3 above). A trend, perhaps significant, that can be noted from the dating of the deposits is that the on-set of peat accumulation gets younger from North (A) to South (D), with the exception of layer E, which is the earliest and most shortlived of all the deposits, but which is at an altogether lower altitudinal level.

Low Hauxley E has provided the earliest dating evidence (sample 750 in table (6.3), and this is in line with expectations given that the layer is at a lower elevation than the other observed peat layers. The dates of 5330-5210 cal BC and 5220-4990 cal BC, show that this peat formed during the late Mesolithic period in the final centuries of the 6th millennium cal BC. This layer also contained worked timber showing cut marks, apparently made by stone tools, and the impressions of human and animal footprints were also observed on its surface. Although the sample only provided dates for the basal deposit, the deposit is very shallow being only 6cm thick, and so was probably only shortlived as a wet peaty deposit. In order for the footprint impressions to have survived the peat must have been soft and damp when they were made and then become dried out, and perhaps covered in sand, very shortly afterwards. Therefore, it is difficult to entertain a scenario whereby the footprints could be much later than the *terminus* post quem provided by the Late Meoslithic dates from the base of the deposit. This makes both the peat, the footprints and the substantial quantity of worked wood surviving in this deposit highly significant historic assets, and extremely rare ones, which are undoubtedly worthy of further investigation (see section 7.3.2), particularly as this is a section of coastline under continuous and severe erosion due to rising sea levels. This peat layer has high potential to yield further archaeological material and dating evidence for this significant period of human history about which little is known from this region. Furthermore, it has the opportunity to shed light on much bigger questions relating to the final drowning of the North Sea, the Mesolithic coastal settlement of northern England as well as details of how people lived, procured resources and adapted to and manged their environment. These are questions of national and international significance and this site, which is under severe and continuous erosion, has the ability to contribute significant information to these questions. . The layer is currently protected by up to 1m of sand in places, however this is removed during storm events and the peat layer exposed and further eroded. As a result once this peat layer becomes exposed again, usually in the winter months, further recording and sampling should be undertaken. This is discussed in further detail in section 7.3.2.

Low Hauxley A was the next oldest dated layer, returning dates from the base of the layer of 4850-4710 cal BC and 4930-4720 cal BC. This immediately post-dates the layer containing the footprints and also started to form in the Late Mesolithic. Investigation of this layer has revealed numerous protruding tree trunks and logs indicating that it has the potential to produce worked timber similar to that retrieved from Low Hauxley E. Mesolithic flints have also been reported as coming from this layer (Jim Nesbitt pers comm.). The upper lens of this sediment unit was dated to the Late Bronze Age 1060-840 cal BC during an earlier study by Innes and Frank (1988). This is evidently another significant prehistoric resource of high palaeoenvironmental and archaeological potential. This peat is currently exposed in the cliff face and is actively eroding. This layer can be seen along with the other exposed peat layers and knoll site at Low Hauxley, as the most threatened group of archaeological resources on the North East Coast (see section 7.2). By comparing the accurately surveyed positions of these peat layers and the Mesolithic-Early Bronze Age archaeological site on the knoll with the SMP2 projected coastlines, the SMP projection data can be seen as woefully inadequate. The current position of the exposed peats are already beyond the projected 20 year and 50 year shoreline projections and are only just within the 100 year projected future coastline (see Fig 6.9). Clearly, the SMP2 study has underestimated the rate of coastal erosion along this stretch of coast and it is in need of urgent review.

Low Hauxley B has already been dated in some detail by Tipping (see Drury 1995) and so no samples were submitted for dating from this unit. This unit is under the same ongoing threat as Low Hauxley A. This peat has a basal date of 3650-3350 cal BC and a date form the top of the horizon of 710-210 cal BC, ie. Neolithic-Iron Age date (see review document, Waddington 2010).

Low Hauxley C has returned dates of 3630-3360 cal BC and 3630-3360 cal BC from the base and this consistency between the two dates shows a formation period for the peat in the Early Neolithic broadly contemporary with the formation of Low Hauxley B. This layer has also been observed to contain flintwork and is threatened by ongoing and rapid erosion as with Low Hauxley A and B.

Low Hauxley D is located directly south of Low Hauxley C and has returned dates of 2290-2050 cal BC and 3650-3510 cal BC at the base of the deposit and a single date of 790-510 cal BC from the top of the deposit. If the earlier date is correct then it would again indicate a date of formation co-eval with Low Hauxley B and C. However, the later, Beaker period date, could suggest that the earlier date is from residual material. Given that this sample is from a natural deposit though, it is equally possible that it is the sample producing the later of the two dates that is intrusive. Currently it is not clear either way which date more accurately reflects the on-set of peat accumulation at Low Hauxley D. Either way it appears that peat formation ceased in the Early Iron Age. This layer also contains significant sized logs and tree stumps that can be seen protruding from the deposit. This excellent survival indicates the potential for the presence of more worked timber, as with all the other peat beds at Low Hauxley.

The dating programme undertaken by this project has provided a much more detailed understanding of the various peats and their formation and cessation dates at Low Hauxley. They have provided date ranges from the Mesolithic through to the Iron Age, with one peat, Low Hauxley A, appearing to encompass the Mesolithic-Neolithic transition. The extent of survival of archaeological and palaeoenvironmental remains along the coast at Low Hauxley providess a unique opportunity to investigate the development and change of a prehistoric landscape through Late Mesolithic – Iron Age times. Additional evidence, such as the preserved human and animal footprints and timber worked with stone tools that have only been recently discovered, shows the high potential for further remains and discoveries to be made, as well as the undoubted significance of these palaeoenvironmental and archaeological resources. The level of threat, especially to the north of the Bondicarr Burn (Fig 6.9), can be seen as extremely high with

many significant archaeological and environmental deposits experiencing ongoing destruction.



7. MANAGING THE RESOURCE

7.1 Introduction

This chapter deals with coastal heritage management issues, in the light of the results of the NERCZA project to date, and the special interest of the sites identified. The assessment of site significance and prioritisation is inevitably partly subjective and is based on the professional judgement of Archaeological research Services Ltd staff in consultation with other stakeholders, although it is based on the results of consistent and objective survey. The prioritisation of sites for archaeological intervention, as outlined below, and the discussion related to each of them, are intended to provide a starting point for discussion and consideration of how best to manage sites and target resources. Given that the coastline is such a dynamic environment the condition of sites will change, as will knowledge of certain types of sites, and as a consequence the list of prioritised sites should also be revised in the light of such changes. Consequently, the priority list, and this chapter generally, should be considered a 'live' document that will change subject to further discussion across the curatorial sector and in the light of physical changes on the coastline. It is, therefore, not intended as a definitive statement but rather an aid to discussion and subsequent decisionmaking and actions.

In some cases archaeological features have been assessed individually and, where appropriate, others have been assessed as a group. For example, the surviving Second World War military features at Alnmouth have been assessed as a group, but considered separately to the 19th century oyster beds recorded at the same location. Each set of records has been assessed on the basis of their condition and level of special interest (see below), and their value as a group of surviving archaeological features also considered. Where appropriate, single features have been assessed, for example an isolated feature of high archaeological significance that is under threat from erosion or removal. Examples of this can be seen with the Budle Bay and Scremerston Second World War batteries in Northumberland.

A list of sites ranked by level of threat, condition and special interest has been produced and is displayed in Table 7.1, and the top quartile of most significant sites under threat (a total of 13) are discussed in further detail in Section 7.2. There is duplication of the numbering of policy units in the North East and Northumberland SMP2 documents and so each has been colour coded in table 7.1 to distinguish between them.

The assessment of each of the ranked sites shown in Table 7.1 has been based on five criteria. These criteria are: threat from erosion, condition, significance, potential for further investigation and rarity. Each of these criteria has been scored out of ten using the principals set out in DCMS guidance for Scheduled Ancient Monuments

http://www.culture.gov.uk/images/publications/ScheduledMonuments.pdf (formerly Annexe 4 of PPG 16) and reflects the professional opinion of the ARS Ltd project team. The scoring is based upon data collected during the NERCZA project including that from the desk-based assessment, aerial photograph transcription, field survey and consideration of current and future sea level models.

The potential for some of these sites to be proposed for consideration for designation has also been reviewed. This is not scored but stated as Yes/No/Already designated within the table. The attribution given for these sites remains the opinion of the NERCZA project team and not the current position of English Heritage.

The scoring of the various criteria gives a total out of fifty. The table lists sites in their rank order with the site considered to be at most threat and greatest significance ranked number 1. All sites listed in the table are of special interest and face some risk from erosion, and a low ranking does not mean that the site is of low significance. It is only sites of special significance that have made it on to the list in the first place, as many hundreds of recorded features have been excluded as they are not considered to be at risk in the short or medium term. The criteria for assessing each site are discussed in more detail below.

Threat: This comprises the perceived level of threat to the site from coastal erosion or other ongoing erosion. It includes consideration of land use and the potential for the site to be removed artificially. A highly threatened site facing multiple types of erosion would score 9 or 10 while a site located in a stable location with little threat from erosion over the next 100 years would score 1. SMP2 predicted shorelines for 2025, 2055 and 2105 where also used in conjunction with the project GIS to assess the possible long term threat to each site. If the archaeological site was to be lost within 20 years using these predictions the threat would score higher, whereas if the shoreline projection indicated that it could survive for a further 100 years the score would be lower.

Condition: This score is based on the current condition of the site in question; a site which is an exceptional example of its type which survives mostly intact would score highly, while a site that survives in fragmentary form, or is mostly destroyed, will have a low score. The context of a site was also considered in this assessment. An archaeological site removed from its original context by later development would score lower than a site which has survived in its original context. This means that a well-preserved military site surviving *in situ* would score higher than a ploughed out Second World War crop mark site.

Significance: Assessment of significance has been based on the professional judgement of the project team with reference to the known information value, status, or historical significance of a site. This has been guided with reference to some of the criteria set out in the DCMS guidance for Scheduled Ancient Monuments

http://www.culture.gov.uk/images/publications/ScheduledMonuments.pdf (formerly Annexe 4 of PPG 16). A highly significant site will have rare archaeological features with considerable information potential and may contain components from multiple periods. A less significant site will typically comprise a single, more common archaeological feature.

Potential: This is the potential for the site to yield further knowledge or evidence which will make a significant contribution to our understanding. A site which survives intact, and is rare, may contribute more than a site that is already well known and has been extensively investigated. The score is an overall assessment

of how beneficial further archaeological work would be to furthering understanding and contributing to place-making and public enjoyment/wellbeing.

Rarity: This is the assessment of how common the site type is, while also considering its degree of preservation and integrity. Here a standard pillbox which survives intact will score slightly lower as there are numerous examples surviving along the North East coast, however a barrow or a prehistoric monument, such as the enclosures seen at Fenham and Overdale Wyke, will score more highly as there are many fewer examples.

Potential for Designation: Sites in highly threatened locations may be less likely to be considered, however significant they are. Very significant sites in stable locations are more likely to be put forward for consideration. Sites that are already designated are also highlighted. This assessment is not a direct proposal for designation but an indicator of what sites could usefully be considered for putting forward for designation, based on the opinion of the NERCZA project team. It is important to note that even sites in extremely threatened positions may still be considered for proposal for future designation and this has been taken into consideration when putting forward the opinion of the project team.

Table 7.1 sets out the key heritage assets of special interest within the study area displayed in ranked order of priority as evaluated by the project team. The sites have been divided into a hierarchy of colour-coded quartiles with red being those sites considered under 'imminent risk', orange being those considered to be under 'high risk', yellow being those considered at 'intermediate risk' and green being those sites at 'low risk'. Sites at 'imminent risk' are discussed individually in more detail within section 7.2 with specific reference to the threats faced. Sites considered to be at imminent risk are those scoring 40 or higher in the assessment. Those sites that scored between 30 and 40 are considered to be at high risk. Those scored between 20 and 30 are considered to be at intermediate risk and those lower than 20 are considered to be at low risk. Imminent risk is considered to be where there is an immediate or on-going threat to the surviving remains recorded on site and where there is also a clear need for further work. High risk is where the archaeological resource is threatened but the threat may not be as immediate, the site only being imminently threatened within the 20 year SMP2 coastline predictions. Intermediate risk sites are threatened in the long term and will only be directly threatened within the 20 - 50 year SMP2 coastline predictions. Low risk sites are those which will become threatened in the long term, the 50 - 100 year SMP2 coastline predictions, or possibly not at all using current data.

However by comparing the SMP2 predictions with the coastline as recorded by the NERCZA project the limitations of using the SMP can be seen. Using Low Hauxley as an example, the SMP2 predicted shorelines can be seen as woefully inadequate along this stretch of coast. The current line of the shore, in particular the location of the surviving peat layers known to contain archaeological material, can be seen to be further inland than the SMP2 predictions for the shoreline in 50 years time (Fig 7.2) (See also Chapters 5 and 6). This huge discrepancy demonstrates the limitations of using this data in assessing the threat to heritage assets, certainly in this part of the North east coastline, and an urgent review of the SMP2 shoreline predictions for this area is required. It would seem that the current data can only be used as a rough guide as to what will happen in the future.

Position	Sito Namo	Sito Tupo		Policy Lipit	Policy	Throat	Condition	Significanco	Potential	Pority	Potential to Designate	Total
FUSILION	Site Name	Sile Type	NERCZA OID	Folicy Offic	FOILCY	meat	Condition	Significance	Fotentia	Канцу	Yes/No/Already Designated	/50
1	Low Hauxley	Mesolithic and Bronze Age Site	332	17.3	MR	10	10	10	10	10	No	50
2	Low Hauxley	Prehistoric footprints and other peats	700	17.3	MR	8	10	10	10	10	No	48
3	St Cuthbert's Isle	Hermitage	386	4.7	NAI	10	9	9	10	10	Yes	48
4	Fenham	Late prehistoric enclosure	472	4.3	NAI	10	8	9	9	9	Yes	45
5	Budle Bay	Gun Emplacement	500	4.5	HTL	8	10	9	8	9	Yes	44
6	Trow Point	Possible Bronze Age burial	132	3.1	NAI	10	7	8	9	9	No	43
7	Amble	6 Hulks of coal wherries	352-356	15.2	MR	10	7	9	9	8	Yes	43
8	Scremerston	Late prehistoric enclosure	4526	3.1	NAI	10	7	8	8	9	No	43
9	Scremerston	Gun Emplacement	463	3.1	NAI	6	10	9	8	9	Yes	42

7.2 Prioritised list of threatened heritage assets on the North East coast of England based on the results of the NERCZA Project.

10	Budle Bay	Fish traps	520	4.5	HTL	10	8	8	8	8	Yes	42
11	North Gare	WW1 Seaplane base	201	13.4	NAI	10	6	9	8	9	No	42
12	Hartley	Roberts Battery	451	24.1	HTL	10	9	7	7	8	Already designated	41
13	Holy Island	Lithic Scatter	518	5.1	NAI	8	8	9	7	8	No	40
14	Hartley	Fort House	443	24.1	HTL	7	7	9	8	8	No	39
15	Alnmouth	Oyster Ponds	223-228	13.4	HTL	10	6	7	7	8	No	39
16	Goldsbrough	Military camp	417	21.3	NAI	9	8	7	8	8	No	38
17	Hummersea	Alum works	52	17.3	NAI	10	8	7	7	7	Already designated	38
18	Alnmouth	19th Century Battery	214	13.1	MR	6	8	8	9	7	Already designated	38
19	Loftus	Alum works	195	17.3	NAI	10	8	7	7	7	Already designated	38
20	Sandsend	Alum works	415	21.3	NAI	10	8	7	7	7	Already designated	38

21	Overdale Wyke	Prehistoric enclosures	170	21.3	NAI	7	6	9	7	9	No	38
22	Kettleness	Alum works	426	21.3	NAI	10	7	7	7	7	Already designated	38
23	Trow Point	19th century disappearing gun and WW2 defences	119	3.1	NAI	9	7	8	6	7	Already designated	37
24	Whitburn	Fishing Trap	419	6.2	HTL	7	7	8	8	7	No	37
25	Alnmouth	Chapel	232	13.8	HTL	8	8	7	7	7	Yes	37
26	Greatham Creek	WW2 Decoy site	198	13.5	NAI	7	7	8	6	8	No	36
27	Newton Point	WW2 Radio station	666	9.1	NAI	7	10	7	6	6	No	36
28	Kettleness	Mineral railway	422	21.3	NAI	7	7	8	7	7	Already designated	36
29	Saltburn	Rutways	27	16.1	NAI	8	7	7	7	7	No	36
30	Saltburn	Rock cut features	29	16.1	NAI	8	7	7	6	7	No	35
31	Saltburn	Alum works	23	16.1	NAI	10	6	6	6	7	Already designated	35

32	North Gare	Medieval Salterns	184	13.3	NAI	5	8	8	7	7	Already designated	35
33	Druridge Bay (North)	WW2 Defences		17.4	MR	9	7	7	6	6	No	35
34	Druridge Bay (South)	WW2 Defences		17.3	MR	8	7	7	6	6	No	34
35	Bamburgh	Early Medieval burials	378 - 385	6.1	NAI	3	7	9	7	8	Yes	34
36	Dunstanburgh	WW2 military complex	640-660	10.1	NAI	7	7	8	6	6	No	34
37	Craster	WW2 Radar station	634	10.1	NAI	5	8	7	8	6	Already designated	34
38	Ross Links	WW2 Military remains	800	4.5	HTL	6	7	8	8	5	No	34
39	Sandsend	Railway	416	21.3	NAI	8	7	6	6	7	No	34
40	Druridge Bay	Gun Emplacement	264	17.4	MR	8	5	7	7	7	No	34
41	Goldsbrough	Roman signal station	429	21.3	NAI	2	7	8	7	9	Already designated	33
42	Sandsend	WW2 Defences	433	22.1	HTL	7	6	6	7	7	No	33
43	Budle Bay	Quarry complex	502	4.5	HTL	8	6	7	6	6	No	33

45	Greatham Creek	WW2 Defences	141-152	13.5	NAI	7	7	6	7	6	No	33
46	Holy Island	Fort	402	4.8	HTL	5	8	6	8	6	Already designated	33
47	Bamburgh	Military complex	363 - 374	6.1	NAI	7	7	6	7	6	No	33
48	North Gare	WW2 Defences	153 -190	13.2	NAI	7	7	6	6	7	No	33
49	Skinningrove	WW2 Defences	30	17.2	HTL	7	7	6	6	7	No	33
50	Boulmer	WW2 Airfield + Defences	612	11.2	NAI	6	7	7	6	6	Yes	33
51	Fenham	Grange	470	4.3	NAI	4	8	7	8	6	Already designated	33
52	Seahouses	WW2 Trench Network	670	7.1	NAI	9	7	6	5	6	No	33
53	Crimdon Dene	WW2 Defences	81-113	11.1	MR	8	6	7	6	6	No	33
54	Skinningrove	Ironstone mine	19	17.1	NAI	7	5	7	6	7	Already designated	32
55	Frenchmans Bay	WW2 Defences	140	3.2	NAI	6	6	6	8	6	No	32
56	Trow Point	WW2 Defences	120 - 139	3.1	NAI	8	7	6	6	5	No	32
57	Scremerston	WW2 Radar Station	467	3.1	NAI	4	7	7	6	7	Yes	31

58	Druridge Bay	Bombing range markers	259, 292 + 280	17.4	MR	8	4	7	4	8	No	31
59	Embleton Bay	WW2 military earthworks	640-660	9	NAI	7	5	6	6	7	No	31
60	Beadnell	WW2 Trench Network	662	8.2	HTL	7	6	6	6	6	No	31
61	Holy Island	Quarry	519	5.1	NAI	9	7	5	5	4	Already designated	30
62	Budle	Lime Kiln	497	4.5	HTL	9	7	5	6	4	No	30
63	Scremerston	Lime Kiln	458	3.1	NAI	9	6	5	6	4	No	30
64	Alnmouth	Post medieval barn	237	13.8	HTL	6	6	6	4	7	No	29
65	Alnmouth	Disguised pillbox	230	13.8	HTL	9	4	5	2	8	No	28
66	Crimdon Dene	Mesolithic flint scatter	99	11.1	MR	10	4	9	5	9	No	27
67	Scremerston	Defensive position	459	3.1	NAI	10	3	5	4	4	No	26
68	Scremerston	Pillbox	460	3.1	NAI	8	6	3	3	2	No	22
69	Alnmouth	Enclosure	218	13.1	MR	4	5	4	4	4	No	21

70	Whitburn	WW2 AA battery	550	6.1	HTL	2	5	5	1	7	No	20
71	Sandsend	Former Railway Station	406	22.1	HTL	2	10	2	1	4	No	19
72	Whitburn	Rifle range	561	6.1	HTL	2	10	2	1	3	No	18
73	Kettleness	Former Railway station	415	21.2	HTL	2	8	2	1	4	No	17
74	Kettleness	18 th century church	418	22.1	HTL	4	6	2	2	3	No	17
75	Cresswell	WW2 Military remains	236	17.5	MR	1	4	1	5	5	Yes	17
76	Alnmouth	Possible later medieval stock enclosure	219	13.8	HTL	2	4	1	3	4	Yes	14
77	Embleton (town)	WW2 military remains	612	9.1	NAI	2	3	2	2	3	No	12
78	Alnmouth	Beacon	216	13.8	HTL	1	2	1	2	2	No	8
79	Greatham Creek	Possible military buildings	139	13.5	NAI	2	1	2	1	1	No	7
80	Seahouses	WW2 military fragmentary remains	632	7.1	HTL	2	1	1	1	1	No	6

3.1	SMP2 Policy Unit for Northumberland
3.1	SMP2 Policy Unit for North East
Druridge Bay	Imminent Risk
Druridge Bay	High Risk
Druridge Bay	Intermediate Risk
Druridge Bay	Low Risk

Table 7.2 Key to colours used in table 7.1

7.3 Priority sites of special interest at 'imminent risk'

The following is a site by site discussion of the sites identified as being at 'imminent risk' in the ranked assessment shown in Table 7.1. The reasons for the scoring of each site are discussed and possible management options for the threatened archaeological remains are identified and discussed and placed in order of preference.

As three separate archaeological elements at Low Hauxley scored high enough to be included they have been grouped into two in this discussion according to their geographic position either side of the Bondicarr Burn.

7.3.1 Low Hauxley Mesolithic site and Beaker-Bronze Age burials
 Low Hauxley; Mesolithic site and Beaker-Bronze Age cemetery (NU 28412 22705)
 Druridge Bay, Northumberland
 Policy Unit 17.3
 Managed Retreat

The archaeological asset comprising the Mesolithic occupation site and Beaker-Bronze Age burial site at Low Hauxley is especially significant. The potential for further investigation and the potential for that to further our understanding of Mesolithic settlement and Beaker period - Early Bronze Age burial, together with questions of colonisation and immigration in prehistory, makes the significance and rarity of this site score maximum. The condition of the archaeological resource that survives, being sealed under calcareous sand dunes, is excellent even when considering the ongoing effects of erosion on the archaeology. On this evidence the site scores maximum on potential and condition criteria. However, it is not known how much more of the site survives given the quantity of material that has already fallen out. Therefore, there is a need to establish in more detail what still survives on the site. In addition to the archaeology there is a sequence of inter-tidal peats immediately adjacent to this site that have considerable palaeoenvironmental, geoarchaeological as well as archaeological potential. The series of radiocarbon dates for the peat layers retrieved as part of this project demonstrate that the earliest of these sediment units formed during the Late Mesolithic and they continue to at least the Late Bronze Age in the area to the north of the Bondicarr Burn. The peats represent an archaeological resource of high significance containing Mesolithic flints and are sealed by dune sand that has revealed evidence for many other archaeological features including a pristine Late Bronze Age rapier and a circular stone-built structure, probably a roundhouse, that has now been destroyed and washed away. These peat layers are under daily erosion, and given that they are known to be, in part, contemporary with the activity represented at the Mesolithic-Bronze Age site, present a resource of considerable potential. The combination of archaeological and palaeoenvironmental deposits together at the same location ensure the site is of high significance.

The threat to the archaeological resource is serious and ongoing (Fig 7.1), with destabilisation of the cliff (Fig 7.2) a daily occurrence. In addition, the site is also under threat by robbing from members of the public as evidenced by the

wedging of a side slab for one of the small cists back into position after human bone material had been scooped out and dropped below – it is thought that a pottery vessel is likely to have been removed by this action (see chapter 5.9). Here the threat level has also scored maximum. The threat to the site is so bad that there is considered little point in designating a site that only has a few years left before its inevitable removal, and designation will not assist in its survival.



Fig. 7.1 Location of a previous archaeological excavation trench backfill (above the black plastic sheet) eroding out of the cliff at Low Hauxley.

Management options

The management options for the site are listed in order of preference, with the first being the most preferable strategy.

- Archaeological evaluation to assess scale and cost of rescue works, followed by an appropriate level of excavation and recording in conjunction with a parallel programme of palaeoenvironmental investigation. This could be combined with recording and investigating Peat E, its footprints and worked timber. In addition, further monitoring and recording of the peats and eroding remains, with community involvement, as part of the wider "Coal and Coast" project.
- Ongoing regular monitoring of exposed archaeological sediments to assess if any more significant archaeological features are exposed and record what one can of them as they fall out.
- Do nothing and allow for loss.

The favoured option of the project team is the highest possible level of recording as this would provide the most information and preserve the resource through record before its removal due to natural processes. The significance of this site, combined with the complex multi-period archaeology, requires the attention of a structured archaeological evaluation in the first instance undertaken by professional archaeologists with community support. This could involve community groups and schools and outreach allowing local people to engage with their coastal heritage and enhancing the experience of visitors, whilst also allowing possible access to wider sources of funding to allow this work to be undertaken.

7.3.2 Low Hauxley Footprints

Low Hauxley Prehistoric footprints and worked wood (NU 28302 77257) Druridge Bay, Northumberland Policy Unit 17.3 Managed Retreat

The human and animal footprints identified at Low Hauxley, Northumberland, are visible in the inter-tidal zone, in a thin layer of intermittently exposed peat, Low Hauxley E (Fig 7.3). Their extent has been recorded and a sample of worked wood was retrieved (see section 5.9 and section 6) together with a peat sample that has produced Late Mesolithic calibrated date ranges of 5330-5210 cal BC and 5220-4990 cal BC for the on-set of peat formation (see also Chapter 6). Given the shallow depth of this peat it is clearly a relatively short-lived sediment and is likely to have only been accumulating for a short period and therefore the footprints, which would have had to be formed when the sediment was still very soft and wet, are likely to date to the final wet phase of the sediments before it dried out. Therefore, it is considered very likely that the footprints also being to the Late Mesolithic period although radiocarbon dates from the top of this sediment unit are still required to provide a more accurate date for the footprint formation.

The peat containing the footprints represents a newly identified sediment unit at a lower elevation, and of an earlier date, than the other previously known peats, and therefore is extremely important in its own right as it contains a wealth of environmental evidence concerning the immediate Late Mesolithic environment in this area. Considering the existence of abundant worked wood within the layer, which was seen when the footprints were recorded, the importance and potential of the site as a resource for gaining further knowledge about human activity and the environment during the latter stages of the Mesolithic is great. The presence of human and animal footprints impressed into this layer is also extremely important as there are only three other examples of preserved prehistoric footprints in Britain, at Formby on the Lancashire coast (Cowell 2001), the Severn estuary (Allen 2004) and Hartlepool Bay (Waughman 2005).

This site scored very highly in the table as it is of very high rarity and significance, being one of only four known sites where such footprints survive. The presence of abundant worked wood within the sediment layer, together with the potential of the peat for further plant micro and macro fossils only increases the significance of the site, and inspection of the peat showed that there was a huge quantity of timber surviving within it that could shed light not only on human activities, technology and woodland management, but also a rare and detailed insight into the type of vegetation and landscape setting in this area at the time the footprints were made. The dating of this layer to the Late Mesolithic indicates that it could be contemporary with the Late Mesolithic occupation site below the

Bronze Age cairn cemetery at Low Hauxley (see above) that was radiocarbon dated by Bonsall (1984) to a similar time bracket ("A single radiocarbon determination on a sample of shells from the midden suggests an age of about 5000 bc"), although the latter date has never been fully published. If the two sites are contemporary, as seems possible, then the group value of these remains is even higher and provides a very rare opportunity to understand a Mesolithic occupation site in relation to a submerged peat, with clear evidence for human activity within it, as well as human interaction with the environment in terms of woodland management and associations with animal activity. This site could provide a counterpart to Star Carr, except in this case it would provide an unparalleled level of detail on human activity in the Late Mesolithic prior to the introduction/adoption of farming and in a coastal setting.

The threat to the site from coastal erosion is significant as the peat layer is scoured clean of the overlying sand during storm events and the site scored moderately high as a result of this assessment, despite it being sometimes covered by beach sand - that protects the resource during the calmer summer months. When revealed again the peat in which the footprints and wood are preserved is very shallow and would be prone to erosion from the tide and beach walkers as it is situated within the inter-tidal zone. During a storm event this peat could be scoured away completely removing the evidence of the footprints, and probably the entire layer of peat and worked timbers altogether.

The potential to propose the site for consideration for designation is low as it is difficult to see how any kind of designation would help the site in terms of its survival, given that it is being affected by an inexorable natural process.



Fig. 7.2 Location of Footprints and samples at Low Hauxley.



Fig. 7.3 The briefly exposed footprints at Low Hauxley showing how much of the sediment has already been eroded away, and the position of the worked wood find, with the Bondicarr Burn outflow in the background, looking north (Scale = 2m).

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being considered the most preferable strategy.

Full recording of the footprints next time they are revealed, utilising an accurate GPS plot of the full extent together with a full, hand drawn plan of the whole peat layer at a scale of 1:50 and detailed drawings, measurements and photographs for each footprint at 1:20. As well as this recording strategy, casts of some of the best-preserved footprints could be taken and a full photographic survey of their visible extent made. This would allow analysis on the direction of travel, the number of individuals, and possibly even ages, sex, as well as the species of animals and approximate number of individuals. Further samples of wood should be taken to allow more detailed analysis of woodland management and woodworking techniques (Taylor 2010) with the wood specialist involved on site in selecting samples for analysis. This would undoubtedly contribute to the understanding of prehistoric life in this part of Britain as there are very few examples of prehistoric worked timber known from the region, and from this period more generally. Further samples of the peat to be taken for environmental assessment and examination fro archaeological residues such as worked flints and suitable radiocarbon dating samples from the top of the peat layer and any other significant parts of the sediment unit or deposits within it.

- Ongoing monitoring of the visible extent of the footprints and basic recording to assess their condition and any increase/change in the nature of the threat faced in this location.
- Do nothing and allow the resource to be lost.

The favoured option of the project team is the highest possible level of recording as this would provide the most information and contribute significantly to our understanding of the Late Mesolithic period both at a regional scale, and nationally/internationally, whilst also preserving the resource through record before it is lost as a result of natural processes. If time and funding does not allow for full recording then the next best approach is to utilise and encourage motivated local amateur archaeologists, such as Jim Nesbit, to continually monitor the exposure and condition of the site. The least favourable option is to do nothing as the sediment unit will eventually be completely exposed and removed through natural process, losing a nationally valuable archaeological resource.

7.3.3 St Cuthbert's Isle

St Cuthbert's Isle; Hermitage (NU 12289 772568) Holy Island, Northumberland **Policy Unit 4.7 No Active Intervention**

The site comprises the surviving structural remains of a hermitage, believed to have been the site initially occupied by St Cuthbert in the 7th Century AD. What is visible now represents a later medieval structure built on the site. The site is thought to have been initially occupied by the saint who eventually settled on Farne Island near Bamburgh. However, the site is still of historical significance as there is a possibility of well-preserved medieval archaeology relating to a small early medieval hermitage as well as the later chapel that still survives in ruinous state on the site. For this reason the site is potentially highly significant and, considering the threat faced by its location (Fig 7.5), can be seen to be placed at high risk. There are also well-preserved remains of at least two small buildings and associated earthworks, and the site therefore scored very highly on condition, potential and significance.

The possible link to St Cuthbert adds considerable significance and potential importance to the site, and justifies the high rarity value score. The threat to the site is also very high as archaeological deposits are being actively eroded at every high tide (Fig 7.4), with part of the western wall of the structure on the isle already having been lost. This is the reason for the site scoring a maximum in this category.



Fig. 7.4 Actively eroding archaeological remains on St Cuthbert's Isle, Northumberland, viewed at low tide looking east.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable strategy of the project team.

- Full Level 3 detailed survey of above ground remains on St Cuthbert's Isle at an appropriate scale, followed by rescue excavation of the western limits of the monument already being lost to erosion and ongoing monitoring of future erosion. There is also potential for geophysical survey, test-pitting and full excavation if the threat increases, possibly as a part of a wider community project.
- Level 3 survey of the whole of St Cuthbert's Isle at a scale of 1:500 and on-going regular monitoring of exposed archaeological sediments to assess if any significant archaeological features are exposed.
- Do nothing.

The favoured option of the project team is the highest possible level of recording as this would provide the most information and important knowledge gain whilst preserving the eroding resource through record, prior to its removal by natural processes. The site is exposed and archaeological remains are rapidly being eroded away, making at least Level 3 survey and recording of the exposed section a priority. If full survey or archaeological excavation cannot be undertaken in the near future some form of ongoing monitoring to evaluate the situation must be undertaken. This could be done by a local group, or island residents, as it would only require regular photography passed on to the local authority and English Heritage. However, without the scope to react to further erosion the monitoring would in itself be of little value. It would only serve to highlight a problem, raise expectations and local feeling, only for it to be dashed by no action being taken and the remains left for their inexorable removal.



Fig. 7.5 Location of St Cuthbert's Isle, off the south coast of Holy Island.

7.3.4 Fenham Prehistoric Enclosure Fenham, Northumberland (NU 42705 46881) Policy Unit 4.3 Hold the Line

The degraded earthwork remains at Fenham of a prehistoric enclosure, probably a substantial 'palisade' site, warrant further investigation. Although the upstanding remains are slight the importance of the site and the imminent nature of the erosion mean that this is an archaeological resource of high potential and rarity value. The site could yield significant information about lowland enclosures, settlement and farming activities during later prehistory as well as help address the problem of the dating of palisaded sites in northern England, important objectives of the regional research framework. The site is, therefore, of high regional significance and although its surviving condition remains broadly unknown, though it evidently has substantial cut features surviving given the cropmark formation, the threat and significance increase the score of this monument. Furthermore, a significant portion of this large site has already been lost to the sea and the site is continuing to erode. There is potential for this site to be considered for putting forward for designation.



Fig. 7.6 The location of a slump below the Fenham enclosure. The ranging pole shows the location of the original centre of the enclosure bank which can be seen as a slight upstanding earthwork on the ground surface above (Scale = 2m).

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being considered the most preferable strategy.

- Detailed level 3 field survey of surviving earthworks, followed by closespaced fieldwalking and geophysical survey, targeted evaluation trenching and cutting back of the cliff section through the defences to gain a better idea of what survives, the condition of preservation and assess the date of the monument and its ability to answer key research questions. Production of report followed by on-going monitoring and further works if necessary.
- Continued monitoring of the cliff face and environs of the site to assess the effects of erosion.
- No further work.

The favoured option is the level 3 recording as this would provide the necessary information to gain some understanding of the date of these features, how the site was built and how it functioned, before further erosion degrades the integrity of this large complex. Here, a sensitive archaeological approach is required so as not to further destabilise the cliff edge. Fieldwalking and geophysical survey followed by targeted evaluation to gain further information on the preservation and extent of what remains is considered a priority. The site should at least be subject to ongoing monitoring to assess the extent and nature of any archaeological deposits that are exposed in due course. This site could provide a useful counterpart to the well-known sites in East Lothian, such as Broxmouth and Dryburn Bridge, and shed light on later prehistoric coastal settlement in North East England. The latter two sites were similar lowland enclosures under the plough and in near coastal locations, and these sites have added very significantly to the understanding of later prehistory in the region, as well as revealing evidence for being far more complex multi-period sites than the first impression of the cropmark remains suggested.



Fig. 7.7 Location of the late prehistoric enclosure at Fenham.

7.3.5 Budle Bay gun emplacement

Budle Bay, Northumberland, NU 16112 28151. Policy Unit 4.5 Hold the Line

The site at Budle Bay comprises a small military battery formerly served by a small camp, now a caravan park (Fig 7.11). The surviving remains comprise a post-medieval industrial complex serving a quarry located on the golf course at Bamburgh, with a Second World War gun emplacement (Fig 7.10) constructed on top of it. It is this structure that is the subject of this assessment. The military building survives extremely well and has several unusual features that make this structure one of only a pair on the North East coast, the other being located at Scremerston, that are unparalleled elsewhere in the country. For this reason the condition, significance and rarity scores for this structure are high. These structures are far bigger and more complex than any other emplacements that can be seen to house the same calibre gun. For some reason greater emphasis and attention was paid when constructing these particular emplacements. There is also a possibility that these are based on German military designs, although this remains to be confirmed. The reason for the substantial nature of these positions is not currently clear from their location alone.



Fig. 7.8 The large gun emplacement at Budle Bay viewed from the south (1m scale visible).

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable strategy.

• Architectural, photographic and Level 3 standing building survey with associated Level 3 earthwork survey of the environs of the site including the quarry and kilns. This to be followed by proposal of the site for

consideration for future designation. Also, continued monitoring of the site over the long term with the assistance of volunteers.

- Photographic and basic Level 1 building recording survey, followed by continued monitoring of the site.
- No further work.

The favoured option is the Level 3 standing building survey and on-going monitoring as this would provide an adequate information base for a very rare monument in advance of future coastal erosion encroaching into this area. If this is not possible the remains should be at least subject to Level 1 recording to allow direct comparison with other examples of gun emplacements. Continued monitoring could be undertaken by groups, such as the Fortress Study Group, which comprises a motivated and knowledgeable group of enthusiasts.



Fig. 7.9 The features surrounding the Budle Bay battery and fish traps.

7.3.6 Trow Point Barrow and Military Remains Trow Point, South Shields (NZ 38361 72439) Policy Unit 3.1 No Active Intervention

The whole of Trow Point is threatened by ongoing coastal erosion, and although the Second World War military remains face the same threat they are not as rare as the possible barrow. However, the nineteenth century 'disappearing gun' is a rare military monument, although only a small fragment of the original Victorian structure survives, it having been subject to later alteration. The gun that is currently visible at Trow Point is a much later twentieth century replacement that has been placed there to aid public interpretation. The presence of a surviving Bronze Age barrow has not been confirmed but the NERCZA survey has put forward a sub-circular earthwork, truncated by later features, as a possible candidate (Fig 7.6). It is positioned at the escarpment edge at the rear of the point and faces the threat of cliff collapse due to wave action destabilising the cliff edge to the north and south, which will lead to complete collapse over time (Fig 7.7). The potential significance of this monument is high, as it was thought to have been lost to quarrying, and is known to have produced a cist burial with a socketed Late Bronze Age axe head found nearby. The site, therefore, scored highly on threat, significance, potential and rarity. Having multi-period remains on the site, including those from WW1 and WW2, adds to the significance of the site which is being battered by wave action on a daily basis.



Fig. 7.10 Trow Point viewed from the west, the possible barrow is located on the high point to the right of the gun position.

The condition of the monument is currently unknown but is likely to be truncated due to antiquarian investigation and later impacts from the construction of military features. For this reason the condition scored lower. Given that this is an actively eroding site it is not thought practical to propose this site for consideration for designation.



Fig. 7.11 Trow Point and its archaeology showing projected loss of archaeological features.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable strategy of the project team.

- Level 3 archaeological survey and investigation of all of the remains on Trow Point, followed by targeted test pits and/or small evaluation trenches to evaluate the potential survival of prehistoric and wartime remains. To be followed by continued assessment of impacts of ongoing erosion and monitoring of the remains with volunteers. It is not thought that geophysics would be a suitable technique at this site given that there is so much metal around the site due to wartime activity.
- Continued monitoring of the site to assess the effects of erosion based on the NERCZA field survey. Further investigation of the potential barrow utilising test pits and evaluation excavation.
- Ongoing monitoring of effects of erosion. No further archaeological work

The favoured option is the intrusive investigation approach as this would provide an appropriate evidence base upon which to devise future management options for the site and to establish the status of the possible barrow site. One way to achieve this is to construct a community-based research, monitoring and interpretation project based on community involvement, in co-operation with the National Trust, who currently manage the site. This could involve training in archaeological techniques and monitoring as well as the production of suitable interpretation and outreach opportunities for local schools which would assist in the local community taking some ownership of its historic assets and also helping to access funding streams.

7.3.7 Amble 19th century hulks

Amble, Northumberland, 19th century hulks (NU 26382 97995) Policy Unit 15.2 Managed Retreat

The Amble hulks are located in the inter-tidal zone of the estuary of the River Coquet in Northumberland (Figs. 7.12 and 7.13). They have been the subject of a limited programme of research and are still poorly understood. The NERCZA field survey identified them as being threatened by every high tide and, although photographed and accurately located with basic measurements taken, a detailed survey of these inter-tidal hulks still has not been undertaken. There are many sites in Britain where inter-tidal hulks have been recognised, however there is no comparable assemblage of hulks from a similar period which survive to this extent along the North East coast. Others have been seen at Newburn on the River Tyne, but these do not survive as well and have already been surveyed (Taylor and Williams 2009). For this reason the Amble hulks scored highly against the significance, rarity and condition criteria.

The threat faced to these vessels by every high tide, and the build up of inter-tidal mud, has led to what remains being scored highly in terms of the threat criteria. They are also well within the Environment Agency flood zone (Environment

Agency 2007), and could eventually become permanently submerged with rising sea levels. For these reasons the threat level also scored highly.

In order to more fully understand the nature of these vessels detailed work needs to be undertaken, including detailed measured survey and analysis by experienced maritime archaeologists or historians. This would add to the public's knowledge of the historical maritime industry in the North East and help to tie down the exact function and date of the vessels. For this reason the hulks also scored highly against the 'potential' criterion. However, the potential to designate is currently considered low until the results of any further work are analysed. This consideration may change in the light of any future information.



Fig. 7.12 Three of the hulks in the inter-tidal muds in the Coquet estuary, Amble.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable strategy.

- Detailed measured survey of each of the hulks at Amble, followed by detailed study and comparison of them with other similar vessels regionally and nationally. Production of a report and assessment on these findings followed by ongoing monitoring utilising volunteers if possible.
- Continued monitoring of the site to assess the effects of erosion.
- No further work

The favoured option is the detailed recording as this would provide the most information and preserve the resource prior to erosion or burial by inter-tidal mud. This is considered the most appropriate approach as the remains are still relatively poorly understood despite having been subject to rapid survey. Monitoring could be undertaken by suitably experienced individuals. Two experienced archaeologists, Alan Williams and Patrick Taylor, have already expressed an interest in recording these remains and have already surveyed similar remains at Newburn on the Tyne. They could be included in a project to further record and investigate the remains and manage their long-term monitoring.



Fig. 7.13 The location of the Amble hulks in the mouth of the River Coquet.

7.3.8 Scremerston late prehistoric enclosure Scremerston, Northumberland (NU 03177 72173) Policy Unit 3.1 No Active Intervention

The late prehistoric enclosure identified at Scremerston as a cropmark was not visible on the surface as part of the field investigation. The current condition of this monument is therefore unknown and as a result the site warrants further investigation. If significant below ground remains do survive the site could provide valuable information concerning lowland enclosures, settlement and farming activities during later prehistory as well as help address the problem of the dating of such sites in northern England, important objectives of the regional research framework. The site is of high regional significance and although its surviving condition remains unknown, it evidently has substantial cut features surviving given the cropmark formation. The threat and potential significance increase the score of this monument. A significant portion of this site has already been lost to both the sea and the cutting for the East Coast mainline (Fig.14). The site is also continuing to erode as can be seen from the small section surviving to the north of the Railway (Fig.15). However the presence of the East Coast mainline will most likely lead to investment in sea defences along this stretch of coastline, ultimately protecting the enclosure although it has heavily truncated the monument.



Fig. 14 Location of the Scremerston late prehistoric enclosure, viewed looking South.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being considered the most preferable strategy.

• Close-spaced fieldwalking and geophysical survey, targeted evaluation trenching and cutting back of the cliff section through the defences to gain a better idea of what survives, the condition of preservation and assess the date of the monument and its ability to answer key research questions. Production of report followed by on-going monitoring and further works if necessary.
- Continued monitoring of the cliff face and environs of the site to assess the effects of erosion.
- No further work.

The favoured option is geophysical survey as this would provide the necessary information to gain some understanding of the survival of below ground features, how the site was built and how it functioned, before further erosion degrades the integrity of what survives. Here, a sensitive archaeological approach is required so as not to further destabilise the cliff edge and avoid any impact upon the railway cutting. Fieldwalking followed by targeted evaluation to gain further information on the preservation and extent of what remains is considered a priority. The site should at least be subject to ongoing monitoring to assess the extent and nature of any archaeological deposits that are exposed to the east of the railway cutting in due course.



Fig 7.15 Location of late prehistoric enclosure at Scremerston

7.3.9 Scremerston gun emplacement

Scremerston, Northumberland (NU 03177 72173) Policy Unit 3.1 No Active Intervention

The Gun emplacement at Scremerston is built to the same specification as that at Budle Bay (Fig 7.16), although the setting at Budle Bay led to a slightly different final shape being used. As discussed in Chapter 5, these structures are the only two emplacements of this type built to this high standard seen in the country. They are more akin to German designs of the 1940s seen in Hitler's "Atlantic Wall". This has scored the same in most of the criteria as the emplacement at Budle and for the same reasons. However, there is slightly less direct threat to this monument from the effects of erosion (Fig 7.17), and it is less likely to be demolished and removed. This has led to the threat being scored slightly lower than the battery at Budle. The lime works and kiln upon which the battery is situated are also under threat of erosion. However the significance and rarity of these remains means they have scored lower than the surviving military archaeology.



Fig 7.16 Gun emplacement at Scremerston, built on top of a trackway associated with a limestone quarry and associated kilns (Scale = 2m).

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable strategy.

- Architectural, photographic and Level 3 standing building survey with associated Level 3 earthwork survey of the environs of the site, including the quarry and kilns. To be followed by proposal to be considered for designation. Also, continued monitoring of the site in the long term with an appropriate volunteer group.
- Photographic and basic Level 1 building recording survey, followed by continued monitoring of the site.
- No further work

The favoured option is the Level 3 standing building survey and on-going monitoring as this would provide an adequate information base for a very rare

monument in advance of future coastal erosion encroaching into this area. The proposals are based on the same principals as those for the Budle Bay battery.



Fig. 7.17 The location of threatened features at Scremerston.

7.3.10 Budle Bay fish traps Budle Bay, Northumberland (NU 16112 28151) Policy Unit 4.5 Hold the Line

The fish traps at Budle bay (Figs 7.18 and 7.11) are potentially significant, as they could relate to a grange of Lindisfarne Priory, or to a nearby, but now deserted, medieval village. The remains appear to be wood and stone-built and are exposed to erosive wave action at every high tide. This places the remains high in terms of significance and threat. There are not many well-preserved examples of medieval fish traps nationally and no similar examples regionally. This means that this site scores high against rarity as well. There is potential to designate these remains as they lie within a very shallow protected embayment in an inter-tidal zone that has remained fairly stable for a considerable period of time.



Fig. 7.18 View of some of the surviving Budle Bay fishtraps at low tide.

There is certainly potential for further work including a baseline survey of each of the fish traps and possibly limited excavation and sampling to attempt to gain accurate dating information. In addition, the survival of the remains visible on the surface is also excellent. For this reason the site scored highly in the threat criteria.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being considered the most preferable strategy.

• Detailed Level 3 field survey of surviving structural remains, followed by limited targeted excavation to gain accurate structural details and dating samples for the surviving structures, and to understand their construction

and how the traps are likely to have worked. Production of report and assessment of the survival of the site followed by on-going monitoring.

- Continued monitoring of the site to assess the effects of erosion.
- No further work

The favoured option of the project team is Level 3 recording and investigation as this would provide the necessary information to gain some understanding of the date of these features, how they were built and how they functioned before further erosion degrades the integrity of this large complex. As with other sites investigated, survey and targeted excavation of these features could be effectively facilitated as part of a community project in order to provide training opportunities as well as an outreach programme.

7.3.11 North Gare WWI seaplane base

North Gare, Seaton Carew, Teeside (NZ 53276 21480) Policy Unit 13.4 No Active Intervention

The First World War seaplane base at Seaton Carew is a rare surviving example of one of these installations (Fig 7.19). Although much Second World War heritage survives, the First World War is not as well represented in the archaeological record. To find an undeveloped site with surviving earthwork and structural elements, including the slipway, is exceptional on the North East coast. For this reason the site scored highly against potential, significance, and rarity criteria.

The site has been demolished, but not flattened, as earthwork elements survive, along with two contemporary sheds close to the power station boundary. The condition, therefore, is only average but there is potential for further investigation below ground to locate buildings and perhaps produce a basic plan of the facility. For these reasons, however, there is limited potential to suggest the site for consideration for designation.



Fig. 7.19 The preserved slipway to the First World War seaplane base at Seaton Carew.

The threat to the site is clearly high with every high tide contributing to the gradual degradation of the slipway. The low-lying area of the remainder of the base is also at risk from rising sea levels and falls well within the Environment Agency flood zone (Environment Agency 2007). The threat to the site scores maximum as it is clearly under high and on-going threat.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable strategy.

- Detailed Level 3 field survey of surviving earthworks and structural remains, followed by close-spaced geophysical survey and subsequent test-pitting/evaluation trenching of geophysical anomalies. Production of report, plan and in-depth desk-based assessment of the site followed by on-going monitoring utilising volunteers.
- Continued monitoring of site to assess the effects of erosion.
- No further work

The favoured option is Level 3 detailed survey and investigation as this would provide essential baseline information on this rare site in advance of the inexorable effects of coastal erosion. A project here could involve local communities, history groups and schools and reveal more information about a potentially significant site, with the added value of community engagement. The site should at least be monitored regularly to assess the rate of degradation. This could again be undertaken by motivated local people guided by an experienced archaeologist.



Fig. 7.20 Location of the WW1 seaplane base at North Gare.

7.3.12 Hartley; Roberts Battery

Seaton Sluice, Northumberland (NU 34266 76151) Policy Unit 24.1 Hold the Line

The site at Roberts Battery contains the surviving remains of a military installation constructed between the First and Second World Wars. The visible structures and earthworks on the site have elements of both early defences, an encampment and a large-scale battery built in response to the German bombardments of the North East coast during the First World War (see section 5.5). The site comprises two main components, Fort House, and the remains of the battery itself. The most threatened area is the structural remains of the subterranean gun emplacements, which are very close to the cliff edge, and the only visible surface remains are fragmentary and heavily damaged. For this reason the site scored highly under threat despite the SMP2 policy being Hold the Line and scoring lower on condition.



Fig. 7.21 Location of Robert's Battery earthworks viewed looking North East from Fort House.

The site has scored highly for significance due to the rare elements that survive within Fort House, including a defended latrine block (see section 5.5). There is significant potential for developing the understanding of this type of site through further study of both Fort House and the battery complex. This is still the case when considering the relatively poor condition of the battery site on the surface as the condition of the below ground remains, currently inaccessible, is unknown at present. Establishing the condition of the subterranean element of the battery is key in developing a future management plan, and therefore further investigation of this part of the site would be preferable.



Fig. 7.22 Location of archaeological features recorded at Robert's Battery.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable strategy.

- Detailed Level 3 field survey of surviving earthworks and structural remains, followed by close-spaced geophysical survey to locate the extent of subterranean features of the battery. Production of report and assessment of full survival of the site followed by on-going monitoring utilising volunteers from the local history group.
- Continued monitoring of site to assess the effects of erosion and the retreating cliff face utilising local history group and volunteers.
- No further work

The favoured option is Level 3 detailed survey as this would provide essential baseline information on this site in advance of the inexorable effects of coastal erosion. A project here could involve local history groups or interested military study groups (for example the Fortress Study Group) and reveal more information about a potentially significant site, with the added value of community engagement. The site should at least be monitored regularly to assess the rate of degradation. This could again be undertaken by motivated local people guided by an experienced archaeologist.

7.3.13 Nessend Lithic Scatter, Holy Island Holy Island, Northumberland (NU 12877 43652) Policy Unit 5.1 No Active Intervention

The Lithic scatter at Nessend is a potentially significant and threatened Mesolithic resource. The extent of the scatter has been previously recorded in detail (O'Sullivan and Young 1995) and has now been re-established as part of the rapid field survey (see section 5.14). The area faces two main threats; from erosion of the unstable edge of the former quarry and from run-off over the exposed clay surface into the quarry. The latter of these two processes is exposing the extent of the scatter which is subsequently being scoured by wind blown sand and eroded by run off after periods of rain. Consideration of these factors has meant that the site has scored highly on level of threat.



Fig. 7.23 The area of exposed clay at Nessend containing worked flints looking North.

The site is potentially significant due to the make up of the lithic assemblage (see section 5.14) and as such has scored highly against the significance and rarity criteria. There is potential for further close-spaced fieldwalking and re-mapping the precise extent of visible flints to provide comparative data which could be used in conjunction with the information on the extent of the scatter as described in O'Sullivan and Young (1995). This will allow any changes in the area exposed, and known to contain flintwork, to be accurately calculated. The NERCZA survey has established the approximate extent of the visible flint scatter, but onsite recording with a total station would be required to obtain more accurate locations for individual findspots as part of any further work.



Fig. 7.24 Location and extent of Nessend lithic scatter.

Management options

Three possible recording strategies for the site are listed in order of preference, with the first being the most preferable.

- Close-spaced fieldwalking or gridded surface collection of the area of exposed clay surface followed by close-spaced geophysical and magnetic susceptibility survey, with subsequent targeted evaluation trenching or test-pitting based on the results of this. Production of report and assessment of full survival of the site followed by on-going monitoring utilising volunteers from a local history group, or the Borders Archaeological Society.
- Continued monitoring of the site to assess the effects of erosion on the retreating quarry face utilising local history group and volunteers.
- No further work

The favoured option is the first as this would assist in characterising the site and assessing its significance, as well as there are further remains surviving in addition to the lithic scatter, in advance of coastal erosion and damage to the site from surface water run off. A project here could involve local amateur archaeology groups and reveal more information about a potentially significant site, with the added value of community engagement. The site should at least be monitored regularly to assess the rate of degradation. This could again be undertaken by local people guided by an experienced archaeologist.

7.4 Conclusions

The NERCZA project has identified priority sites at risk from coastal erosion and has suggested various management options for those sites at 'imminent risk'. The assessment of interest and threat set out in Table 7.1 allows for the formation of management options for each of the sites on this priority register. This means that the raw data collected by the NERCZA project can be used as a management tool for forming positive archaeological strategies and actions. It can also be used for assessing condition, protection, recording, and where possible, preservation of archaeological sites.

This was one of the key overarching aims of the project and the value of the new data added to Historic Environment Records by both phases of the project has meant there is now a sound evidence base for future decision-making and actions. This exercise has produced a useful methodology to guide future monitoring of coastal assets that could be repeated at a local scale, at regular intervals, and at relatively low cost, particularly if volunteer groups were included under the supervision of a professional archaeologist. This could be achieved through a series of schemes designed to monitor and investigate the archaeology of the coast. This would allow local communities to further engage with their coastal heritage while contributing to the understanding, investigation and monitoring of heritage assets. Crucially, such projects would provide the necessary sustainability, particularly for monitoring work, into the future. Projects following this format would facilitate partnerships between professional archaeologists and volunteers through community inclusion, outreach and training. Such projects would not only help rescue remains from destruction without record, but they would also generate public interest, enjoyment and knowledge gain. Funding could be sought from a variety of organisations and could include the Heritage Lottery Fund, English Heritage, Defra, Natural England, Environment Agency, Leader Plus and perhaps maritime businesses such as North Sea oil companies.



Fig. 7.25 Rapid recording of an eroding pillbox at Warkworth in Northumberland.

The whole of the North East coastline could be broken down into chunks, perhaps based on the SMP policy unit areas, with an archaeological project set up to cover each area. Alternatively projects could be set up on a site by site basis according to need. Such projects would enable community engagement with coastal heritage, combined with ongoing monitoring of heritage assets. As an example, North Yorkshire and Teesside could effectively be covered by one overarching project, due to the overlap in the North Yorkshire Moors National Park and Teesside Historic Environment Records and the relatively small area concerned. This project could investigate the ongoing condition of the surviving alum works and expand on the work of the rutways survey project run by Tees Archaeology (Green 2009).

There is great potential for extensive community involvement in such projects, including local groups, schools, as well as visitors to the coast. Widespread involvement would aid in raising awareness amongst the public, capacity building within the heritage sector as well as locking in the volunteer sector. If such projects could be delivered then not only would the ongoing recording and monitoring of eroding assets continue into the future, but it would help maximise the benefit of such work to society whilst also reducing its cost.

8. REFERENCES

Ainsworth S, Bowden M. and McOmish D. 2007. Understanding the Archaeology of Landscapes: A guide to good recording practice. Swindon, English Heritage.

Allen, J. R. L., Bell, M.G. and Scales, R.R.L. (2004) Animal and human footprint tracks in archaeology: description and significance. Archaeology in the Severn Estuary, 14. pp. 55-68.

Bonsall C. 1984. Low Hauxley, Northumberland: Mesolithic/Bronze Age coastal site. *Proceedings of the Prehistoric Society* 50: 398.

Boomer I, Waddington C, Stevenson, AC and Hamilton, D. (2007). Holocene coastal change and geoarchaeology at Howick, Northumberland, UK. *The Holocene* 17(1): 89-104.

Bronk Ramsey, C, 1995 Radiocarbon calibration and analysis of stratigraphy, Radiocarbon, 36, 425–30

Bronk Ramsey, C, 1998 Probability and dating, Radiocarbon, 40, 461-74

Bronk Ramsey, C, 2001 Development of the radiocarbon calibration program, Radiocarbon, 43, 355-63

Bronk Ramsey, C, 2009 Bayesian analysis of radiocarbon dates, Radiocarbon, 51: 337-60

Bronk Ramsey, C, Higham, T, and Leach, P, 2004 Towards high precision AMS: progress and limitations, Radiocarbon, 46(1), 17–24

Bronk Ramsey, C, Higham, T F G, Owen, D C, Pike, A W G, and Hedges, R E M, 2002 Radiocarbon dates from the Oxford AMS system: *Archaeometry* datelist 31, *Archaeometry*, 44(3), Supplement 1, 1–149

Burn A and Knight D 2010. A rapid survey of Second World War remains within the Northumberland Coast Areas of Outstanding Natural Beauty. Archaeological Research Services Ltd

Burgess C and Rushton S (Eds) 2010. Explore the Archaeology of the Northumberland Coast Area of Outstanding Natural Beauty and Heritage Coast. Northumberland Coast AONB Partnership.

Carruthers R. 1930. Geology of the Alnwick district. *Memoir of the Geological Survey of England and Wales* 105.

Chatterton R and Waddington, C 2009. North East Rapid Coastal Zone Assessment Phase 2 updated project design. Archaeological Research Services Ltd

Clarke B 2008. The archaeology of airfields. Tempus Publishing

Cocroft W D and Thomas R J C 2004. Cold War, building for nuclear confrontation 1946 - 1989. English Heritage

Cruickshank D 2001. Invasion; defending Britain from attack. Boxtree

Dobinson C 2000. Fields of Deception, Britain's Bombing decoys of World War Two. Swindon, English Heritage

Dobinson C, 2001 AA Command: Britain's Anti Aircraft defences of World War II. Swindon, English Heritage

Dobinson C, 2008 Building Radar Forging Britain's Early-Warning Chain, 1935-45. Swindon, English Heritage

Drury D. Howard-Davis, C and Newman, R. 1995. Low Hauxley, Northumberland: Archaeological Evaluation. Lancaster University report for English Heritage.

Foster J 2004. Guns of the North East, coastal defences from the Tyne to the Humber. Pen and Sword

Fowler A. 1936. Geology of Country around Rothbury, Amble and Ashington. *Memoir of the Geological Survey of England and Wales* 135. Green G 2009. *Saltburn Rutways Survey 2009 report*. Tees Archaeology

Hart C 2004. The coast from South Shields to Whitburn, archaeology and history. The Arbiea Society, South Shields.

Hardie C and Rushton S 2000. The tides of Time: Archaeology on the Northumberland Coast. Northumberland County Council

Hegarty C and Newsome S 2007. Suffolk's defended shore, coastal fortification from the air. English Heritage

Holbrow W, 2003 Historic Military Aviation sites, conservation management guidance. Swindon, English Heritage.

Howes R. 1864. On the glaciation of the counties of Durham and Northumberland. *Transactions of the North of England Institute of Mining and Mechanical Engineers* XIII May.

Hunt A and Ainsworth S 2006. Craster, Northumberland, An archaeological investigation of a World War II radar station complex. English Heritage

Hunt A, Pearson T, Sinton P, Carter A, Andrews D, Beckett N, Tovey S, and Clowes M 2004. Loftus alum works, Redcar and Cleveland: an archaeological and historical survey. English Heritage

Innes J.B. and Frank, R.M. 1988 Palynological evidence for Late Flandrian coastal changes at Druridge bay, Northumberland. *Scottish Geographical Magazine* 104: 14-23.

Kueppers, L M, Southon, J, Baer, P, and Harte, J, 2004 Dead wood biomass and turnover time, measured by radiocarbon, along a subalpine elevation gradient, *Oecologia*, 141(4), 641–51

Lowry B (Ed) 2002. 20th Century Defences in Britain, An introductory guide. York Council for British Archaeology Practical Handbook 12.

Mook, W G, 1986 Business meeting: recommendations/resolutions adopted by the twelfth International Radiocarbon Conference, Radiocarbon, 28, 799

Osbourne M 2008. Pillboxes of Britain and Ireland. Tempus Publishing

Osbourne M 2004. Defending Britain; 20th Century military structures in the landscape. Tempus Publishing

Oswald A, Ashbee J, Porteous K, and Huntley J 2006. Dunstanburgh Castle, Northumberland; archaeological, architectural and historical investigations. English Heritage

O'Brien C and Elliott L 2010. Low Hauxley, Hartlepool Bay, Crimdon Dene; pollen assessment and radiocarbon preparation. Archaeological Services Durham University

Oswald, A and Burn, A 2009 Analytical earthwork survey of a prehistoric promontory fort at Boltby Scar, North Yorkshire. English Heritage

Oswald, A and Pearson, T 2001 An Iron Age Promontory Fort at Roulston Scar, North Yorkshire. English Heritage

Petts D and Gerrard C 2006. Shared visions: The North East regional research framework for the historic environment. Durham County Council

Powlesland, D 2009 Geophysical Surveys at Boltby Scar, North Yorkshire (SE 506 856 \sim SMR No 26932). The Landscape Research Centre

Raistrick A. and K.B. Blackburn. 1932. The late-glacial and post-glacial periods in the North Pennines, Part III - the post-glacial peats. *Transactions of the Northern Naturalists Union* 1: 79-103.

Raistrick, A. and T.S. Westoll. 1933. A prehistoric site on the south Durham coast. *Vasculum* 19: 139-144.

Reimer, P J, Baillie, M G L, Bard, E, Bayliss, A, Beck, J W, Blackwell, P G, Bronk Ramsey, C, Buck, C E, Burr, G S, Edwards, R L, Friedrich, M, Grootes, P M, Guilderson, T P, Hajdas, I, Heaton, T J, Hogg, A G, Hughen, K A, Kaiser, K F, Kromer, B, McCormac, G, Manning, S, Reimer, R W, Remmele, S, Richards, D A, Southon, J R, Talamo, S, Taylor, F W, Turney, C S M, van der Plicht, J, and Weyhenmeyer, C E, 2009, INTCAL09 and MARINE09 radiocarbon age calibration curves, 0–50,000 years cal BP, *Radiocarbon*, 51(4), 1111–50

Schofield J (ed), 1998 Monuments of War, the evaluation, recording and management of twentieth century military sites. Swindon, English Heritage.

Scott, E M, 2003 The third international radiocarbon intercomparison (TIRI) and the fourth international radiocarbon intercomparison (FIRI) 1990–2002: results, analyses, and conclusions, *Radiocarbon*, 45, 135–408

Shennan I., Horton B., Innes J., Gehrels R, Lloyd, McArthur J. and Rutherford M.2000. Late Quaternary sea-level changes, crustal movements and coastal evolutionin Northumberland. *Journal of Quaternary Science* 15(3), 215-237.

Shennan I., K. Lambeck, et al. 2000. Modelling western North Sea palaeogeographies and tidal changes during the Holocene. In I. Shennan and J. E. Andrews (eds.) *Holocene Land-Ocean Interaction and Environmental Change around the North Sea*. London, Geological Society, Special Publication 166: 299-319.

Slota Jr, P J, Jull, A J T, Linick, T W, and Toolin, L J, 1987 Preparation of small samples for ¹⁴C accelerator targets by catalytic reduction of CO, *Radiocarbon*, 29, 303–6

Stuiver, M, and Reimer, P J, 1993 Extended ¹⁴C data base and revised CALIB 3.0 ¹⁴C age calibration program, *Radiocarbon*, 35, 215–30

Stuiver, M, and Polach, H A, 1977 Reporting of ¹⁴C data, Radiocarbon, 19, 355-63

Stuiver, M, and Kra, R S, 1986 Editorial comment, Radiocarbon, 28(2B), ii

Stuiver, M, and Reimer, P J, 1986 A computer program for radiocarbon age calculation, Radiocarbon, 28, 1022–30

Taylor M 2010. Waterlogged Wood from Low Hauxley, Northumberland, and Assesment.

Tipping, R. 1994. Low Hauxley. Palaeoenvironmental Reconstructions. Unpublished report submitted to Clive Bonsall and Northumberland County Council.

Tolan-Smith C 2008. North East Rapid Coastal Zone Assessment (NERCZA) Phase 1 report. Archaeological Research Services Ltd

Trechman C.T. The submerged forest beds of the Durham coast. *Proceedings of the Yorkshire Geological Society* 27: 23-33.

Vandeputte, K, Moens, L, and Dams, R, 1996 Improved sealed-tube combustion of organic samples to CO_2 for stable isotope analysis, radiocarbon dating and percent carbon determinations, *Analytical Letters*, 29(15), 2761–73

Waughman M 2005. Archaeology and environment of submerged landscapes in Hartlepool Bay, England. Tees Archaeology Monograph Series Vol 2

Waddington C 2010. Low Hauxley, Northumberland: a review of archaeological interventions and site condition. Archaeological Research services Ltd

Waddington C (Ed) 2007. Mesolithic Settlement in the North Sea Basin, A case study from Howick, North-East England. Oxbow Books

Waddington C and Pedersen K (Eds) 2007. Mesolithic Studies in the North Sea Basin and Beyond, proceedings of a conference held at Newcastle in2003. Oxbow Books

Waddington C and Passmore D 2004. Ancient Northumberland. English Heritage

Ward, G K, and Wilson, S R, 1978 Procedures for comparing and combining radiocarbon age determinations: a critique, *Archaeometry*, 20, 19–31

Wilson P. et al. 2001. Late Holocene (post-4000 years BP) coastal dune development in Northumberland, northeast England. *The Holocene* 11(2): 215-229.

Xu, S, Anderson, R, Bryant, C, Cook, G T, Dougans, A, Freeman, S, Naysmith, P, Schnabel, C, and Scott, E M, 2004 Capabilities of the new SUERC 5MV AMS facility for ¹⁴C dating, *Radiocarbon*, 46, 59–64

























